ELEMENTAL COMPOSITION OF ATMOSPHERIC PARTICULATE, WATER AND SOIL SAMPLES FROM URBAN AND POLUTED ENVIRONMENTS.

CHARLES McDONALD

Thesis presented for

the Degree of

Doctor of Philosophy.

October 1977

Agricultural Chemistry, University of Glasgow. ProQuest Number: 13804130

All rights reserved

INFORMATION TO ALL USERS The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 13804130

Published by ProQuest LLC (2018). Copyright of the Dissertation is held by the Author.

All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

> ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 – 1346

ACKNOWLEDGEMENTS

I am indebted to Dr. H.J. Duncan who supervised the work described in this thesis. His guidance and advice during all stages of the work has been of immense value.

I am grateful to Mr. I. McNeil of the department for help in many ways, and for helpful discussions on atomic absorption technique.

Other people and centres who must be thanked are:-

The National Engineering Laboratory at East Kilbride for accessibility to reactor facilities.

The Education Authorities of Glasgow, Dunbartonshire and Ayrshire for permission to use schools as sampling sites and the headmasters of the individual schools used.

The Glasgow Weather Centre for access to weather data.

Mr. Christie, the Glasgow Smoke Officer (Department of Health), for access to total smoke and sulpher dioxide records.

Various members of the computing department for help with the computer programmes.

The Science Research Council for financial support from 1972-75.

I am grateful to Mrs. Hosie for undertaking the typing of this thesis and her efficiency in doing so.

Finally I wish to thank Margaret, my wife, for her help in preparing the final draft of this thesis and for her patience and understanding during the course of the work.

i

SUMMARY

Chapter I deals with the reasons for the increased public concern over pollution of the environment, including the deaths of 43 people in Japan in 1953, and the death of over 50,000 seabirds in the Irish Sea in 1969.

Chapter II discusses methods which have been used in the analysis of metals in environmental samples with comments on their strengths and weaknesses. The two methods which were used in this work, Activation Analysis and Atomic Absorption Spectroscopy are discussed in more detail.

Chapter III deals with atmospheric levels of some 24 elements. Reasons for a study of elemental levels are given.

Sampling problems such as best filter paper, filter blank values, pressure drop across the filter, etc. are considered.

The accuracy of the methods used was checked using standard air pollution filters.

The results of one site sampling over a period of 13 months are given and compared to other world centres. Attempts were made to correlate the levels with weather parameters with only partial success. It was found that although the weather has an effect on levels; fog and calm conditions increasing levels while rain decreased levels, this is not linear. Seasonal variations were also studied with peak values for most elements in winter except for Se and Sb which tend to peak in spring and summer.

ii

Levels at 13 other city sites and 2 rural sites were also measured but no area was found to be consistently high in all elements. The rural sites did in general show lower levels than the city.

Particle size distribution of elements was also studied using a 5 stage cascade impactor the principles of which are discussed. The results are discussed in terms of implications to human health with disturbing results for Cd, Pb and Ni, large proportions of which can penetrate deep into the lung. The results are also compared favourably with those of Liege (Belgium).

A term often used in studying atmospheric levels of elements is the enrichment factor (see Section 3.5.3.3) where atmospheric levels are compared to some standard. When compared to soil the elements show two trends. One class has ratios close to those in soil while the other is enriched by 100 - 10,000 times. It has been suggested that this is consistent with a fuel burning source and is fully discussed.

Good agreement has been found with this theory in all sections of this work. The results of 1 site sampling agree well with those of Liege (Belgium) while the enrichment factors for the other city and rural sites are remarkably constant in the first class (close to soil ratios) with higher and more varied factors for the others. Again the factors of the first class vary little with particle size, the variations being found in the second class.

iii

Chapter IV deals with metal levels in Glasgow's water supply a constant source of worry due to a combination of soft water and lead pipes. A survey was carried out analysing for 7 elements. The results indicate that Pb levels are higher than W.H.O. limits while Cu could become a problem if W.H.O. desirable limits are applied. Other elements are not a problem at this stage.

Leaching was also studied by a controlled experiment using standard lengths of pipes in common use. Results indicate that altering the material of plumbing will only change the problem of metal levels and not necessarily remove them.

Chapter V deals with some problems associated with the dumping of coal mine waste on land (bings). Bing samples were analysed for 8 metals and the results discussed in terms of plant growth. Only Cu had higher than normal values for normal soil ranges. The effect on surrounding land was also investigated in terms of drainage into streams and seepage and blow-off on to land. The results suggest that these might be occurring but further work is required to confirm the tentative findings of this work.

iv

CONTENTS

Page

ACKNOWL	EDGEMENTS	i
SUMMARY		ii
TABLES		ix
FIGURES		xii
<u>CHAPTER</u>	I - INTRODUCTION	Ī
	AIMS OF PROJECT	6
CHAPTER	II - ANALYSIS	
2.1	INTRODUCTION	8
2.2	X-RAY SPECTROSCOPY	8
2.3	ANODIC STRIPPING VOLTAMETRY	IO
2.4	SPARK SOURCE MASS SPECTROMETRY	12
2.5	EMISSION SPECTROSCOPY	13
2.6	GAS CHROMATOGRAPHY	14
2.7	ACTIVATION ANALYSIS	17
2.8	ATOMIC ABSORPTION SPECTROSCOPY	24

CHAPTER III - AIR POLLUTION

3.1	INTRO	DUCTION	38
3 2	סאסאס		41
J•2	IANI		15
3.3	SAMPI	LE COLLECTION	+) - •
3.4	REPRO	DUCIBILITY OF FILTER PAPER BLANKS	54
3.5	ATMOS 'THE I	SPHERIC LEVEL OF TRACE ELEMENTS AND IR VARIATION WITH WEATHER	59
3.	5.1 \$	Sampling	59
3.	5.2 <i>I</i>	Inalysis	61
3.	5.3 H	Results	71
	3.5	.3.1 Correlation of Elements	83
	3.5	5.3.2 Seasonal Variations	86
	3.5	5.3.3 Enrichment Factors	93
	3.5	5.3.4 Climatic Variations	98
3.	5.4 0	Conclusions	134
3.6	VARIA	TION IN ELEMENTAL CONCENTRATION WITH LOCATION	136
3.	6.1 s	ampling	136
3.	6.2 A	nalysis	I40
3.	6.3 F	desults	I49
	.3.6	.3.1 Correlation Between Elements	I4.9
	3.6	.3.2 Enrichment Factors	τ52
	3.6	.3.3 Variation of Elements with Sites.	<u>1</u> 56
	·6	.3.4 Effect of Wind	159
	3.6	.3.5 Comparison of City and Rural Areas	161
3.0	6 . 4 0	onclusions	163

3.7 PARTICLE SIZE DISTRIBUTION OF ELEMENTS IN THE ATMOSPHERE	I64
3.7.1 Sample Collection	165
3.7.2 Principles of Impaction	I65
3.7.3 Analysis	17 0
3.7.4 Results	171
3.7.4.1 Inhalation	171
3.7.4.2 Mass - Median Diameter	I75
3.7.4.3 Size Types	I75
3.7.4.4 Enrichment Factors	I76
3.7.5 Conclusions	179
3.8 ASSESSMENT OF PROJECT AND POSSIBLE FUTURE WORK	180

CHAPTER IV - WATER POLLUTION

4.1	INTRODUCTION	181
4.2	METALS STUDIED	1 82
4.3	SAMPLING	I83
4.4	ANALYSIS	1 84
4.5	RESULTS	187
4.6	LEACHING FROM PIPES	192
4.7	CONCLUSIONS	I94
4.8	ASSESSMENT OF PROJECT AND POSSIBLE HUJTURE WORK	I95

CHAPTER V - INDUSTRIAL WASTE

5,1	INTRODUCTION	I96
5.2	TRACE METAL LEVELS IN SCOTTISH COAL MINE WASTE	I98
5	.2.1 Sample Collection and Preparation	I98
5	.2.2 Analysis	200
5	.2.3 Results	203
5.3	EFFECT ON SURROUNDING LAND	207
5	.3.1 Effect on Streams	207
5	.3.2 Effect on Land	210
5•4	CONCLUSIONS	251
5.5	ASSESSMENT OF PROJECT AND POSSIBLE FUTURE WORK	252

REFERENCES

.

TABLES

ix

Page

.

CHAPTER III - ATMOSPHERIC POLLUTION

.

3. 1	VARIATION IN ELEMENTAL CONCENTRATION WITH HEIGHT	48
3.2	CONCENTRATIONS OF ELEMENTS IN FILTERS	53
3.3	VARIATION IN MILLIPORE BLANK VALUES WITH BOX	55
3.4	VARIATION IN BLANK VALUES OF COPPER IN MILLIPORE FILTER PAPER	56
3.5	AVERAGE BLANK VALUES IN MILLIPORE FILTER PAPER	58
3.6	NUCLEAR DATA FOR ISOTOPES MEASURED	64
3.7	ELEMENTAL CONCENTRATIONS FOUND IN STANDARD FILTERS	. 70
3.8	"TRUE" ELEMENTAL CONCENTRATIONS SPOTTED ON STANDARD FILTERS	70
3.9	ATMOSPHERIC LEVELS OF TRACE ELEMENTS	72
3.10	CORRELATION COEFFICIENTS OF ELEMENTS WITH EACH OTHER AND SMOKE AND SO2	84
3.11	SELENIUM - SULPHUR RATIO	8 7
3.12	ENRICHMENT FACTORS	95
3.13	CORRELATION COEFFICIENTS OF ELEMENTS WITH WIND SPEED, DURATION AND DISTANCE	100
3.14	CORRELATION COEFFICIENTS OF ELEMENTS WITH VECTORS OF WIND PARAMETERS	103
3.15	CORRELATION COEFFICIENTS OF ELEMENTS WITH CALM HOURS	103
3.16	CORRELATION COEFFICIENTS OF ELEMENTS WITH RAIN	I09
3.17	CONCENTRATIONS OF ELEMENTS IN RAINWATER	IUð
3.18	CORRELATION COEFFICIENTS OF ELEMENTS WITH FOG	III
3.19	CORRELATION COEFFICIENTS OF ELEMENTS WITH VARIOUS WEATHER PARAMETERS	113
3.20	ELEMENTS AFFECTED BY E/NE WINDS	117

3.21	ELEMENTS AFFECTED BY NUMBER OF CALM HOURS	II9
3.22	FREQUENCY OF INCREASE OR DECREASE OF ELEMENTS	I2 3
3.23	ELEMENTS AFFECTED BY AMOUNT OF RAIN	I24
3.24	ELEMENTS AFFECTED BY FOG	1 26
3.25	ELEMENTS AFFECTED BY CLOUD COVER	12 8
3. 26	RESULTS OF MULTIPLE REGRESSION COEFFICIENT PROGRAMME	129
3.27	DESCRIPTION OF SITES THROUGHOUT THE CITY	13 8
3.28	"TRUE VALUES" OF ELEMENTS SPOTTED ON STANDARD FILTERS	I44
3.29	ELEMENTAL CONCENTRATIONS FOUND IN STANDARD FILTERS	I45
3.30	ATMOSPHERIC LEVELS OF TRACE ELEMENTS	I50
3.31	CORRELATION COEFFICIENTS OF ELEMENTS WITH EACH OTHER	151
3.32	ENRICHMENT FACTORS FOR GLASGOW SITES	I53
3.33	AVERAGE ENRICHMENT FACTORS FOR GLASGOW AND LIEGE	I54
3.34	AVERAGE LEVELS OF ELEMENTS AT CITY SITES	157
3.35	ORDER OF SITE - GLASGOW UNIVERSITY RATIOS	I58
3. 36	EFFECT OF WIND DIRECTION ON ELEMENTAL LEVELS	I60
3.37	ATMOSPHERIC LEVELS OF ELEMENTS IN RURAL SITES	1 62
3. 38	ENRICHMENT FACTORS FOR RURAL SITES	162
3.39	CONCENTRATIONS OF ELEMENTS AS A FUNCTION OF PARTICLE SIZE	I72
3.40	CONCENTRATIONS OF ELEMENTS COLLECTED AT EACH STAGE AS PERCENTAGE OF TOTAL	173
3.41	MASS-MEDIAN DIAMETERS OF ELEMENTS AND REPORTED RANGES	173
3.42	ELEMENTS CLASSED AS A SIZE FUNCTION	177
3.43	ENRICHMENT FACTORS OF ELEMENTS ON EACH STAGE	177

CHAPTER IV - WATER POLLUTION

4.1	VARIATION IN METAL LEVELS IN DRINKING WATER DURING ONE WEEK	I88
4.2	AVERAGE LEVELS OF METALS IN RUNNING TAP WATER	I89
4.3	COMPARISON OF METAL LEVELS IN STANDING AND RUNNING WATER	190
4.4	METAL LEVELS IN LEACHING WATER	193
CHAPTER	V - INDUSTRIAL WASTE	
5.1	MAP REFERENCES OF SAMPLE SITES	199
5.2	RESULTS FOR STANDARD ROCK	202
5.3	METAL LEVELS IN SCOTTISH COLLIERY WASTE	204
5.4	METAL LEVELS IN SEDIMENT AND WATER OF STONEYBURN STREAM	211

5.5 pH VALUES OF SOIL AROUND STONEYBURN BING 250

xii

FIGURES

2.1	GRAPH OF ACTIVITY V TIME FOR RADIOACTIVE DECAY	21
CHAPTER	III - AIR POLLUTION	
3. 1	SELENIUM CONCENTRATIONS IN THE ATMOSPHERE DURING MAY 1973	47
3.2	SAMPLING RATE OF MILLIPORE FILTERS WITH CHARLES AUSTIN M361 PUMP	60
3.3	APPARATUS FOR COLLECTION OF ATMOSPHERIC PARTICULATES	62
3.4	X-RAY SPECTRUM OF SHORT-LIVED ISOTOPES	65
3.5	X-RAY SPECTRUM OF LONG-LIVED ISOTOPES	66
3.6	GRAPHICAL REPRESENTATION OF ATMOSPHERIC LEVELS OF ELEMENTS, SMOKE AND SO2	73
3.7	MONTHLY AVERAGES OF ATMOSPHERIC LEVELS OF ELEMENTS, SMOKE AND SO2	88
3.8	COMPUTER PLOTS OF SELENIUM V WIND PARAMETERS	104
3.9	GRAPHICAL REPRESENTATION OF CITY SITES	I37
3.10	SAMPLING RATE OF MILLIPORE FILTERS WITH CHARLES AUSTIN CAPEX MARK II PUMP	1 41
3.11	MULTIPLE PEAKS OBTAINED WITH CHROMIUM	I47
3.12	PARTICLE SIZE PENETRATION INTO RESPIRATORY TRACT	166
3.13	GRAPHIC ILLUSTRATION OF CASCADE IMPACTOR	167
3.14	PRINCIPLES OF IMPACTION	169
3.15	GRAPHIC REPRESENTATION OF CASCADE IMPACTOR RESULTS	I74

CHPATER IV - WATER POLLUTION

CHAPTER II - ANALYSIS

,

Tel Constitution for poinsito with southi	4.1	QUESTIONNAIRE	FOR	DOMESTIC	WATER	SURVEY	I	[8]	3	, I	2
---	-----	---------------	-----	----------	-------	--------	---	-------	---	-----	---

,

Page

CHAPTER V - INDUSTRIAL WASTE

5.1	CHART OF STONEYBURN BING AND SURROUNDING LAND		208
5.2	GRAPHIC REPRESENTATION OF METAL LEVELS IN,		213
	DOID MICOUD DING	×	21)

INTRODUCTION

Public concern over levels of metals in the environment has been increasing in recent years, and nowhere more so than in Japan.

The following precis is intended to illustrate some of the events which have led to the increased awareness, and where appropriate, to give examples of biological potency, to human health although the dangers to plants and animals should not be ignored.

Japan in recent years has been showing the effects of industry expanding unchecked into rural peasant areas. The first danger signs came in Minamata, a town that in 1953 was part industrialised, part an overgrown village, where people still caught their own fish in the bay as part of their staple diet. Later it was discovered that a plastics factory had dumped 200 tons of mercury into the bay between 1949 and 1953. The inevitable of course happened - the consumption of contaminated fish caused the deaths of 43 humans and left a further 68, many of them infants and children, permanently and In the following year (1954) 5 cut of grotesquely disabled. 12 new cases died and in 1955, 3 out of 15 new cases also died. In 1956, of 50 new cases noted, 10 died and of these 7 were infants born of mothers who did not themselves show any signs A further outbreak occurred in Niigata City in of poisoning. 1970 resulting in 28 severely affected cases, 5 of whom died (1).

Slowly the world came to appreciate the dangers of mercury, but as the element had been widely used for some time in industry and as a seed-dressing in agriculture most of the damage had already been done. The troubles in Japan caused world wide concern, particularly in Sweden which makes considerable connercial use of freshwater fisheries. A national programme was set up to investigate the problem. The findings were disturbing. Eggs and meat produced in Sweden contained more mercury than was found in similar products of neighbouring countries. The findings of the report also included - (1) higher death rates than normal amongst wild seed-eating birds; (2) apparent disappearance of birds of prey; and (3) reduced breeding success was noted with the survivors. Although mercury was found in the various tissues it would be unwise to blame the element by itself.

Mercury becomes bound to the sediments of rivers and lakes, being released only slowly back into the environment, causing long term effects. One lake in Sweden was found to be still highly contaminated in 1970 - 25 years after dumping of mercury ceased. The problem became so serious in Canada that by the end of 1970 fishing was banned from 4 major lakes, 2 major rivers and almost 100 other smaller rivers (1).

Japan was again to be the bearer of further evidence of heavy metal toxicity - this time cadmium and lead. In an isolated village in Northern Japan irrigated by the Jintsu River into which a zinc mine and smelter was pouring its effluent for many years, grain crops and fish became contaminated with cadmium and lead. People living on these

foods eventually exhibited severe softening of bones with many fractures and deformities and much pain. Between 1962 and 1968 over 200 cases of this "Itai-Itai-Byo" disease in the above area were reported. The disease itself had been known to the Japanese authorities since 1955 but this was the first major outbreak of the disease (1).

Heavy metals have always of course been in the environment in the minerals of rocks and soil. When they weather, metals are liberated into the atmosphere and surface water. However, in the last 200 years man has significantly affected the natural levels through industrial growth and an ever increasing demand for power. This has increased the levels of metals such as Sn, Se, Sb, Cd, etc. which were of low natural levels and as such, life developed without the need for elaborate mechanisms of detoxification. For example, Pb is stored mainly in bones. In children this can be released during times of high metabolic activity, e.g. during fever, and could lead to lead encephalitis even although blood levels are below that normally regarded as toxic.

The increased levels of elements have been demonstrated by analysing the composition of ice-sheets at the Poles. For 2,000 years prior to 1725 the lead content of the ice at the North Pole was negligible. From 1925 to 1948 with industrial growth a steady rise was recorded, followed by a more rapid increase from 1948 to 1970. The 1970 concentration was 15 times greater than the 1948 levels. No corresponding increase was found at the South Pole since this is in the relatively industrial free Southern Hemisphere (2, 3).

Probably the biggest natural disaster in Britain was the death of 50,000-100,000 seabirds in the Irish Sea in the Autumn of 1969. Subsequent analysis of victims revealed high levels of heavy metals and poly-chloro biphenyls (PCB). Death was attributed to complex inter-acting, but debilitating, contamination effects which left them too weak to survive when they were challenged by moderately harsh conditions. This was said to be caused by the combined inter-action of sub-lethal doses (4).

One of the most dangerous aspects of body absorption of metals is that many nervous pathways are not principally required for one function but duplicate or triplicate other pathways in case of blockage or breakdown. This means that by the time chemical symptoms of nervous disorders appear enormous damage has already been done to the brain and nervous system. In addition of course extensive damage can occur without chemical symptoms being evident.

The metal levels in animals (both wild and domestic) and plants is of course important to humans, particularly if they will eventually be concumed by them. One of the main fears was that of biomagnification which is simply the increase in metal levels by repeated transfer up the food chain. In one study in particular it was found that the top predatory fish had the highest concentration of mercury compounds. However, this is now thought to be less important than originally feared and bioconcentration, i.e. the ability of an organism to

concentrate metals from its growth medium is now considered to be the main danger (5). Shell-fish are the best concentrators of all, rapidly concentrating metals to several hundred times the level found in the water in which they live (1).

From the above brief discussion it will be evident that environmental levels of metals are extremely important to the wellbeing of plants and animals alike, since these toxic metals do not degrade and once in the active cycle they remain active until buried deep into sediments. Their period of biological potency has to be measured in thousands of years.

Although for ease of presenting results the environment is divided into atmosphere, land and water, it must be remembered that they are all inter-connected. The air continually maintains contact with land and sea. There are many examples where initially air pollution has caused problems to plants in surrounding areas (6 - 19) and caused deaths of grazing livestock (20 - 24) due to factory emissions. Atmospheric fall-out has also been shown to be responsible for increased levels in Lake Michigan (25). On a more general basis dumping of waste material next to rivers has been known to cause problems in river systems.

AIMS OF PROJECT

The overall aim of the project is, as the title suggests, to study the elemental composition of atmospheric particulate, water and soil samples from urban and pelluted environments.

Under this blanket approach the basic aim is to ensure that results quoted are analytically correct within the limits given, and that sufficient samples are taken to include as many contributing conditions as possible.

Chapter II gives a brief account of the main methods used in the analysis of environmental samples, including the advantages and disadvantages of each method. The two methods which were used in this work, Activation Analysis and Atomic Absorption are dealt with in more detail.

Chapter III deals with the elemental composition of atmospheric particulates. Analytical methods used were tested for accuracy by the use of standard filter papers. Some thought was given to sampling problems, collection medium, etc., and to the most economical way of handling the large amount of data acquired. Computer programmes were considered for this latter problem and in most cases found to be useful.

Chapter IV deals with a brief survey of domestic drinking water. This is relevant to the Glasgow area due to the use of lead plumbing in this soft water area. Again consideration is given to the validity of one-off sampling and tests carried out.

Chapter V deals with chemical problems associated with industrial waste and more specifically with metal content of coal mining waste. (Coal bings). Problems in terms of reclamation of the actual bing and the effect of seepage, blow-

off and run-off on surrounding countryside are investigated. Again thought was given to the choice of site to study and the accuracy of results checked.

CHAPTER II

ANALYSIS

2.1 INTRODUCTION

In pollution studies the analytical techniques employed must be accurate, rapid, capable of determining elements in the 10^{-9} g range and if possible allow for the simultaneous determination of a number of elements.

A number of techniques have been employed in the analysis of inorganic pollutants each with their own claimed advantages over other techniques employed, and in some cases the limitations are conveniently over-looked.

It is relevant therefore to discuss briefly the techniques most often employed pointing out their particular advantages and disadvantages.

2.2 X-RAY SPECTROSCOPY

The general principles of this method are well documented in a review by Carr-Brion <u>et al</u> (26) and can be briefly summarised as follows -

When electrons are ejected from the inner orbitals of an atom their replacement results in the emission of an X-ray. Each element emits X-rays at certain characteristic energies that are equal to the energies of the corresponding electron transitions. Thus if a substance is suitably excited so that electrons are ejected, the elements composing it can be determined and, as the X-ray intensity at each characteristic energy level depends on the concentration of the element emitting it, quantitive and qualitative analysis become possible.

Existing instruments permit the detection of all elements except hydrogen and helium, although different excitation sources must be used. If sealed X-ray tube excitation is used the range is generally limited to those with atomic numbers higher than oxygen because of the difficulties associated with the excitation of characteristic X-rays from the lighter elements. Radioisotopes can also be used as excitation sources, emitting \ll , β , \aleph , and X irradiation as well as protons. X-ray fluorescence (so called when X-rays from a tube or a radioisotope are used) has been employed in air pollution studies (27-37) and water analysis (38-42). Protons as activating particles have also been used in air pollution studies (43-45) and in water analysis (46).

There are various methods employed in the detection of X-rays, the one most commonly used in pollution studies being energy dispersion using solid state detectors. As the name suggests the X-rays are separated according to their energies, the resolution required being obtained by the semi conductor detector made from Lithium drifted Silicon or Germanium. The spectrum is usually displayed on an oscilloscope screen of a multi-channel analyser and printed out on paper or punched on to computer tape.

2.2.1 Main Advantages

- i) It is non-destructive.
- ii) No pre-treatment of air pollution filters is necessary with the filter paper being placed directly into the instrument.
- iii) It allows for the simultaneous determination of many elements.

iv) It is extremely rapid.

v) Detection limits for most elements are in the 10^{-9} g range.

vi) Since a thin filter paper is used there is no matrix effects due to absorption or scattering of emitted X-rays.

2.2.2 Main Disadvantages

i) It requires sophisticated equipment and skilled operators.

ii) Solutions must be evaporated to obtain a solid sample.

2.3 ANODIC STRIPPING VOLTAMETRY

A comprehensive review of the principles including equations governing the various steps of this method is given by Barendrecht (47).

Basically the process is very simple, based on a reverse polarographic principle. There are three electrodes, a test electrode (hanging mercury drop or wax treated mercury-coated graphite rod), and Ag/AgCl reference electrode and a platinum counter electrode. These are inserted into a cell containing the unknown sample or standard solution. A negative potential (which may be varied) is applied to the test electrode which plates the trace metals out of solution on to the electrode based on their formal potential in the particular matrix. The potential is then reversed and a linear sweep of the required range mide, thus stripping the metals off the electrode; the order of stripping being in accordance with the characteristic half-wave potential of the elements. Plating times are usually of the order of 30 minutes and the stripping time about 20 seconds.

Since oxygen interferes in the plating nitrogen is usually bubbled through the solution before plating commenses.

Since the concentration is obtained by comparison with standards and as the amount plated obviously depends on plating time, it must be confirmed that the solution is not completely depleted of ions of interest over the plating time used.

This method is obviously very useful in the analysis of water (48-52) but has also found use in air pollution studies after dissolution of the sample (53-60) and also for biological samples (61-64).

2.3.1 Main Advantages

- i) It is very rapid. In the case of water samples results canbe obtained within 2 hours of receiving the sample.
- ii) Detection limits are in 10⁻⁹g range.
- iii) It is non-destructive in the case of water.
 - iv) It allows for the simultaneous determination of a number of elements.

2.3.2 Main Disadvantages

i) The range of elements which can be determined is limited to those which can be reduced by mercury and which have a halfwave potential less than that of hydrogen because of the "over-voltage" phenomena. However, it must be said that this limited range does include some of the important metal pollutants.

The range is: Sb, Bi, Cd, Cu, Au, ¹/_b, Hg, Ag, Tl, Zn. ii) It is highly dependent on the purity of solvents.

iii) There is the possibility of metal - metal complexes forming,e.g. Sn-Cu, Zn-Pb, etc.

II

2.4 SPARK-SOURCE MASS SPECTROMETRY

This method is based on the well known principles of mass spectrometry which are well documented elsewhere. (65).

Electrodes formed from the sample are mounted in the ion source and a spark passed between them by applying a pulsed rf voltage. Positive ions representative of the sample are produced in the discharge and are accelerated through an ion gun and passed through resolving slits. An electrostatic analyser provides energy focusing of the ion beam to compensate for the wide spread of initial kinetic energies of the ions in the spark discharge. An electromagnetic analyser then separates the ions according to their masses and focuses them on a single plane. Detection can either take place on a photoplate which effectively integrates the ions detected at each mass thereby reducing the effects of ion beam fluctuations caused by the variations inherent in the spark discharge, or electrically by collection on the first dynods of an electron multiplier and the resulting current amplified. The latter method is much faster and therefore is preferred.

This method has been used in the analysis of air filters (66-69). Spectral interference due to hydrocarbon species is removed by the use of a liquid mitrogen cold trap or a cryogenic pump. It has also been used in water analysis (70-73) and other environmental samples (74-76).

2.4.1. Main Advantages

i) It is fairly rapid.

ii) Detection limits are in 10^{-9} g range.

iii) It allows for simultaneous determination of a number of elements.

2.4.2 Main Disadvantages

- i) It is a destructive method since the sample is ground with pure graphite.
- ii) Data output is extremely complicated and requires computer processing if backlogs of data are to be avoided.

2.5 EMISSION SPECTROSCOPY

This method depends on excitation of orbital electrons of atoms either by a flame (flame emission spectroscopy) or by a d.c. arc between two electrods (d.c. arc emission spectroscopy). As the electrons return to lower energy states light of energy characteristic to the element of interest is emitted. In flame emission the sample must be presented to the instrument in solution and the emitted light usually detected by a photocell. Flame emission has been used in the analysis of atmospheric particulates (77).

In d.c. arc spectroscopy the sample is ground with pure graphite and forms one of the electrodes across which the d.c. arc is passed. The emitted light is passed through a prism and the resulting line spectrum focused and recorded on a photographic plate; the degree of darkening being proportional to the amount of element present. This is usually done by relating intensity to the known weight of an internal standard. This method has been used in the analysis of atmospheric particulates (78-86). One set of workers (86) actually used the graphite rod as a sampling filter and used this directly as the electrode. It has also been used in water analysis (87).

2.5.1 Main Advantages

- (a) Flamei) The method is fairly rapid (especially for water samples).
 - ii) In the scanning mode it allows for simultaneous analysis.
- (b) d.c. arc i) The method is fairly rapid.

ii) It allows for simultaneous analysis.

2.5.2 Main Disadvantages

- i) Both methods are destructive.
- ii) Detection limits tend to be in 10⁻⁶g Range.

2.6 GAS CHROMATOGRAPHY

. Until recently gas chromatography was used exclusively in the analysis of organic materials. However, recently more and more attention has been given to the use of this method in metal analysis, and an excellent review has been prepared by Rodriguez-Vazquez (88).

The metals are complexed to either make them volatile or to increase their volatility, using β - diketones (e.g. 2, 4 pentane - dione), acetylacetone or their tri - and hexa fluoro derivatives. The criteria required are that the complex must be volatile and thermally stable. The sample is then injected on to a suitable column of an inert stationery phase and an active liquid phase. The metals separate according to their retention times on the column, with or without temperature programming. At the end of the column the metals are detected usually by flame icrusation where ions are formed by a flame. The base of the flame is at a negative potential while the top is neutral. The ions in the flame alter the electric current, the amount of alteration being proportional to the number of ions. Many alternative detectors exist. One often employed is electron capture where the metals are excited by a radioactive source and the radiation emitted by the unstable isotope formed, measured.

This method is usually used with an internal standard for quantitive analysis.

There are several papers on the general principle of metals by gas chromatography (88-90); gas chromatography in the analysis of aerosols (91-93); in the analysis of water (94-97); and in the analysis of other environmental samples (98-106).

It is too early to discuss the advantages and disadvantages of this method in any detail. Advantages include -(i) It is fairly rapid. (ii) Detection limits are in the 10^{-9} g range at least. (iii) The method enables one instrument to be used in the analysis of both organic and inorganic pollutants thus reducing costs.

Disadvantages include - (i) The method is destructive. (ii) At the moment only & few elements have been determined simultaneously.

The two methods which are probably most widely used in environmental analysis are activation analysis and atomic absorption spectroscopy.

As these are the methods which were used in this work a deeper understanding of the theories was required in order to appreciate limitations, overcome problems which may arise during analysis, and to realise where errors will occur.

For these reasons in the following two sections a more detailed discussion is given of these two methods.

Additional reasons for using atomic absorption and neutron activation analysis are :-

- i) Atomic Absorption is comportively cheap and has wide element coverage.
- ii) Neutron activation analysis is more sensitive than atomic absorption and better element coverage, and
- iii) Both are available.

2.7 ACTIVATION ANALYSIS

The principles of this method are well documented (106). When a suitable element is bombarded with sub-atomic particles it is converted to another isotope. If this isotope is unstable it reverts to a lower energy state emitting the excess energy in the form of irradiation ($\checkmark, \beta, \forall$ rays) - the energy being characteristic of the element.

There are three main nuclear reations:-

- i) Energy is given to the nucleus by high energy photons or electrons, the excess energy being released by emission of nuclear particles (X, n) (X, p). This method has been used to analyse atmospheric particulates (107-110).
- ii) Charged atomic particles (protons, deutrons) suitably accelerated can be used as bombarding particles giving rise to emission of other nuclear particles as in (i)
 e.g., (p, ± n) (d, x n) (d, p) or incorporated into the nucleus producing a nucleus of different mass, charge and radioactive status. This has also been used for atmospheric particulate and also water samples (113-115).
- iii) Neutrons can be incorporated into the nucleus. Neutrons are uncharged and are therefore incorporated more readily than charged particles which must overcome coulombic repulsion between the nucleus and bombarding particles. Neutrons are the most widely used bombarding particles because they are more readily available mostly from nuclear reactions.

Although fast neutrons have been used to determine silicon in aerosols (115), thermal neutrons are more widely used because of their higher capture cross sectional area: they have been used for atmospheric particulates (55, 116-131), and water (132-138).

2.7.1 Interaction of Neutrons and Nuclei

Neutrons of thermal energy are obtained by moderators which slow down fission neutrons from reactor fuel. These thermal neutrons are easily captured by nuclei of most elements to give an (n, \checkmark) type of nuclear reaction. If the isotope produced by the absorption of neutrons into the nucleus is unstable the resulting radioactivity can be measured.

Factors which determine the amount of radioactivity under the influence of neutron bombardment are:-

i) the weight of element present, (W)

- ii) the fractional abundance of the isotope undergoing the (n, \mathbf{X}) reaction, (\mathbf{O})
- iii) the activational cross-section for the nuclide concerned, (σ) iv) the neutron flux, (ϕ)
 - v) the half-life with which the radioactive species decays, $\left(t_{\frac{1}{2}}\right)$ vi), the length of time of irradiation $\left(t_{\frac{1}{2}}\right)$

Radioactive atoms follow the normal law of radioactive decay

$$\mathbf{A}_{t} = \mathbf{A}_{o}e^{-\boldsymbol{\lambda}t}$$

 A_t = activity in disintegrations per unit time at time = t A_o = activity in disintegrations per unit time at time • 0 λ = radioactive decay constant which is related to halflife

$$t_{\frac{1}{2}} = \frac{0.693}{\lambda}$$

If N* is the number of radioactive rtoms formed after a time t_i seconds the rate $\frac{dN^*}{dt}$ at which the number of radioactive

atoms increase is given by $\frac{dN^*}{dt} = \phi \sigma N - \lambda N^*$ (1)

where N = number of atoms of isotope being activated present in the sample and is given by:-

$$N = (6.025 \times 10^{23}) \Theta \cdot \frac{W}{M}$$

M = mass number of the isotope being activated.

N represents the rate of formation of radioactive atoms and λ N* their rate of decay.

Integration of (1) gives the number, N*, of radioactive atoms present after time t_i .

Since N* is very small compared with N, N can be assumed to remain constant.

Integration gives: $\phi \sigma Ne^{-\lambda t} = \phi \sigma N - \lambda N^*$

Activity A_t in dps due to atom N* produced after t is given by

$$A_{t} = \lambda N^{*} = \phi \sigma N (1 - e^{-0.693t} \frac{t_{1}}{2})$$

= 6.025 x 10²³ $\phi \sigma \theta \left(\frac{W}{M}\right) (1 - e^{-0.693t} \frac{t_{1}}{2})$
When t_i becomes long compared to t₁ then $(1 - e^{-0.693t} \frac{t_{1}}{2})$

approaches unity and the saturation activity A_{∞} is obtained, i.e., for $t_i \gg t_{\frac{1}{2}}$

 $\mathbf{A}_{\mathbf{0}\mathbf{0}} = 6.025 \times 10^{23} \boldsymbol{\phi} \boldsymbol{\tau} \boldsymbol{\Theta}(\frac{W}{M})$

This is shown in Figure 2.1 and its consequences will be discussed later.

At the end of the irradiation the activity of the radioactive isotope will decay with its characteristic half-life so that at time t_d after the end of the irradiation the activity A_d will be given by:-

$$\mathbf{A}_{d} = 6.025 \times 10^{23} \phi \boldsymbol{\sigma} \boldsymbol{\theta}(\frac{W}{M}) (1 - e^{-0.693t} i/\frac{t_{1}}{2}) (1 - e^{-0.693t} d/t_{\frac{1}{2}})$$

Thus the highest activity and therefore the highest sensitivity for a given weight of element is obtained if the neutron flux, isotopic abundance and capture cross section are high.

This method is rarely if ever used because of difficulties associated with accurate determination of neutron flux and total disintegration rate. It is more common to use a comparative method by running samples and standard amounts of element together.

2.7.2 Methods of Neutron Activation Analysis

There are three main approaches:-

i) Destructive method in which the radioactive isotope of interest is separated completely from interfering active isotopes by chemical methods. It has been used in the analysis of environmental samples usually where the matrix itself is a strong &- ray emitter (139-143). Separation is achieved



FIGURE 2.1

ACTIVITY V TIME FOR RADIOACTIVE DECAY
either by passing through an ion exchange column or by adding a fixed amount of inactive carrier after irradiation. In the latter case the chemical yield is determined and hence quantitative recovery is not required.

- ii) Group separation of isotopes into sample fractions such that interfering isotopes are separated from each other. This allows for a larger number of elements to be determined than (i) and has also been used in environmental analysis (114-146).
- iii) Tristrumental X-ray spectroscopywhere the elements are separated according to energies of X-rays. This allows for the simultaneous determination of many elements but they must have reasonable natural abundance and active isotopes which decay with X-ray emission of suitable energy free from interference from other isotopes. A reasonable half-life is also required.

The development of solid state Ge (Li) detectors has made this possible for as many as 33 elements in the one sample. The theory behind this is well documented elsewhere. (147). Instrumental &-ray spectrom etry is the most widely used in environmental analysis. (55, 116-131).

2.7.3 Main Advantages

- i) Detection limits are in 10⁻⁹g range.
- ii) Simultaneous determination of a number of elements (by instrumental method) is possible.

iii) Non-destructive methods are unaffected by reagent impurities.

2.7.4 Main Disadvantages

- i) There must be a readily available reactor or other neutron source for isotopes with short half-lifes.
- ii) The instrumental method requires a long cooling period for some isotopes to allow interfering isotopes to decay.
 This will be discussed in more detail in the appropriate chapter (See Section 3.5.2.2).

2.8 ATOMIC ADSORPTION SPECTROSCOPY

This is probably the most widely used method of metal analysis because of the ease of handling of the equipment, the sensitivity and the range of elementswhich can be detected. The principles nove been well discussed (148, 149).

In brief the method can be simply defined as the absorption of radiant energy by atoms. This Δ bsorption and its quantitative correlation with the concentration of metal ions originally present in a sample solution serves as the basis of analytical atomic absorption. The atoms are produced from their compounds either by a flame or a rurnace and by absorbing radiation of a particular wavelength will reach a higher allowed state. The allowed transitions are such that spectral rules are not broken, i.e., in terms of Russell-Saunders coupling, L should change by $\frac{+}{2}$ and J by $\frac{+}{2}$ or 0. Also the multiplicity of the two spectral terms should remain the same. (Occasionally these rules are broken by heavy metals such as Hg).

This absorption of energy in practical terms means that the intensity of the light passing through the sample is reduced and **this** reduction is proportional to the concentration of the metal in the sample is solution.

The Beer-Lambert Law:

$$I_{\lambda} = I_{o}e^{-K_{\lambda} I}$$

where I_0 = the intensity of source at wavelength λ I_λ = the intensity of transmitted radiation

 k_{1} = the absorption coefficient

1 = the length of light path through sample is difficult to apply directly to atomic absorption since there is seldom a system of atoms in a steady state of homogeneous distribution. However, the principles of the Law must apply in that the intensity of unabsorbed light is a function of the negative log of the absorption coefficient.

The value of K_{λ} will vary with wavelength since the absorption line has a finite width and according to the classical theory of dispersion, the integrated absorption coefficient $\int K_{\lambda} d\lambda$ (within the limits $\lambda \rightarrow \lambda + d\lambda$) is given by

$$\int \mathbf{K} \boldsymbol{\lambda} \, \mathrm{d} \boldsymbol{\lambda} = \prod_{\mathbf{m} \mathbf{C}} \mathbf{N}_{\mathbf{\lambda}} \mathbf{f}$$

where e = electron charge

m = electron mass

C = velocity of light

- N_{λ} = number of atoms per cm³ capable of absorbing radiation of wavelength λ .
- f = oscillator strength = average number of electrons per atom which can be excited.

Thus for a transition initiated from the ground state for which N_{λ} for practical purposes is equal to N (the total number of atoms per cm³) the integrated absorption coefficient is proportional to the concentration of free atoms in the absorbing medium.

For most elements the characteristic wavelength that is used is the spectral resonance line, that is the transition from the ground or unexcited state to the lowest excited state. Since this transition involves the lowest energy it is usually the strongest line in the metal spectrum. However, the neutral atom can absorb radiation in discrete steps at different wavelengths which can be used in concentrated samples or in the case of spectral interferences from other elements.

2.8.1 Light Source

At temperatures between 2000 and 3000[°]K the width of an o absorption line is about 0.02[°]A The factors which establish the line width are:

- i) The natural width of the line due to the finite life-time of any atom in an excited state. (of the order of 10^{-4} Å).
- ii) Doppler broadening caused by the thermal motion of the emitting or absorbing atoms relative to the observer.
- iii) Pressure broadening due to collisions with neighbouring atoms, (resonance broadening if due to same kind of atoms).
- iv) Stark broadening due to external electric fields or charged particles (i.e., splitting of electronic levels of atoms).

All these effects manifest themselves depending on the source used. (ii) is the main factor if atomic vapour is produced in a low pressure furnace, (ii) and (iii) occur in flame sources and (ii), (iii) and (iv) are all associated with arc and spark excitation sources.

These factors could make it difficult to measure the integrated absorption accurately. However, by measuring the absorption coefficient at the centre of the line using a sharp

line source which emits lines with a much smaller half-width than the width of the absorption line, accurate measurements can be made provided high resolution equipment is available. This is usually obtained with a hollow cathode lamp in which the cathode is made from, or contains, material of the element to be determined; each element of interest requiring a separate lamp. Because of the sharpness of the source, minor fluctuations in output become more critical but can be controlled using double beam instruments in which the beam is chopped in two, one passing through the sample and the other not. The beams are focused after the sample, and the ratio taken. If the shape of the absorption line is determined completely by Doppler broadening then

$$K \max = \frac{2}{D} \frac{\lambda^2}{\lambda} \quad \left(\frac{\log_e^2}{\Pi} \right) \quad \frac{1}{2} \quad \frac{\Pi e^2}{m c^2} \cdot N.f$$

where K max = absorption ccefficient at the centre of the line

D = Doppler line width

and

$$D_{\lambda} = \frac{1.67\lambda_{0}}{C} \begin{bmatrix} \frac{1}{2RT}^{2} \\ Ma \end{bmatrix}^{1}$$

here	R	=	gas constant
	C	=	speæd of light
	Т	=	temperature
	λο	=	wavelength at centre of line
	Ma	=	atomic weight of absorbing species.

Hence since D_{λ} is proportional to femperature, K max does not vary significantly with small changes in temperature and the relation between absorption and concentration of the free atoms in the absorbing medium is maintained.

It practice it is extremely difficult to measure the absorption coefficient and the absorbance of the sample solution is compared to the absorbance obtained from standard solutions of known concentration of the element of interest.

2.8.2 Production of Free Atoms

There are two main methods of producing free atoms.

2.8.2.1 Flame

Here a combination of gases is burned in the light path. The sample solution is aapirated into the flame where the solvent is dried off to form a clotlet. The clotlet is thermally dissociated into free atoms and finally the free atoms absorb radiant energy. The whole process takes a matter of seconds.

The main types of flames used are:-

- (a) Air/Acetylene with a temperature of approximately 2300°C and is the preferred flame for about thirty-five elements.
- (b) Nitrous oxide/Acetylene with a maximum temperature of 2900°C and is used for determination of elements which form refractory oxides, e.g. aluminium.
- (c) Argon/Hydrogen with an extremely low temperature of 300-800°C using diffused air as an oxidant shows considerably less absorption in the far U V (190-220 nm) than other flames and therefore is useful for the determination of As (194 nm) and Se (196 nm). Because of the low temperature, chemical and

matrix interferences may be encountered and the use of exactly matched standards or the method of Standard Additions is strongly recommended.

(d) Air/Hydrogen with a temperature of about 2000°C is useful for the determination of alkali metals (Cs, Rb, K, Na) as the lower temperature considerably reduces ionization.

2.8.2.2 Flameless Atomisation

As the title suggests the flame is replaced either by a carbon furnace or a tantalum filament with maximum temperatures of about 2600° C and 3200° C respectively.

The improvement in sensitivity in many cases allows for direct analysis of a solution where previously pre-concentration was necessary. With suitable matrices solid samples can also be measured directly. Detection limits are also increased to 10^{-13} g range.

In the flameless system the stages which the sample goes through (drying, etc.) can be controlled electrically. The solution is first dried at a temperature just below the boiling point to prevent spluttering and then the temperature is increased (stepwise or programmed) to a value at which the accompanying material will be thermally destroyed without causing a loss of the element of interest. Finally the temperature is very rapidly increased so that the element will be thermally dissociated into its atoms. If residues remain after the atomisation they may be removed by heating at maximum temperature.

In flames interferences can be minimised by preparing the solutions in a chloride medium thus helping to overcome the most prevalent type of chemical interferences, i.e., formation of stable oxides or mixed oxides. In a furnace (or tantalum ribbon) however, it appears that there are fewer oxide type interferences and that the hot carbon surface provides an efficient reducing surface. Interferences are clearly reduced in fact when solutions of salts such as nitrates or sulphates which break down thermally via the oxide, are used (150) (in complete contrast to the flame). Chloride solutions appear to result in the volatilisation of molecular chlorides of many elements and these are not subsequently atomised in the vapour With many metals the use of nitrate solutions produces phase. metal oxides which remain on the surface of the atomiser until they are reduced to the metal and released as metal atoms. It has been suggested (151) that this reaction is

 $MO(s) + C(s) \longrightarrow M(g) + CO(g)$

When thermodynamic data is used to calculate the lowest terperature at which this reaction is favourable for suitable elements the results correlate well with the temperature at which the same elements appear in atomic form in the graphite furnace (152). This correlation can only be considered valid if it is assumed that when the appropriate temperature is reached, the rate of reaction of metal oxide with carbon is fast so that there is no delay in the appearance of metal atoms.

In a study on the kinetics of copper in the carbon furnace (153) the following was suggested:-

$$Cu_{2}0 + C \xrightarrow{\text{slow}} 2 Cu (S/1) + CO(g)$$

$$Cu (S/1) \xrightarrow{\text{fast}} Cu (g)$$

where S, land g represent the solid, liquid and gas phases respectively.

This theory tends to break down with elements which form stable carbides, e.g. Al, Cu, Cr, Mo, Ti, V.

2.8.3 Interferences

There are five main types:-

2.8.3.1 Chemical Interferences

These occur when the element of interest combines with some other cation or anion in solution to form a compound which influences the degree of reduction to free atoms, i.e., reducing the number of atoms capable of absorbing radiation at the resonant energy.

Examples are - the reduction of the calcium signal in the presence of phosphate and the reduction of the magnesium signal

Chemical interferences can be overcome or controlled in two ways -

- i) Use of a higher temperature flame which produces additional energy to break down the compound formed.
- ii) By addition of a releasing agent, that is, a chemical species which when added to the sample solution will preferentially react with the element of interest or the interferent.

2.8.3.2 Matrix Interferences

These occur when the physical characteristics of the sample and the standards differ considerably, e.g., when the sample solution contains high concentrations of dissolved salts or acids (i.e., a different viscosity which alters the acriation rate) or when the sample and the standards are at a different temperature.

Matrix effects can be controlled by

- i) Dilution of the sample until the effect of dissolved salts or acids become negligible.
- ii) Matching sample and standard solutions for major constituents.
- iii) Use of the method of Standard Additions. Three aliquots of the sample are taken. One is diluted to a known volume with solvent. The second and third aliquots are diluted to the same volume with suitable quantities of known standards added so that the final solutions contain different additions of the metal to be determined. The absorbance is measured for each solution and plotted against the added concentration. The resulting straight line is extrapolated through zero absorbance and the intercept on the concentration axis gives the concentration of the metal in the diluted sample solution. For an accurate determination, the working curve must be linear over the concentration range covered by the sample plus additions and additions should be of approximately the same concentration as that anticipated for the diluted sample solution.

2.8.3.3 Ionisation Interferences

These occur when the flame temperature is sufficiently high to generate the removal of an electron from a neutral atom giving a positively charged ion and reducing the number of metal atoms in the flame.

The effect can be controlled by -

- i) Using a cooler flame which will have less energy available to produce ionisation.
- ii) The addition to both sample and standard of a large excess of an easily ionised element (e.g., alkali metals) which works due to a form of mass action effect, i.e., as the amount of an easily ionised element in a flame is increased the proportion of ionised to unionised atoms of the element of interest is decreased with a corresponding rise in the number of atoms in the atomic ground state.

For example in the analysis of Na, 1000-2000 ppm K is added to sample and standards

Na X
$$\stackrel{\Delta}{\longrightarrow}$$
 Na^o + X^o
Na⁺ + e⁻₁
KX $\stackrel{\Delta}{\longrightarrow}$ K^o + X^o
K⁺ + e⁻₂
e⁻_{1,2} + Na⁺ \rightleftharpoons Na^o

2.8.3.4 Spectral Interferences

Spectral interferences can occur when an absorbing wavelength of an element present in the sample but not being detected falls within the width of the absorption line of the element of interest giving an erroneously high result.

This can be overcome by

i) Use of a smaller spectral slit width.

ii) Using an alternative wavelength of the element of interest.

2.8.3.5 Background Absorption

Background absorption is the collective term used to describe the combined effects of flame absorption, molecular absorption and light scattering.

- i) Flame absorption which becomes increasingly significant below 250 nm may be controlled by properly set fuel and oxidant flow rates or by using a flame with less absorption at the wavelength of interest.
- ii) Molecular absorption can often be controlled by using a higher temperature flame with sufficient energy to break down the absorbing molecular species.
- iii) Light scatter is a term used for the scattering of light by solid particles in the flame and which erroneously increases the amount of light which appears to have been absorbed.

These three interferences can also be removed together in two ways -

 i) Since molecular absorption, flame absorption, and light scatter take place over a wider wavelength than the atomic absorption then by measuring the absorbance at a

near-by non-absorbing wavelength of the element a measure of the absorbance due to background can be assessed and subtracted from the absorbance of the element wavelength. This second line, however, to be effective must be within 1 or 2 nm because of the variation of molecular absorbance with wavelength.

ii) A better method is to use a deuterium background corrector. This is a continuous source which is time shared along with the hollow cathode lamp. Both beams are passed through the flame and a ratio taken. (The line absorption of the continuum source is negligible compared to the background over the bandwidth of the monochromater). However, above about 350 nm the continuum source is weak and the adjacent line correction must be used.

Interferences can also be removed in many cases by selectively extracting the metal (s) of interest into organic solvents by controlling the pH. e.g., metals can be complexed with ammonium pyrrolidine dithiocarbamate (APDC) and then extracted into methyl iso-butyl ketone (MIBK).

2.8.4 Main Advantages

- i) The method is fairly rapid depending on the number of elements of interest.
- ii) Detection limits in the flame are in the 10^{-6} g range and with the carbon atomiser are in the 10^{-13} g range.
- iii) Equipment is relatively easy to handle with no long training period required.

2.8.5 Main Disadvantages

i) By its very method of detection this method is destructive.ii) It does not allow for simultaneous detection of a number of elements, although a number of elements can be determined on the one sample, the limiting factor being the amount of sample.

iii) Samples must be brought into solution before analysis.

This method is best suited to the analysis of a number of samples in a batch because of the time involved in setting up equipment and fresh standards.

Atomic absorption has been used as an analytical technique in air pollution studies, both with flame atomisers (154-167) and flameless systems (168-177). Several workers devised systems whereby the air was drawn through the carbon rod which was subsequently placed directly into the spectrophometer. (170-176). Atomic absorption has also been used in the analysis of water, again using flame (178-182) and flameless systems. (183-187).

More recent advances in atomic absorption include the hydride system in which the hydrides of Se, Sb and As are produced chemically in an enclosed system, concentrated in a cold trap and aspirated directly into the flame. This removes matrix interferences and also as aspiration in the flame is 100% since it is in the form of a gas compared with 10% for solutions, the detection limits are greatly improved. (1000 times in the case of Se). There is also now available a mercury system, similar to the above, mercury is formed chemically and the gas pumped round an enclosed system using a quartz cell in place of the flame.

;

AIR POLLUTION

3.1 INTRODUCTION

It is difficult to judge when air pollution was first considered to be a problem. Recent years have seen a large increase in the concern for metal and/or element levels in the atmosphere as illustrated in Chapter I. However, in England as long ago as 1377, Richard III, and Henry V in 1413, regulated and restricted the use of coal in London because of the health hazards of smoke and odours.

The reason for the increased concern is two-fold -

- i) The population growth of recent years had led to a larger demand for manufactured products and electrical power, increasing even low levels of pollutant output to a major problem.
- ii) The enormous increase in the use of motor vehicles which has been shown to elevate lead levels in roadside vegetation and insects (188-193).

Cause for Concern

There have been numerous episodes in the past of air pollution resulting in illness and death, several examples of which are listed below.

- i) Glasgow 1909 During two periods of foggy conditions in the Autumn 1,063 deaths were attributed to the noxious conditions and the smoke (194).
- ii) Meuse Valley, Belgium 1930 which is 15 miles long and
 300 feet deep A thermal inversion confined emitted
 pollutants from the large number of industrial plants,

killing 60 people and causing the death of many cattle (194-195).

- iii) Donora, Pennsylvanie, U.S.A. 1948 This was the site of three large industrial plants in a U-shaped valley. A thermal inversion combined with foggy weather caused 43% of the 14,000 population to be ill and killed 20 people. A subsequent study carried out in 1952 and also in 1957 found
 that the exposed population had a less favourable mortality rate than those not exposed in 1948. This at least indicates the possibility of long term effects from breathing polluted air (195).
 - iv) Poza Rica, Mexico 1950 Hydrogen sulphide accidentally released from a sulphur recovery plant during foggy weather with weak winds and low inversion layer hospitalised 320 people of whom 22 died (194-195).
 - v) London, England 1952 lying in the broad Thames valley -During a four day period of fog and temperature inversion between 3,500 and 4,000 excess deaths were recorded. The mortality rate remained elevated for several weeks after the weather improved, again implying that acutely increased pollution may start a process which could continue to operate even after conditions have returned to normal.

This latter incident caused records to be checked and further relationships between mortality and bad weather to be sought. There were excess deaths in the Winters of 1948, 1891, 1882, 1880, 1873, 1862 and 1859 (194-195).

 vi) New York City, U.S.A. 1953 - Increased pollutant levels due to widespread atmospheric stagnation caused an excess of deaths compared to average (195).

The following points should be emphasised from the above incidents -

- i) Most of the deaths occurred in people who were known to have heart and lung conditions and also in the old and the very young. If, however, these are the people most at risk then safety standards ought to take this into account and be set below their threshold and not based on that of the average person.
- ii) Pollution incidents occurred during periods of either thermal inversion, low wind speed or fog, all of which cause the pollutants to remain concentrated round their source for longer periods than normal.
- iii) In most cases SO₂ and particulate matter were shown to be involved.
 - iv) There is evidence to suggest that the effects of high pollutant levels remained long after the pollutant levels returned to normal.
 - v) There is a need for monitoring levels of pollutants and being able to predict how levels will vary under differing climatic conditions.

It is this final point which forms the basis of the work in Section 3.5 of this Chapter.

Of the two pollutants which have caused most problems (SO₂ and particulate matter) it was decided that a study of particulate matter would be more profitable as the SO₂ levels in Glasgow are already well documented by the Smoke Officer of the Environmental Health Department.

3.2 PARTICULATE MATTER

The terms, atmospheric particulate and atmospheric aerosol, are taken in this work to be synonymous, although strictly speaking an aerosol can contain either solid or liquid particles whereas particulate matter contains only solid particles. The amount of particulate matter in the atmosphere is important for several reasons -

- i) As chemical species <u>per se</u> some constituents exhibit high orders of toxicity to plants and animals.
- ii) As physical entities they scatter and absorb light. It has been estimated that as much as 20% of the energy available to heat the earth-atmosphere system from the sun can be lost in some cases due to back-scatter of the solar beam. Until recently both Pittsburgh and St. Louis (U.S.A.) had palls of smoke which required the street lights to be lit during the day (195).
- iii) As charged particles they can lead to condensation and coalescence of other particles and gases. This has also been shown to cause an increase in the amount of rainfall due to the increased number of condensation nuclei, i.e. affect climate (196-198). Although it must be said in a similar study (199) no evidence for this was found despite the production of sufficient condensation nuclei.
 - iv) They are suppliers of large specific surface areas for absorption of pollutants. This can have the effect of retaining harmful substances in the respiratory tract and lungs for longer periods and also of enhancing the penetration of these substances deeper into the lungs.

4I

It is also known that air polluted with SO_2 and particles of charcoal produce more rapid corrosion of metal surfaces than SO_2 alone.

v) They afford an opportunity for catalysis of normally slow reactions to take place. For example, in the presence of Mn, Fe or V the production of sulphuric and nitric acids from their oxides in the atmospheric aerosol is greatly accelerated.

iv) and v) above combine to give a chemical breeding groundeither for more or less noxious products.

- vi) They can land on the surface of leaves and interfere with light interception, respiration and transpiration, leading to stunted growth. In fact it has been suggested that particles could actually prevent the stomata closing properly thus allowing gaseous pollutants more ready access (200).
- vii) They soil clothes and buildings. Although this is the most obvious effect of particulate pollution, and although offensive to the eye it is the least dangerous.

As would be expected the Smoke Officer has comprehensive records on the total particulate levels of the city of Glasgow's atmosphere and to add new information it is therefore the actual constituents of the particulate matter and their concentrations which have to be studied.

3.2.1 Metals in Particulate Matter

Metals are important constituents of particulate matter for the following reasons -

- i) It is mainly metals which are the catalysts in atmospheric reactions.
- ii) There are many known emitters of metals iron and steel
 works, electrical power stations, coal and oil burners,
 smelter works, car exhausts to name but a few.
- iii) Several metals exhibit low thresholds of toxicity, e.g.Pb, Cd, Ni, Sb, Be and Hg.
 - iv) Biochemically, metals can interfere with enzymic reactions both in animals and plants (201-207).

3.2.1.1 Choice of Metals

The choice of which metals to include in the study is a difficult one. Obviously the choice is limited to those which can be detected by the methods to be used, but even within this range a choice has to be made.

These metals which are highly toxic (see above) are of course a first priority, but the answer is not so simple, as metals can alter the biological activity of other metals or other pollutants. They can do this in four ways -

i) Potentiation - There is an enhanced response of two or more substances over and above that of additive or predicted effects and in which one substance is inactive or by itself produces a different effect, e.g. a NaCl aerosol which is normally inactive enhances the irritation caused by SO₂ (195).

- ii) Addition The effects of more than one pollutant are the sum of the individual effects.
- iii) Synergism The effect of more than one pollutant is greater than the individual effects, e.g. Pb increases the toxicity of As and vice-versa. Cu has a similar effect on Hg(1), Co increases the detoxification time of CO (208) and O₂ increases the toxicity of Pb (209).
- iv) Antagonism Pollutants work against each other to reduce the effect, e.g. As blocks the toxic action of Se (1). Several other examples are cited elsewhere (210-214).

Since there is not sufficient data available to catalogue all the effects of pollutants on each other, metals cannot therefore be chosen solely on this basis with any degree of confidence.

Another complicating factor is that certain metals have very low toxicities but are extremely valuable in determining the source of other metals, e.g. Al has been used in the calculation of enrichment factors normalised to soil. (See Section 3.5.3.3).

The safest approach would therefore be to make as few <u>a priori</u> judgements as possible about the importance of individual metals and to measure as many as possible. This was the approach adopted here.

Where neutron activation analysis is used several nonmetallic elements can be easily determined, and based on the above reasoning should also be measured (e.g. Br in Section 3.5.2.1).

3.3 SAMPLE COLLECTION

Sample collection is one of the most important considerations for the analyst. The factors which can affect the accuracy and meaning of results are -

i) size of sample (i.e. volume of air)

ii) rate of sampling

iii) duration of sampling

iv) collecting medium

v) location.

It will become apparent that the choice of one of these in many cases automatically fixes the condition of the remainder.

3.3.1 Size of Sample

The quantity of air required must be known. This requires knowledge of the minimum pollutant concentration of interest as well as the sensitivity of the analytical technique to be employed. When information is not available the amount of air required must be determined by trial and error.

3.3.2 Rate of Sampling

The permissible rate of sampling is determined by the collection device selected and is dependent on the allowable head loss as well as the experimentally determined optimum flow rate, i.e. that flow rate permitting highest efficiency of collection. For example, a unit capable of efficient operation at high flow rates is essential where a short sample duration is desirable. Units which separate particles, e.g. a cascade impactor must be run at the quoted rate.

3.3.3 Duration of Sampling

The duration is selected which will provide the information desired. For peak concentrations and their causes, as short a sampling period as possible should be used. For background or average values weekly or monthly sampling is sufficient.

Any sampling period will only indicate the average concentration during that time period. Figure 3.1 illustrates the daily values of Se concentrations for May 1973 and the weekly and monthly averages calculated from these. The value of the average concentration for the whole year is given for comparison.

There is of course a **limiting** condition where very short sample duration results in a rapidly fluctuating record which may be difficult to read and perhaps uneconomical.

3.3.4 Height of Sampling

The choice of sampling height depends on the information required. If human exposure is to be measured then the ideal sampling height is at the height of the average person. If transport or area differences are being measured then rooftop or sampling towers are appropriate thus avoiding the pick-up of re-suspended particles.

Ambient sampling is usually at 5-15 metres. Table 3.1 illustrates the concentration difference at 5 and 52 metres for Zn, Fe, Mn and Cr at one site (Glasgow University). The results also illustrate the necessity of maintaining a constant sampling height for studying different areas.



	TABLE 3.1	VARIATION IN ELEMENTAL CO (ATMOSPHERIC VARPICULATES	NCENTRATION WIT	THNE (ng/M ³)
SAMPLE NO	<u>5 Metres</u>	Zinc 52 Metres	<u>]</u> 5 Metres	Iron 52 Metres
١	121.3	202.5	156.0	429.7
2	36.0	59.6	210.0	256.0
3	90.0	158.1	118.0	737.0
4	31.4	53.4	140.0	685.9
5	27.1	294.1	1.37.0	223.7
6	43.4	143.9	86.0	584.9
ר'	280.6	14.4	364.5	240.1
¢	240.5	343.6	284.0	875.3
٩	153.8	539.2	114.0	1026.2
10	307.9	282.3	247.5	875.3

Manganese			Chromium		
	5 Metres	<u>52 Metres</u>	5 Metres	<u>52 Metres</u>	
١	13.7	34.6	4.0	13.7	
z	6.6	9.9	2.8	5.8	
ઝે	7.5	20.5	2.3	13.5	
4	6.6	14.9	2.6	9.4	
5	8.2	6.1 `	1.8	8.9	
ŀ;	7.4	21.5	2.0	24.7	
· .7	22.7	19.7	8.7	5.0	
8	15.1	18.3	3.3	19.3	
٩	11.3	29.9	1.0	41.9	
10	7.8	13.6	1.6	10.2	

•

•

.

3.3.5 Location of Sampling

Again the location depends on what is required from the results. If a suspected source is being examined then the samples must be placed upwind of the source. If, however, ambient background levels are required then the samples must be placed such that it is representative of the area and that it is not adversely affected by a near-by pollution source. Values for different parts of Glasgow will be discussed later (see Section 3.6).

3.3.6 Collection Medium

The collection device must be suitable for the specific pollutant involved. It is not necessary that the collection be 100% efficient but it must be known and reproducible. An efficiency below about 75%, however, is generally not suitable for air pollution sampling (215).

Several collection devices have been used in air metal pollution studies.

In some cases natural collectors have been used. For example, some types of moss (e.g. Spanish Moss) are epiphytes, i.e. they derive all their nutrients from air and can therefore be used as indicators of metal levels in the surrounding air (216). Both moss growing naturally (217, 218) and suspended in nylon bags (219, 220) have been used.

In countries of heavy snowfall the snow has been used as a collector (221,222). The Pole snow profile experiments are

of course well known (3, 223, 224) but snow has also been used as an indicator around factories (225, 226) and roads (227-229) and the effects of draining and dumping into rivers and lakes investigated (227, 230).

The annual growth of tree rings has been shown to be useful in gaining historical information on metal levels (231, 232). However, in a recent review article (235) this method was critically assessed in terms of metal uptake through roots and leaves and the possibility of metal movement within the rings.

City pigeons when compared to their country counterparts were found to have much higher lead contents in their hard tissues (234), and it was guggested that these might serve as indicators of pollution levels.

Finally, utilising the fact that due to the hot climate in Turin most houses have air conditioning units, it was suggested that the air filters in these units could be used as dust collectors (235).

The most common collection device is an air filter, i.e. a porous bed through which the air must pass. The filter materials most frequently used are paper, glass fibre and membrane.

<u>Paper Filters</u> (e.g. Whatman's) - These are mats of randomly orientated cellulose filters which force the air to change direction many times in an abrupt fashion so that inertial forces or direct collisions will bring the particles into contact with a relatively large surface area of the filter. These filters depend heavily on impaction to obtain high efficiencies for removal of small particles. (0.3 µm or less).

<u>Glass Fibre Filters</u> - These are similar to paper except that they are made of glass fibre rather than cellulose. They achieve high efficiencies without requiring impaction velocities due to the electrostatic nature of the matrix which tends to attract particles from the air stream.

<u>Membrane Filters</u> - The method of manufacture of these filters permits a uniformly controlled pore size (0.05-10 µm commonly used in air pollution). The pores constitute 80-85% of the filter volume and are made of esters of cellulose (especially cellulose acetate). These filters also tend to be electrostatic. Due to this, and by the building up of a precoat of collected particles on the surface these filters can collect particles much smaller than the pore size would suggest, e.g. 0.45 µm can collect particles down to about 0.1µm.

These filters are also well suited for particle sizing since most of the particles are retained at or near the surface.

3.3.6.1 Choice of Filter Material

The choice of filtering material depends on what is required from the sample. Because of their high flow rates paper and glass fibre are useful for total particulate measurements, but tend to be hygroscopic and require equilibrating at constant humidity before weighing. Membrane filters on the other hand are non-hygroscopic and have extremely high collection efficiencics but also have a large resistance to air flow thus reducing the permitted sampling rate and consequently increasing the sampling time.

5I

Another important factor when choosing a filter for elemental analysis is the impurity level in the filter paper itself. Table 3.2 illustrates the levels of impurity in glass fibre (GA - 6) Whatman (No. 1) and membrane (E.H.W.P.) as estimated by neutron activation analysis. It can be clearly seen that glass fibre is useless for analysis by this method due to the high Na content alone which masks other peaks. It can also be seen that membrane has the lowest impurity levels of all elements measured except therein.

Considering all the above factors it was decided that membrane filters were the best choice for this work.

The advantages are -

i) low background levels of impurities

ii) non-hydroscopic nature enabling direct weighing

- iii) extremely high collection efficiencies (98-99% down to an aerosol diameter of at least 0.3µm).
 The disadvantages are -
 - i) They tend to be brittle and electrostatic especially after irradiation. This, however, was found to be less of a problem after experience and practice in handling the filters was obtained.
 - ii) The high collection efficiencies mean that loading and clogging takes place more rapidly than with other types of filters. This, as said previously, limits the rate of sampling and increases the minimum sampling time required and also of course decreases the maximum sampling time allowed.

Element	Millipore EHWP	Glass Fibre GA-6	Whatmans No. 1
Br	8.65	·+	ND
Co	0.29	1.21	0.72
Cr	20.04	35.12	2.97
Fe	376.6	1.9 *	0.31 *
Mn	4.72	+	5.80
Na	496.8	12.53 *	3. 02 *
Sb	0.20	14.5	0.54
Sc	0.008	1.49	0.02
Se	8.1	ND	9.07
Zn	10.20	94.9	11.58

+ = not determined due to high sodium activity

 $* = mg/cm^2$

ND = not detected.

MILLIPORE EHWP = 2.5em. diameter GLASS FIBRE CA-6 = 15em. diameter WHATMANS No.1 = 9em.

For Cu- see page 56 Table 3.4

53

TABLE 3.2 CONC

CONCENTRATIONS OF ELEMENTS IN FILTERS (ng/cm²) (UNLESS OTHERWISE STATED)

3.4 <u>REPRODUCIBILITY OF FILTER PAPER BLANKS</u>

The filter paper blanks for millipore in Table 3.2 were those obtained from the first box of 100 used by choosing five filter papers at random. A similar blank was carried out on the succeeding boxes as they were used. The results are given in Table 3.3 and indicate that inconsistencies exist from box to box. To determine whether these inconsistencies exist within one box the following experiment was carried out.

The boxes of 100 filters were separated into four lots of 25. In one of these lots the filters were divided into 5 sets of 5 consecutively from the top. In the remaining three, 5 filters were selected at random from each omitting the top and bottom 3 filters.

This process was repeated with another box.

The samples were analysed by atomic absorption for 9 elements. To give all the results was considered unnecessary at this stage since all nine elements showed the same pattern as depicted by Cu in Table 3.4 which was -

- i) The first and last 5 filters had higher blank values than the inside filters (x20 in the case of Zinc). This indicated the wisdom of rejecting the first and last 3 filters when collecting samples.
- ii) Apart from these filters the blank values within a lot of25 were reproducible.
- iii) Filter papers within one box of 100 had reproducible blank values.
 - iv) Filter paper blanks varied from box to box.

TABLE 3.3	VARIATION IN I	MILLIPORE BLAN	K VALUES WITH	$BOX (ng/cm^2)$
Element	Box 1	Box 2	Box 3	<u>Box 4</u>
Br	8.6	7.9	9.3	8.4
Co	0.29	0.41	1.1	0.65
Cr	20.04	18.68	14.13	10.51
Fe	376	97. 8	407	133
Mn	4.72	4.95	4.0	4.28
Na	497	99.2	489	503
Sb	0.2	6.68	14.26	10.38
Sc	0.01	0.03	0.06	0.04
Zn	10.2	9.2	3 9.5	5.5
Se	8.1	6.2	6.5	7.4

•

•

.

.

.

VARIATION IN BLANK VALUES OF COPPER IN MILLIPORE FILTER PAPER (ng/cm²) TABLE 3.4



The average value obtained for each box with the standard

deviations are given in Table 3.5 along with the values obtained from a random sample of 5 filter papers for Perces land 2. used for sample collection.

The small variations which do exist within one box are so small compared to the weight of element collected as to be insignificant.
TABLE 3.5	AVERAGE BLAN	K VALUES IN MIL	LIPORE FILTER :	PAPER (ng/cm ²)
<u>Element</u>	Box 1	Box 2	<u>Used 1</u>	Used 2
Al	31.8 - 6.9	15.5 - 1.42	73.7	51.4
Cđ	0.64+0.05	0.5 + 0.22	1.5	0.3
Cr	11.8 ⁺ 0.62	9.6 - 0.19	26. C	20.3
Cu	21.1 - 1.04	50.5 - 0.92	30.1	30.6
Fe	115.4-4.7	133.7 - 6.1	92.6	140.8
Mn	4.0 + 0.33	8 .5 +0.50	5.5	4.9
Ni	18.2 - 2.07	26.1 <mark>-</mark> 1.12	27.5	12.8
Pb	10.0 - 1.33	5.0 + 0.37	3.8	1.3
Zn	5.5 ⁺ 0.05	44.7 - 14.6	10.9	32.6

3.5 ATMOSPHERIC LEVELS OF TRACE ELEMENTS AND THEIR VARIATION WITH WEATHER

3.5.1 Sampling

As explained in the previous Section the sampling time is dependent on what is required from the results. In this case where variations in element levels were being examined and the effect of climate investigated daily samples would be ideal since the weather centre observations are on a 24 hour basis.

To summarise the previous section the factors which must be taken into account when sampling are -

- i) Enough sample must be collected in order that the element can be measured.
- ii) The sample must not be so large that clogging of the filter occurs.
- iii) The sampling rate must be such as to take account of the filter's resistance to flow.

By trial and error it was found that $20-30M^3$ of air were required for analysis and that this amount did not clog the filter. However, it was found that collecting this volume over 24 hours led to a large pressure drop across the filter with time. It was therefore necessary to collect over 48 hours. Figure 3.2 illustrates the variation of volume with time which indicates a uniform sampling rate and no clogging over this 48 hour period.

For practical reasons three samples a week were collected. Sampling was started at 9.00 a.m. on Monday morning to coincide with the weather centre data and changed at the same time on Wednesday and again on Friday. This meant that the week-end sample was running for 72 hours but this did not clog the filter. SAMPLING RATE OF MILLIPORE FILTERS WITH CHARLES AUSTIN M 361 PUMP



Figure 3.3 illustrates the apparatus used to collect the sample and is similar to that used by Warren Springs Laboratory (236). The pump was a Charles Austin model M 361 and was designed for continuous running. The sampling funnel was situated about 5 metres above ground level outside the Chemistry Department of Glasgow University approximately 50 metres from the nearest main road, USING a 2.5 cm. diameter membrane filters (E.H.W.P)

After collection, the filter was heat sealed in clean polythene sheeting.

3.5.2 Analysis

The samples were analysed by neutron activation analysis followed by \mathbf{X} -ray spectroscopy.

This method is suitable for air pollution filters because -

- i) Carbon, nitrogen and oxygen, the major constituents in air particulates, have low neutron capture cross sections and hence do not exert matrix effects.
- ii) Samples are thin and therefore there is little self absorptionof X-rays and shielding of neutrons.

3.5.2.1 Irradiations

Two irradiations of these samples were required. The first was for 15 minutes at a thermal mentron flux of $3 \ge 10^{12} \text{ n/cm}^2/\text{ sec.}$ at the Scottish Reactor Centre at East Kilbride for the determination of Na, Mn and Br. The second was for 96 hours at $6 \ge 10^{12} \text{ n/cm}^2/\text{ sec.}$ in the Dido Reactor, A.E.R.E., Harwell, England, for the other elements measured. For both irradiations standards were prepared by dissolving spectroscopically pure



metals or metal oxides in suitable acids and placing 10 µl aliquots on filter discs. Each standard was prepared in triplicate and corresponded to 2-20 µg.

Standard kale whose elemental concentration is known (237) was irradiated along with the samples and standards to check the accuracy of the standards.

3.5.2.2 Counting

Table 3.6 presents the nuclear data for the elements determined (238). 1 hour after the first irradiation the samples and standards were counted with a 60 cm³ detector (Nuclear Enterprises 2.2 KeV resolution at 1332 KeV; peak to compton ratio 24 : 1) in conjunction with a Nuclear Data 4400 system coupled to a Nuclear Data N D 812 computer. The dust samples were counted for 15 - 20 minutes for Na, Mn and Br and the standards for 1 minute. Figure 3.4 is a typical χ^{-} -ray spectrum obtained. The areas under the χ -ray peaks were calculated by the method described by Covell (239).

The longer lived isotopes produced during the 96 hour irradiation were counted 10 days and 30 days after the end of the irradiation.

These dust samples required count times of 1 - 2 hours and the standards 1 - 5 minutes. Figure 3.5 is a spectrum obtained after 10 days decay.

In an attempt to reduce manual handling the raw data from the spectrometer was punched on paper tape for processing by a KDF9 computer and peak areas, standard deviations and energy calibrations calculated (240). This, however, proved to be unsuccessful as mistakes were invariably made in punching and

		Thermal neutron capture		
		cross-section of parent	Tsotonic	Energy of
<u>Isotopes</u>	Half-life	(Barns	Abundance (%)	$\frac{\Delta^{-1ay}}{Meausred}$ (KeV)
24 _{Na}	15h	0.40	100	1368
46 _{Sc}	88.8d	25 .	100	889 , 1120
51 _{Cr}	27.8d	16	4.3	320
56 _{Mn}	2.58h	13.3	100	847
59 _{Fe}	45.lh	1.23	0.33	1099, 1291
60 _{Co}	5.27Y	, 17	100	1173, 1332
65 _{Zn}	243.6d	0.82	48.9	1115
75 _{Se}	120.4d	55	0.87	265, 279
82 _{Br}	35.5h	0.26	49.5	554, 777
110m _{Ag}	253d	4.2	48.2	937
124 _{Sb}	60.3d	3.4	42.7	603 , 1691
134 _{Cs}	2.05Y	27.4	100	796
140 _{IA}	40.2h	8.8	99.9	487
169 _{Yb}	32d	3200	0.14	308
177 _{Lu}	6.7d	2050	2.6	208
181 _{Hf}	42.4d	12.6	35-2	482
233 _{Pa}	27d	7.4	100	3 12
from 232	lp			

•

TABLE 3.6 NUCLEAR DATA FOR ISOTOPES MEASURED BY NEUTRON ACTIVATION ANALYSIS

•



FIGURE 3.4



FIGURE 3.5

the machine had a tendency to stick. Also small photopeaks located on the tails of neighbouring large peaks had still to be processed manually. For these reasons the programme was abandoned and the results calculated by hand.

3.5.2.3 Interferences

In instrumental neutron activation analysis where no chemical separation is carried out, interfering isotopes are removed simply by their decay or by ensuring that the energy of the reactor is such that they cannot be activated.

The first method is used when an 'interfering' isotope has a much shorter half-life than the isotope of interest. In Figure 3.5 it can be seen that ⁸²Br is still masking some peaks but after 30 days these Br peaks have virtually completely decayed, allowing the other isotopes to be measured, e.g. 554 KeV ⁸²Br masking 564 KeV ¹²²Sb.

The second method is used for isotopes with high neutron capture cross sections, e.g. ^{24}Na , ^{56}Mn and ^{82}Br .

However, despite these manipulations sometimes the most prominent photopeak of an isotope cannot be used because of interferences by neighbouring photopeaks of other isotopes, e.g. 844 KeV ²⁷Mg on 846 KeV. ⁵⁶Mn, and 559 KeV ⁷⁶As on 555KeV ⁸²Br and 564 KeV ¹²²Sb. The mono-energetic ⁸⁰³Hg (279.1 KeV) is interferred by ⁷⁵Se (279.6 KeV) and vice-versa. For Hg a correction based on the pure ⁷⁵Se spectrum is applied and for Se determinations the alternative 265 KeV photopeak is used. Similarly there is a mutual interference effect of 65 Zn (1115 KeV) and 46 Sc (1120 KeV). Again the alternative 46 Sc 889 KeV peak is used and for Zn a correction based on a pure 46 Sc spectrum is employed.

3.5.2.4 Limitations

From Table 3.6 it can be seen that Mn with a half-life of 2.58 hours is the shortest half-life used. This is because the minimum delay time between the end of irradiation and the counting of the sample is 45 minutes and represents the travelling time from the Reactor Centre and Glasgow University. This means that such elements as Cu, As, V, S and Al all with half-lives less than 10 minutes could not be measured.

Similarly with the longer irradiation period, because of the delay in getting the samples back to Glasgow (4-5 days) the shortest half-life usable was that of 140 La (40.2 hours) and even this could only be determined on the first five or six in each batch before the activity was lost.

Remembering that saturation activity is reached at irradiation times greater than or equal to seven times the halflife (see Section 2.6) then activities cannot be increased to last longer for counting simply by increasing the length of time in the reactor and must be accepted as a limiting factor in this method when a nuclear reactor is not immediately at hand.

Because of this the 45 elements which can be determined under ideal conditions are reduced to the 17 elements listed in Table 3.6.

3.5.2.5 Accuracy

To test the accuracy of the method three standard filter papers obtained from the International Atomic Energy Agency in Vienna were analysed. The results are given in Table 3.7 and the "true" value and range claimed by the Agency in Table 3.8 (241). As can be seen from these two Tables the method is giving a true indication of the atmospheric levels of these elements.

From counting statistics and reproducibility of replicate analysis the accuracy of the results is estimated as follows -

Br,	Coy	La,	Mn,	Na,	Sb,	Sc,	Zn	-	10%
Cr,	Fe,	Se,	Th					-	20%
Ag,	Cs,	H£,	Lu,	Yb				-	30%.

TABLE 3.7	EI	EMENTAL CONC	ENTRATIONS	FOUND IN S	TANDARD FI	LTERS (Mg)
	M	easured by	INSTRUMEN	THE NEUTRO	IN ACTIVATIO	IN ANALYSIS
		Cr	Fe	Mn	Se	Zn
Filter No.	l	1.91	198.4	4.1	0.85	70.8
Filter No.	2	1.93	201.0	4.3	0.81	72.1
Filter No.	3	1.90	186.6	4.0	0.84	73.4

TABLE 3.8 "TREE" ELEMENTAL CONCENTRATIONS SPOTTED ON STANDARD FILTERS (MG)

.

	"True" Value	Range
Cr	1.85	1.76- 1.94
Fe	194	184.3-203.7
Mn	4.2	4. 15 -4. 41
Se	0.85	0.81-0.89
Zn	74	70.3-77.7

.

3.5.3 <u>RESULTS</u>

The concentration of elements in dust particulates collected in Glasgow are given in Table 3.9. The results for Br, Co, Cr, Fe, Mn, Na, Sb, Se, and Zn represent 165 samples from 1st August 1972 - 31st August 1973. The results for La represent 50 samples at various times during the above period for reasons already discussed. The results for Cs, Hf, Lu, Th and Yb represent 60-70 samples again at various times during the above period (not all necessarily from the same sample). This is due to the fact that the appearance of these elements in the samples was sporadic and the range and mean were calculated from those samples in which the element was detected.

The results for similar industrial cities in other countries and also the threshold limit values for 1973 for factories in Great Britain are included for comparison. The individual results for the 10 elements collected over 13 months are illustrated in Figure 3.6, and emphasise the wide scatter of results and the need for some sort of interpretation.

The results indicate the wide variation in element content of atmospheric particulates, but in spite of these fluctuations the mean values obtained are generally in agreement with results quoted for other parts of the world.

It is immediately obvious that even the maximum values found for these elements in ambient air are in no case anywhere near the threshold values. However, it is unwise to depend upon industrial exposure data and to use industrially allowable levels as community standards because -

TABLE 3.9	ATMOSPHERIC LEVELS OF	TRACE ELEMENTS (ng/M ³)
Industrial Limits (mg/M ³) (244)	0.01 0.5 0.5 0.5 0.5	5.0 2.0 (as hydroxide) 0.5 0.2 - 5 (as oxide)
<u>Glamorgan</u> Wales (117)	56 13.5 6090 1	24 1020 2.9 0.21 1.7 0.16 - 367
<u>Paris</u> France (243)	- 433 6.67 15.1 3500 3.42 -	82.5 50.8 0.7
San Francisco U.S.A. (121)	0.05-0.2 0.05-0.2 0.46-1.6 2.2-16.0 610-3000 0.69-3.1	5-31 2600-6300 0.68-3.8 0.48-2.1 0.31-1.9 0.06-0.36 27-500
Texas U.S.A. (242) (50 Stations)	20-470 50-1100 100-10,000 -	20-1700 - - 5-1770
Heidelberg Germany (129)	30.5 2.22 2.22 1041 1.1	23.6 224 6.3 0.50 - -
Niles Michigan U.S.A. (124)	1 43 9.5 1.3 1.3	62 170 6.3 1.2 2.5 0.27 140
<u>Chicago</u> U.S.A. (124)	2.4 67 2.6 113 13,800 5.9 -	255 455 3.1 3.8 1.3 1400
۲ ۱۳۹۲ Mean	6.07 0.08 0.09 0.03 0.03 0.03 0.03	23 331 331 331 319 319 319
Glasge Strig - Augu <u>Kange</u>	0.3-12.9 11-700 0.06-14.4 0.1-89.8 0.06-7.0 17-7100 0.03-0.7 0.07-2.38 0.01-0.8	0.6-290 66-2075 66-2075 0.01-2.85 0.01-2.85 0.01-2.85 0.01-1.0 0.3-4.3 7.7-3646
Auza	A H S L S L H H H H H H H H H H H H H H H	MM Na Na Z N D Na Z N D Na Z N D Na Na Na Na Na Na Na Na Na Na Na Na Na

- = Not Determined



GRA	PHICAL	REF	PRESENTATION	N OF	ATMOS	SPHEI	RIC	
	LEVELS	OF	ELEMENTS, S	SMOKE	AND	SO2		
AT	CLASE	OW	UNIVERSITY	Auc	UST "	12-	AUGUST '-	13,



FIGURE 3.6 (CONTD.)















-







FIGURE 3.6 (CONTD.)







8I



- i) Workers are usually in a controllable occupational atmosphere which is confined in its exposure to a limited area (2).
- ii) Exposed persons are usually in good health and are removed if they show idiosyncracies to a particular pollutant (2).
- iii) They are usually younger and healthier than the general population (2).
 - iv) In the normal working week of 40 hours they have 180 hours of non-exposure (2).
 - v) They are usually exposed to a single pollutant. (However, it must be pointed out that these workers are probably more at risk generally since they are also exposed to general urban environment outside working hours) (2).
- vi) Threshold limit values for industrial exposure reflect the capacity for detoxification of pollutants by the body and may be inapplicable to air pollution since continuous exposures do not give the same possibilities for excretion or detoxification of pollutants during times when more are being absorbed (195).

3.5.3.1 Correlation of Elements

As can be seen from Figure 3.6 in general the elements, smoke, and SO₂ trends follow each other. However, in order to attempt to place these results on a mathematical basis, straight line correlation coefficients were calculated by computer programme on the 158 samples. The results are given in Table 3.10 and all are significant at the 95% level or greater (30 out of 36 being at the 99.9% level). Total smoke and SO₂ levels were calculated for the

TABLE	<u>3.10</u>	CORRELA	TION CO	DEFFICIE	NTS OF	ELEMENT	S WITH	EACH O	THER	
		AT MOS	PHERIC	PARTICU	LATES	- GLASK	AND SMC	DKE AND	SO22 1979-1	975)
Cr	Cr									
Co	0.61	Co					•			
Br	0.23	0.44	Br							
Fe	0.58	0.93	0.34	Fe		1				
Mn	0.27	0.42	0.62	0.30	Mn					
Na	0.43	0.73	0.52	0.66	0.54	Na				
Sd		0.22	0.27	0.17	0.23	0.12	Sb			
Sc	0.56	0.91	0.27	0.93	0.30	0.61	0.19	Sc		
Se				0.16			0.41		Se	
Zn	0.43	0.55	0.61	0.44	0.69	0.56	0.36	0.44		Zn
Smoke		0.26	0.64	0.14	0.54	0.35	0.26			0.3 9
50 ₂		0.12	0.61		0.59	0.31	0.33		0.28	0.25

.

sampling periods from daily levels at the Glasgow Art Gallery's site of the Smoke Officer. This was the most suitable site being adjacent to the University. The missing values (Table 3.10) were deliberately omitted as their coefficients of significance were less than 95%.

Some very good correlations are evident, namely, Fe/Sc, Fe/Co, Co/Sc and to a lesser extent Co/Cr, Na/Sc, Mn/Zn, Co/Na and Fe/Na. Many more are in the moderate correlation coefficient bracket and taking into consideration the dynamic nature of the atmosphere they are probably quite meaningful.

Three of the results are in extremely good agreement with those of John <u>et al</u> (121) namely, Fe/Sc, Fe/Co and Co/Sc. John <u>et al</u> also had correlation coefficients of 0.9 and greater for the combinations of Co, Cr, Mn, Sb and Sc. Their samples were, however, taken in a rural setting (San Fransciso Bay area) and were thus mainly soil derived. (See Section 3.5.3.3).

Another interesting point is the relationship between Sb and Se, the two most volatile elements in the group. It will be noted that apart from Fe, Sb is the only element which bears any relationship to Se and that in the case of Sb the relationship with Se is the best of all elements measured.

In these results Br and Mn display the best relationships with total smoke and SO_2 but why this should be so is difficult to understand.

The Se/SO_2 coefficient is disappointing since Se and S are recognised as being isomorphic. Most of the natural sources of Se are from S bearing minerals. The Se/S ratio was calculated from monthly averages (S being calculated from SO_2) and the results

are given in Table 3.11, together with those for Buffalo and Cambridge, U.S.A., and also values for coal and crude oil used in U.S.A. (140). As can be seen the average value for Glasgow is an order of magnitude lower than the other two although the range lies within that of Cambridge. The difference between Buffalo and Cambridge is explained in terms of differing fuel consumption habits. This may well be the explanation for the difference in the Glasgow situation as well. There will almost certainly also be a difference in the Se/S ratio of Scottish fossil fuels compared to those employed in the U.S.A. study.

3.5.3.2 Seasonal Variations

Monthly 'averages were calculated from daily values and are plotted in Figure 3.7.

The following points can be made about the graphs -

- i) Apart from Cr there is good agreement between the results for August 1972 and August 1973.
- ii) The good correlations noted earlier between Br, smoke and SO₂ are evident. Br has two definite peaks: September/ October/November and January/February/March, i.e. the Autumn and Winter months. The December result is curious in that there is a definite reduction in level despite being the middle of Winter.
- iii) The trends for Zn and Mn are obscured by peaks in October. However, by removing two peak values well in excess of the remaining values occurring during 4-5/10/72 and 17-18/10/72 when there were 11 hours of calm winds with 4 hours of fog

TABLE	<u>3.11</u> <u>SELENIU</u>	1 - SULPHUR RATIO TN	ATNOSPHERIC PARTICULATES
		AT GLASSON UNIN	1245/TY 1979-73,
		Range	Average
	Glasgow	0.1x10 ⁻⁴ - 1.1x10 ⁻⁴	0.04x10 ⁻³
	Buffalo	$0.5 \times 10^{-3} - 1.6 \times 10^{-3}$	0.9x10 ⁻³
	Cambridge	$0.1 \times 10^{-4} - 4.4 \times 10^{-4}$	0.1x10 ⁻³
	Crude Oil (U.S.A.)	$0.1 \times 10^{-4} - 0.6 \times 10^{-4}$	0.5x10 ⁻⁴
	Coal (U.S.A.)	$0.5 \times 10^{-4} - 5.0 \times 10^{-4}$	2.9x10 ⁻⁴

0-



FIGURE 3.7 (CONTD.)







9I

and 13 hours of calm winds with 6 hours of fog respectively, the trend becomes more evident. Peak values occur during December and January and lower values during the Summer.

The justification for removing these two peaks will be discussed later. (See Section 3.5.3.4.7).

- iv) Similarly be removing 1 peak value for Na and Co during 7-9/5/73 with 20 hours of calm winds with 10 hours of fog again gives a trend of peak levels covering the October/ January period and lower values during the Summer months.
- v) The patterns of the two volatile elements Se and Sb after
 October are in good agreement as far as peaking and
 troughing are concerned, although the graphs intersect twice.
 This would also help to explain why the correlation coefficient is not as high as might be expected. The peaks
 are also moved slightly to the right and occur more in
 Spring and Summer. Whether this is a reflection of
 temperature affecting the residence times of these volatile
 elements is an open question at this stage.
- vi) Fe does not vary much throughout the seasons. The large value for March is reduced drastically by removing the same sample as for Na and Co. (7-9/3/73).
- vii) For Cr and Sc there are no apparent trends with random fluctuations throughout the year.

To summarise, after removing peaks for the reasons given above there is a general trend of higher levels during Winter months reflecting perhaps more fuel consumption.

The exceptions are Se and Sb the volatile elements which tend to peak during Spring and Summer and their results are in better agreement than might be suggested by their correlation coefficient. Fe is relatively stable throughout the year as might be expected from the continuous processes involved in iron and steel works. Finally Cr and Sc seem to be independent of season and fluctuate randomly.

Although these results by and large are what would be expected it must be said that they are based on one year's sampling and although the August 1972 and August 1973 results are in good agreement a further year's results would have strengthened or refuted the conclusions made and could also improve the Cr and Sc seasonal dependency. (In fact the Cr results for August 1972 $(21.6ng/M^3)$ and August 1973 $(9.4ng/M^3)$ are the only ones which are not in agreement.

3.5.3.3 Enrichment Factors

Several workers have attempted to explain the source of atmospheric elements by comparing the ratio of the elements to some standard in the aerosol with that of a reference material (121, 127, 129, 245-251). The two main reference materials which have been used are soil or crustal rock and the sea. These of course are obvious choices - once thought of - the sea being more relevant to coastal aerosols and soil to inland sites. The standard is usually an element which is relatively abundant in both the reference material and in the aerosol, e.g. Na would be used when the sea is the reference material while Al or Fe would
be used when soil is the reference material. Any element which is present in both the reference material and in the aerosol can, however, be used and in one study (245) Sc which is a relatively minor constituent of aerosols was used.

The enrichment factor, E(x), is defined as -

where Conc. X aerosol and Conc. X reference are the concentrations of element X in the aerosol and reference material respectively. Similarly Conc. R aerosol and Conc. R. reference are the concentrations of the normalising element in the aerosol and reference material respectively.

If E = 1 for element X, the ratio of it to the normalising element is the same in the aerosol as in the reference material. If E = 2, the ratio of it to the normalising element is twice as high in the aerosol as in the reference material, i.e. the element X is enriched by a factor of 2 in the aerosol. If all the elements in the aerosol have E = 1 the composition of the aerosol and the reference material are exactly the same.

In the present study enrichment factors were calculated with reference to soil and to crustal rock where soil figures were not available. The levels used were those of Bowen (252). The factors were calculated using Fe as the normalising element and based on the yearly average value of each element as has been done elsewhere (249). The results are given in Table 3.12, together with those of Rahn for Liege (Belgium) based on Al as the normalising element (251).

TABLE 3.12	ENRICHMENT	FACTORS	OF	ELEHENTS	IN	ATMOS PHERIC	PARTICULATES

	Glasgow (12.13)	Liege
Ag	3490	910
Br+	1706	2200
co+	13	5.8
Cr+	11	
Cs [†]	19	20
Fe ⁺	l	2.9
Hf *	13	
La	3	3.8
Lu [#]	52	
Mnt	4	5.1
Na ⁺	7	1.2
Sb*	118	2500
Sc ⁺	2	1.2
Se [†]	2133	4300
Th ⁺	8	
Yb*	56	
Zn ⁺	822	2100

#-BASED ON ROCK

+ = BASED ON SOIL

r

As can be seen -

- i) The Glasgow values for Cs, Fe, La, Mn and Sc are in extremely good agreement with those of Liege.
- ii) Se, Br and Na are of the same order.
- iii) Ag and Co are respectively 4 and 2 times higher for Glasgow while Zn is 3 times smaller.
- iv) Only for Sb is there any great difference between the two sets of results, Glasgow being 21 times lower.

In the study with Sc as the reference element it was found that Zn, Se, Sb and Pb were enriched by several orders of magnitude (Pb as will be seen later is also highly enriched in Glasgow's atmosphere) and that La, Fe, Hf and Th ratios were comparable with an earth crust origin. This again is the same pattern as was found in Glasgow's atmosphere.

It is clear that Ag, Br, Sb, Se and Zn are greatly enriched However, the compared to soil and Lu and Yb moderately so. fact that the other elements have about soil proportions does not necessarily mean that they indeed originate from soil or rock (although of course wind-raised soil does exist and in rural areas will contribute to atmospheric levels it is difficult to imagine enough wind-raised soil entering the city to be solely responsible for the high levels of elements found). In fact in the Liege study it was estimated that between 70-98% of the masses of the elements were from local pollution. Also Dams et al (247) showed that for industrial N.W. Indiana Fe, Mn, Zn. Sb, Cr, W, Co, Sc, La, Ce, Th, Ca, and Mg were linked to local steel production and supporting activities. The reason for this

anomaly (soil proportions but not soil derived) is said by Rahn (251) to be the presence of fly ash in the atmosphere from fuel burning and other activities. Rahn has suggested (with evidence) that there are two distinct inorganic fractions in coal one of which is the rock fraction consisting of Na, Mg, Al K, Ca, Ti, Cr, Mn, Fe, Zn, Th, La and Ce and this is confirmed by Masson (253). This mineral fraction was deposited at the same time as the original organic material. The second fraction consisting of V, Co, I, Se, As and Sb is not present in the mineral fraction and was probably precipitated from ground water by absorption or chemical reaction as the coal was being formed.

It will be noted that most of the elements in the first fraction are those which are closest to soil or rock proportions; and this fly ash is an alternative source of large quanitites of material which has rock proportions; material which is actually rock particles released nearly unchanged by pollution sources. Zn which has a larger enrichment factor must have additional sources.

As part of the same study Rahn analysed various fuels and emissions from industry and found -

- i) Coal is the largest source of Al, As, Ba, Cd, Cr, Mo, S, Si, Sn, and Ti and contributes important amounts of Ca, Co, Mg, Se and V.
- ii) Fuel oil is most important for Co, Ni, Se and V.
- iii) The iron and steel industry contributes most importantly toCu, Fe, Mg, Mn and Zn.

iv) Eurning gasoline is most significant for Br, Cl and Pb.By studying this list the additional sources of enrichedelements can be seen.

A further explanation of the enrichment of the volatile elements is by volatilisation of these elements during combustion and preferential condensation on to the smallest particles which will have a longer residence time in the atmosphere compared to the larger fly ash particles.

When considering the emissions inventory above which by no means covers the whole industrial spectrum it is probably surprising that the correlation coefficients were as good as they were rather that a disappointment.

3.5.3.4 Climatic Variations

Similar to the question of which elements to study the question of which weather parameters to study arises, and again the best approach is to utilise all the data available and reject, if not appropriate, after use.

ž

3.5.3.4.1 Wind

Wind is the most important climatic aspect of air pollution, acting as a transporter and hence as a diluter. The direction in which the wind is blowing is of course important in determining in which direction the pollution will be carried as is also the wind speed since low winds will allow the pollution to remain concentrated for longer periods.

It has been shown that wind observations taken at local airports are suitable for air pollution studies in the neighbouring city (194), Subsequently Glasgow Airport observations were used in this work. Observations are taken hourly and these values

were averaged over the sampling period in the eight compass directions to give a total time for which the wind was blowing in any one direction, together with the average wind speed during that time. The following is the type of result arrived at -

Sampling Period	Direction from which wind is blowing	$\frac{\text{Average}}{\text{speed}}$	Duration (Hours)
15 / 9 a.m. 17/11/72 9 a.m.	N	2.0	1
·, ·, ·, · · · · · · · · · · · · · · ·	NE	3.0	1
	E	-	-
	SE	2.5	2
	S	2.0	l
· 、	SW	5.3	10
	W	5.9	15
	NW	7.0	8
	Calm Winds		10

The elements were plotted against (i) Wind Speed and (ii) Wind Duration for each direction individually, the assumption being that if an element has a strong source then this should show in the wind speed and duration for that particular direction relative to the sampling position. Straight line correlation coefficients were calculated by computer, but the association in all cases was found to be very weak, r-values of second and even third decimal place being frequent.

Correlation coefficients greater than 0.1 and confidence limits greater than 95% are given in Table 3.13. In each case only about half of the results were even in the low correlation bracket (0.20 - 0.40) with the remainder in the bracket usually reserved for no correlation (0.00 - 0.20). However, the

(AT HOSPH	ERC PARTICULATES	- GLASCOW UN	TION AND DISTANCE
Element/Direction	Wind Speed	Duration	<u>Wind Distance</u>
Br/SW	- 0.19		- 0.14
Br/W	- 0.26	- 0.20	- 0.28
Br/NW	- 0.21	- 0.18	- 0.20
Co/W	- 0.16	- 0.11	- 0.15
Cr/SW		- 0.19	- 0.18
Cr/W			- 0.14
Mn/NE	0.14	0.13	
Mn/E	0.23	0.20	0.15
Mn/SE	0.19	0.31	0.20
Mn/SW	- 0.31	- 0.25	- 0.24
Mn/W	- 0.37	- 0.28	- 0.29
Mn/NW	- 0.21	0.16	- 0.17
Na/SE		0.17	
Na/W	- 0.16	-0.20	- 0.17
Se/SE	- 0.14		- 0.13
Sb/S	- 0.16	- 0.14	- 0.17
Sd/SW	- 0.18	- 0.35	- 0.28
Sd/W	- 0.13		•
Sb/NW		- 0.14	
Zn/SW	- 0.16	- 0.27	- 0.24
Zn/W	- 0.23	- 0.18	- 0.20

 TABLE 3.13
 CORRELATION COEFFICIENTS OF ELEMENTS WITH WIND SPEED,

 DURATION AND DISTANCE
 DURATION AND DISTANCE

reported values were at least 10 times higher than the omitted values and are at least indicative of trends which should be further investigated. The sign is probably meaningless being negative in most cases and caused by the large number of concentrations of the various elements when no wind was blowing in that particular direction.

As said previously the concentration of a pollutant is related to both the length of time and also the speed of the wind in a particular direction. The next step therefore was to combine these two variables into one and this was achieved by using the "Wind Distance", i.e. speed x time and is a measure of the amount of air passing over the sampler in any given direction. These values were again computer plotted against the elements and correlation coefficients calculated. The results are given in Table 3.13 again for values greater than 0.1 only. It will be clear that the results are comparable with those for wind speed and duration and therefore add nothing new in terms of understanding.

Since the elemental value was an average for the sampling period it was thought that an average wind speed and duration for the same period would show a closer relationship. To be meaningful this average must allow weight to be added to the varying influences cf duration and speed in the eight directions. This was achieved through the use of vectors using the formula -

$$R = \sqrt{X^2 + Y^2}$$

where R =

R = resultant force

X = the component of the force acting along the X - axis Y = the component of the force acting along the Y - axis and in the direction given by Tan $\boldsymbol{\theta} = \mathbf{Y}/\mathbf{x}$.

IOI

X is given by
$$A_1 \cos \alpha_1 + A_2 \cos \alpha_2 + \cdots + A_n \cos \alpha_n$$

Y is given by $A_1 \sin \alpha_1 + A_2 \sin \alpha_2 + \cdots + A_n \sin \alpha_n$

where A is the magnitude of the force and $\boldsymbol{\propto}$ is the angle which the line of action of that force makes with the X - axis.

Obviously with different numerical values for speed and duration the angles of action of the resultant forces are going to be different. To obtain an angle which included components of speed and duration therefore "Vector Distances" were also calculated. The zero degree point was taken as East and all angles calculated relative to this direction. For example, for the sampling period given earlier this process gave -

Vector Speed (VS) = 10.8 m.p.h.; Vector Angle for Speed $(VAS) = 352^{\circ}$ (i.e. 8° below east) Vector Time (VT) = 25.7 hours: Vector Angle for Time $(VAT) = 4^{\circ}$ Vector Distance (VD) = 160 miles; Vector Angle for Distance

 $(VAD) = 0^{\circ}$

These again were plotted against the elements and correlation coefficients calculated. The results are given in Table 3.14 and again have confidence level of 95% or more. Again all three sets are comparable and in some cases slightly more encouraging than previously.

With vector angles it is the shape of the graph which is important rather than the correlation coefficient, pockets of points being desired rather than straight lines. Figure 3.8 illustrates the plots obtained for Se and are typical of all the

and

	Vector Speed	Vector Time	<u>Vector Distance</u>
Br	- 0.35	- 0.31	- 0.31
Co	- 0.14	- 0.16	- 0.16
Cr		- 0.22	- 0.22
Fe			
Mn	- 0.28	- 0.23	- 0.27
Na			
Sb	- 0.19	- 0.38	- 0.31
Sc			
Se			
Zn	- 0.18	- 0.31	- 0.30

TABLE 3.14	CORRELATION	COEFFICIENTS	OF	ELEMENTS	* WITH	VECTORS	OF
					WIND	PARAMETT	TRS

TABLE 3.15	CORRELATION	COEFFICIENTS	OF	ELEMENTS	K- WITH	CALM	HOURS
Br		0.42					
Co		0.31					
Cr							
Fe							
Mn	(0.27					
Na							
Sb	(0.32					
Se							
Sc							
Zn	(0.29					

* ATTICSPHERIC PARTICULATES - GLASCON '19-13.

FIGURE 3.8 COMPUTER PLOT OF SELENIUM V WIND PARAMETERS (ATMOSPHERIC PARTICULATES - GLASCOLS UNIVERSITY '72-73)



FIGURE 3.8 (CONTD.)





SENENIUM CONC. (ng/M3)

I06

:

3.<u>3.</u> (* 1.8.)

elements. As can be seen in the speed and time plots it is tempting to divide the points into three groups but it is not until the distance plot is studied that this can be done with justification. The three groups are:-

ii) 144° - 25°° which is the middle group and corresponds to SE- NE winds

This does not mean that the elements are derived from these directions, although they all correspond to industrial complexes, since by the very way in which the vector angles were calculated, a direction may be obtained which did not appear in the sampling period. It does at last show that there is a relationship between elemental concentration and wind, speed and duration both being important.

As mentioned previously periods of low wind speeds lead to a build up of pollutants. To investigate this, elemental concentrations were plotted against the total number of calm hours per sample. The correlation coefficients are given in Table 3.15. The sign of the elements given are all positive, i.e. as the number of calm hours increase the concentration of elements increase. 3.5.3.4.2 Rain

Rain is said to be the most important scavenger of the atmosphere. Uniform rain falling at 1 mm/hr for 15 minutes will scavenge 28% of 10 µm particles. The efficiency decreases with particle size and becomes negligible at 2 µm and smaller (194). There are two modes of action -

i) Washout where particles impact on water droplets.

ii) Rainout or condensation around the particles to form a droplet. There can be up to 25 particles/drop of rain for normal dust concentrations(194).

Rain is measured on a 24-hour basis at the weather centre giving total rain in mililitres and total time for which rain was falling.

These were totalled for the sampling periods to give -

٠.

- i) Total rain which fell in mililitres.
- ii) Time for which rain was falling in hours.
- iii) Rate at which rain was falling (by dividing i) by ii) in mililitres/hour.

These values were plotted against the elements and straight line correlation coefficients calculated. The results are given in Table 3.16. As can be seen the results were extremely disappointing, it being expected that during periods of rain the levels of the elements would be reduced. The omitted values were an order of magnitude less than these given.

Three samples of rain water were collected on 10-11/12/74, 2-3/6/75 and 21-22/8/75 in a previously acid washed plastic container placed on the roof of Glasgow University chemistry building. The samples were acidified with "Aristar" nitric

TABLE 3.16	CORRELATION COEFFICIENTS C	F ELEMENTS WITH	RAIN
	(ATHOSPHERIC PARTICULATES	- 9-ASGOW UNIVE	R31TY '79-13)
`	Duration of Rain	Amount of Rain	Rate of Rain
Br		- 0.13	- 0.14
Sb	- 0.17		
Zn	- 0.15	- 0.14	
	-		

TABLE 3.17	CONCEN	TRAT	IONS	OF	ELEMEN	TS IN	RAINA	<u>ŤER</u>	(ppm)	
COLLECTED AT GLASSOW UNIVERSITY.											
Sample	e Al	Cd	Co	Cr	Cu	Fe	Mn	Na-	Ni	Pb	Zn
1	0.21	ND	ND	ND	0.08	0.05	0.01	7.5	ND	0.04	0.14
2	0.15	ND	ND	ND	0.03	0.03	0.02	0.4	ND	0.04	0.08
3	0.28	ND	ND	ND	0.05	0.08	0.01	1.5	ND	0.03	0.04
•											

ND = Not Detected

acid and filtered through a 0.45 µm membrane filter and then analysed by atomic absorption spectroscopy using the method of standard additions. The results are given in Table 3.17. It can be seen that some of the elements at least are being washed/ rained out although the amount is not in linear relationship to the amount, time and rate of precipitation of rain.

3.5.3.4.3 Fog

Fog is defined as the suspension of water droplets in the atmosphere, with or without particulate matter. Fog with a smoke combination was the original "smog" before American workers extended the meaning to cover photochemical pollution.

In a city atmosphere fog will consist of particulate matter and hence it would be expected that periods of fog would exhibit elevated levels of elements due to their being held in the atmosphere under stagnating conditions for a longer period.

At the weather centre visibility distances are measured When visibility is less than 1 mile the distance . every hour. These latter readings were taken as foggy is given in yards. conditions and were averaged out over the sampling periods to give the average visibility during foggy conditions and the duration. These values were plotted against the elements and correlation coefficients calculated. The results are given in The samples which had no foggy conditions were Table 3.18. given a visibility of 4,000 yards and as the programme limits were set at 3,900 yards these samples were omitted. For this reason the signs are meaningful and reflect what would be expected, i.e. as visibility decreases the concentration

TABLE 3.18	CORRELATION COEFFICIENTS OF ELEMENTS WITH FOG (ATTICSPHERIC PARTICULATES - GLASSCON UNIVERSITY 72-73.)									
	Visibility	Duration								
Br	- 0.36	0.42								
Mn	- 0.29	0.39								
Sb	- 0.28	C.18								

- 0.26 0.29

 $\mathbf{Z}\mathbf{n}$

increases and as duration increases the concentration increases. The actual values are also more promising than previously with values as high as 0.42 being recorded. The fact that both factors are varying at the same time will also tend to obscure any relationship and the correlation may even be better than is suggested here.

3.5.3.4.4 Temperature

Environmental temperature is important due to its effect on the uptake of pollutants by plants, and in modifying the toxic response of inhaled substances in animals with high and low temperatures increasing the undesirable effect. It is also an important factor in the formation of photochemical air pollution together with sunshine.

Kneip et al (254) studying 8 elements in New York City found that V was inversely correlated to temperature and Pb, Cu and Cd were directly correlated with temperature. The effect of temperature was investigated on these results. Readings at the weather centre are taken every hour. However, the hourly variation was not significant and values at 1200, 1800, 0000 and 0600 hours and the daily maximum and minimum values were found to be enough to give an accurate average temperature for the sampling periods. The values were plotted against the elements and correlation coefficients calculated. The results are given in Table 3.19. Again the sign is meaningful and the relationship is inverse. Of the elements in common, Kneip et al found no correlation of temperature with Zn, Mn, or Cr. This was also found to be the case here for Zn and Cr but there was a weak correlation with Mn.

II2

TABLE	3.19 CORRELA	TION COEFFIC	IENTS O	F ELEMEN	TS WITH	VARIOUS	
	_			WEAT	HER PARA	AMETERS	60.00
	ATHOS	SPHERIC PAR	RTICULI	ATES = SI	1 11092A	NVERSIT	<u>y 73-75)</u>
	Temperature	C.C. C.H.	SH.	VP.	RH.	AP.	
Br	- 0.24	0.16	- 0.28	- 0.21	0.34	0.27	
Co						0.17	
Cr						•	
Fe						0.15	
Mn	- 0.25	0.17	- 0.25	- 0.21	0.23	0.16	
Na	- 0.27		- 0.14	- 0.28	0.13		
Sb		0.15 -0.21					
Se	- 0.20			- 0.27			
Sc						0.19	
Zn			- 0.18			0.17	

C.C. = Amount of Sky covered by clouds C.H. = Height of clouds

SH. = Hours of sunshine

VP. = Vapour pressure

RH. = Relative humidity

AP. = Atmospheric pressure

3.5.3.4.5 Cloud Cover

It would be expected that low clouds would elevate concentrations due to enclosing pollution in the lower atmosphere.

At the weather centre cloud cover is recorded in terms of eighths of the sky covered and also the height of that cloud, on an hourly basis. These were averaged over the sampling period to give the average sky covered and average height of cloud. The results of plotting these values against the elements are given in Table 3.19. Again for these elements given, the association is very weak but the signs are as would be expected. As the amount of sky covered increases the concentration increases and as the height increases the concentration decreases.

3.5.3.4.6 Miscellaneous

Other parameters which are measured at the weather centre are sunshine hours, relative humidity, atmospheric pressure and vapour pressure. Sunshine hours are recorded as a total per day and were simply added to give a total number for the sampling period. For the others, which are recorded hourly, it was found necessary only to treat them similarly to temperature readings to obtain an average value. The results of correlating these with the elements are given in Table 3.19.

Although some correlation exists it must be remembered that climatic conditions are inter-related, e.g. relative humidity is related to fog; low atmospheric pressure leads to stagnating conditions, etc., and hence some relationship might be expected.

II4

However, Lininger \underline{et} al (50) did find that some Br concentrations in Cambridge, Mass. "seem to be related to relative humidity".

On the whole this computer approach was disappointing as manifested in the correlation coefficients. One reason for this, as already stated, is trying to cope with more than one variable at a time. In all, including the eight wind directions 30 variables have been studied. Two different approaches were adopted -

- i) Visual inspection where samples were selected to correspond with weather conditions of rain, fog, etc. This approach is of course empirical but does allow an element of choice to account for the varying factors, e.g. rain and fog during the same sampling period might cancel out each other's effect on the average elemental level measured.
 On this empirical basis an element was said to increase or decrease if it was larger or smaller respectively than the samples on either side by more than the experimental error.
- ii) Use of a computer programme capable of handling more than one variable at a time, i.e. obtain a multiple regression coefficient for the elements plotted simultaneously against the 30 variables.

3.5.3.4.7 Visual Inspection

3.5.3.4.7.1 Wind:- From the data as set out in the example previously given samples were sub-divided into wind directions, with no set criteria but simply by judging the

II5

most predominant direction. On this basis 48% of the samples had predominantly the prevailing W/SW winds and 13% had E/NE winds. The remainder were mixed.

Trying to contribute any element to E/NE winds can be seen from Table 3.20 to be impossible due to the other weather conditions prevailing at that time and emphasises what was said previously. No further attempts were made with this approach.

Samples were also collected which occurred during periods of calm. In the majority of cases it was found that these periods of calm were accompanied by haze, i.e. visibility greater than 1,000 yards but less than 1 mile. The cases where there was no accompanying haze can be attributed to two causes.

- i) The hours of calm were scattered throughout the sampling period although giving a high total.
- ii) To the wind measuring instrument calm means a wind speed less than 3 m.p.h. This speed is probably enough to prevent haze forming but low enough to be recorded as calm.

The haze is probably due to the collection of pollution during the stagnating calm periods.

In all cases except three where there was an appreciable amount of calm or haze (greater than 5 hours) at least one element was elevated. In the three exceptions there was an appreciable amount of rain (at least 13.6 mililitres) which would tend to nullify the elevating effect of calm conditions. The results are given in Table 3.21. II**7**

									(ATH	ICSP	HLPI	C PARI	FIRULATES
TABLE 3.2	20	ELEI	TENTS	S AFI	FECTI	ED BY	<u>r e/n</u>	VE WIN	VDS	94	asso.	J UN	いにわらいて	v 12-73)
<u>Date</u>		E	lemer	nts I	Eleva	ated		E	Leme	nts	Decr	ease	d <u>Pa</u>	<u>Other</u> rameters
4/ 5/10/72	Se,	Cr,	Zn,	Co,	Fe,	Mn,	Na							Haze
9/ 11/10/72	Sc,	Sb,	Fe					Se, Z	Zn,	Co,	Na,	Br		Rain
6/ 8/12/72	Se, Mn,	Cr, Na,	Sc, Br	Sb,	Zn,	Co,	Fe,							Haze
15/ 18/12/72	SЪ,	Zn,	Co,	Fe,	Mn,	Br		Se,	Na					Haze
27/ 29/12/72	Se, Fe,	Cr, Mn,	Sc, Na,	Sb, Br	Zn,	Co,								Haze
12/ 15/1/73	Cr,	Sc,	Co)	Na						
17/ 19/1/73	Se, Mn,	Cr, Na,	Zn, Br	Sb,	Co,	Fe,								Haze
12/ 14 /3/ 73	Br						•	Zn,	Fe,	Mn				
20/ 23/4/73	Se							Sc,	Sb,	Zn,	, Co,	Fe,	Na	
23/ 25/4/73	Na							Cr						
2/ 4/5/73	Fe							Se						
16/ 17/5/73	Se,	Fe												
17/ 18/5/73								Se,	Cr,	Sb	, Z n,	Br		
18/ 21/5/73	Sb,	Zn,	Co,	Mn,	Na,	Br								
21/ 22/5/73	Se							Zn,	Co,	Fe	, Mn,	Na		Rain
								Cr,	Sc,	Br				Cloud Clover
22/ 23/5/73 23/	Se							Sb,	Br					
25/ 24/5/73	Se, Fe,	Cr, Mn,	Sc, Na,	Sb, Br	Zn,	Co,								Haze

,

TABLE 3.20(CONPD.)DateElements ElevatedElements Decreased28/
29/5/73SbCr, Zn, Fe, Br16/
18/7/73Se, Cr, Sb, Zn, Fe, Br20/

 20/
 22/8/73
 Cr
 Sc, Sb, Zn, Mn, Na

 22/
 24/8/73
 Se, Cr, Sc, Sb, Zn, Co, Fe, Mn, Na, Br
 Haze

 24/
 27/8/73
 Se, Cr, Sc, Sb, Zn, Co, Fe, Mn, Na, Br
 Haze

<u>Other</u>

Parameters

Haze

							(ATH	csPh	FRIC.	PARTICULARIES
TABLE 3.2	1 ELEMENTS AF	FECTED BY	NUMB	ER OI	CAI	LM HO	DURS	SLA	SULIN	: VNI	LEPSITY 72-11
Date	Hours of Calm	<u>Hours of</u> Slight De	Fog nse		E	leme	nts_	Eleva	ated		
18/ 20/9/72	14	1		Se, Na,	Cr, Br	Sc,	Sb,	Zn,	Co,	Fe,	M.n.,
26/ 28/9/72	29	11	4	Se,	Cr,	Zn,	Co,	Fe,	Mn,	Na,	Br
2/ 4/10/72	23	32	5	Se,	Cr,	Sc,	Sb,	Zn,	Co,	Fe	
4/ 5/10/72	16	17		Se,	Cr,	Zn,	Co,	Fe,	Mn,	Na	
5/ 6/10/72	. 8	2	2	Se,	Sb,	Zn,	Co,	Fe,	Mn,	Na	
11/ 13/10/72	21	2		Cr,	Zn,	Co,	Na,	Br			
16/ 17/10/72	15			Se,	Co,	Mn,	Br,	Na			
17/ 18/10/72	9	13	6	Cr, Br	Sc,	Sb,	Zn,	Co,	Fe,	Mn,	Na,
1 9/ 21/10/72	10	6		Br							
30/10/ 1/11/72	14	4		Cr, Br	Sc,	Sb,	Zn,	Co,	Fe,	Mn,	Na,
13/ 15/11/72	9	4		Se,	Cr,	Sb,	Zn,	Co,	Mn,	Br,	Fe
15/ 17/11/72	10	9	3	Se,	Sc,	Sb,	Zn,	Co,	Na,	Br	
17/ 20/11/72	12	12		Se,	Sc,	Zn,	Fe,	Mn,	Na,	Br	
24/ 27/11/72	15	10		Se							
1/ 4/12/72	22	16	9	Se,	Cr,	Br					
6/ 8/12/72	17	11		Se, Na,	Cr, Br	Sc,	Sb,	Zn,	Co,	Fe,	Mn,
8/ 11/12/72	10	5		Cr,	Sb,	Zn,	Co,	Mn			
15/ 18/12/72	6	22		Sb,	Zn,	Co,	Fe,	Mn,	Br		
20/12/72	16	15		Cr, Br	Sc,	Sb,	Zn,	Co,	Fe,	Mn,	Na,

TABLE 3.21 (CONTD.)

Date	Hours of Calm	<u>Hours</u> c	of Fog Dense		E	leme	nts 1	Elev	ated		
20/ 22/12/72	4	4		Sb,	Zn,	Fe,	Na,	Br			
27/ 29/12/72	2	9		Se, Na,	Cr, Br	Sc,	Sb,	Zn,	Co,	Fe,	Mn,
30/12/72 2/1/73	14	5		None	9						
3/ 5/1/73	6	15		Cr, Br	Sc,	Sb,	Zn,	Co,	Fe,	Mn,	Na,
5/ 8/1/73	32	21		Sc,	Sb,	Z n,	Co,	Fe,	Mn,	Na,	Br
8/ 10/1/73	28	8		Se, Br	Sc,	Sb,	Zn,	Co,	Fe,	Mn,	Na,
15/ 17/1/73	9	3		None	9						
17/ 19/1/73	15	6		Se, Br	Cr,	Zn,	Sb,	Co,	Fe,	Mn,	Na,
22/ 24/1/73	8	4	10	Sb,	Zn,	Co,	Mn,	Br			
26/ 29/1/73	5			Se,	Br						
7/ 9/2/73	6			Cr,	Se,	Sb,	Zn,	Co,	Fe,	Mn	
14/ 16/2/73	31	10	1	Se, Br,	Cr, Co	Sc,	Sb,	Zn,	Fe,	M'n,	Na,
16/ 19/2/73	26	23	11	Cr,	Sc,	Zn,	Fe,	Mn,	Br		
23/ 26/2/73	4	3		Cr,	Sc,	Zn,	Mn,	Br			
26/ 28/2/73	18	14		Se, Na,	Cr, Br	Sc,	Zn,	Sb,	Co,	Fe,	Mn,
7/ 9/3/73	10	2		Cr,	Sc,	Sb,	Zn,	Co,	Fe,	Mu	
9/ 12/3/73	37	20	10	Cr,	Sb,	Zn,	Co,	Fe,	Mn,	Br	
14/ 16/3/73	11	9	4	Se,	Zn,	Fe,	Mn,	Br			

TABLE 3.21 (CONTD.)

Date	Hours of Calm	<u>Hours of Fog</u> Slight Dense		E	lemen	nts_	Elew	ated		
19/ 21/3/73	6		Se,	Cr,	Sb,	Zn,	Mn,	Br		
27/ 30/4/73	5	1 1	Cr							
30/4 2/5/73	6	4	Se,	Sb,	Zn,	Co,	Br			
23 / 24/5/73	6	3	Se, Na,	Cr, Br	Sc,	Sb,	Zn,	Co,	Fe,	Mn,
25 / 28/5/73	12	12	Se, Br	Sb,	Sc,	Cr,	Zn,	Co,	Fe,	Na,
29/ 30/5/73	5	10	Cr,	Sc,	Zn,	Fe,	Br			
20/ 22/6/73	11	6	Se, Br	Cr,	Sc,	Zn,	Co,	Fe,	Mn,	Na,
22/ 25/6/73	9		Se,	Cr,	Zn,	Fe,	Br			
2/ 4/7/73	13		Sc,	Sb,	Zn,	Co,	Fe,	Mn,	Na,	Br
4/ 6/7/73	8		Se,	Cr,	Sc,	Zn,	Co,	Fe,	Mn,	Br
6/ 9/7/73	6		Cr,	Zn,	Fe,	Mn,	Na	•		
13/ 16/7/73	14	4	Se,	Cr,	Co,	Fe,	Na			
16/ 18/7/73	5	3	Se,	Cr,	Sb,	Zn,	Fe,	Br		
20/ 23/7/73	8		None	Э						
23/ 27/7/73	19		Se,	Sc,	Sb,	Zn,	Co,	Fe,	Mn	
31/7 3/8/73 9/	11		Sb,	Zn,	Br					
<i>]</i> 14/8/73	28	13	Se, Na,	Cr, Br	Sc,	SЪ,	Zn,	Co,	Fe,	Mn,
22/ 24/8/73	12	20	Se, Na,	Cr, Br	Sc,	Sb,	Zn,	Co,	Fe,	Mn,
24/ 2 7/ 8/73	17	42	Se, Na,	Cr, Br	Sc,	Sb,	Zn,	Co,	Fe,	Mn,

.

The frequency of elevation of the elements is given in Table 3.22. The results can be seen to be more encouraging than previously especially in the case of Se which only appeared once in the whole of the computer programme results.

3.5.3.4.7.2 Rain:- Samples were selected during which there was 5 mililitres or more of rain. The results are given in Table 3.23. There are 4 samples which increased during rain. In three of these (6-8/12/72, 14-16/2/73 and 30/4-2/5/73there was either foggy or calm conditions during the sampling period.

In general the number of elements which decrease tends to be shorter in samples in which haze or fog occurred. The frequency of reduction is given in Table 3.22.

Again the results are much more encouraging with the lowest percentage of 36 for Sc.

3.5.3.4.7.3 Fog:- Only 14 samples during the year had fog conditions. The results are given in Table 3.24. In only 1 sample was there no elevation of any element. The number of elements elevated was reduced in samples where rain also occurred. The frequency of elevation is given in Table 3.22 and again this is impressive.

3.5.3.4.7.4 Cloud Cover:- Cloud cover is probably an effect/ no effect variable, i.e. only full cloud cover at a low level will exhibit an effect. Samples with total cloud cover below a nominal 4,500 feet were studied. This amounted to 9 samples and

TABLE 3.22	FREQUENCY OF	INCREASE OR DEC	CREASE OF EL	EMTNTS (%) UNIVERSITY 79-4	3)
Element	Calm	Rain	Fog	Cloud Cover	
Br	7 5	46	71	78	
Co	66	51	64	7 8	
Cr	64	41	56	55	
Fe	7 0	46	57	67	
Mn	66	54	71	67	
Na	52	54	43	55	
SÞ	61	51	57	55	
Sc	48	36	21	67	
Se	59	41	50	33	
Zn	82	56	86	78	

TABLE 3.2	<u>3</u> ELEMENTS AFF	ECTED BY AMOUN	NT OI	F RAI	<u>E N</u>					
	(ATTICGPHERIC	PARTICULATES	-GN	ASGO	in: U	NIVE	RSIT	Y 75)-7'	S_)
Date	Amount of Rain	(mili itres)		Ele	ement	ts De	ecrea	ased		
3/										
4/8/72	5.7		Sc,	Na						
7/ 8/8/72	12.6		None	Ð						
9/ 11/10/72	9.4		Se,	Zn,	Co,	Na,	Br			
19/ 21/10/72	6.4	. (Cr, S	Sb, 2	Zn, (Co, 1	Fe, 1	Ja		
27/										
30/10/72	6.6		Se, Na,	Cr, Br	Sc,	SЪ,	Zn,	Co,	Fe,	Mn,
10/ 13/11/72	° 23.1		Se, Br	Cr,	Sb,	Zn,	Co,	Fe,	Mn,	Na,
17/ 20/11/72	17.0		Co,	Br						-
29/11/ 1/12/72	8.6		Se,	Cr,	Mn,	Na				
1/ 4/12/72	12.1		Sc,	Sb,	Zn,	Co,	Fe,	Na		
4/ 6/12/72	9.1		Se,	Cr,	Co,	Fe,	Na,	Br		
6/ 8/12/72	10.5		None	e						
11/ 13/12/72	23.2		Cr,	Sb,	Zn,	Co.	Mn			
13/ 15/12/72	10.4		Cr,	Sb,	Zn,	Co,	Mn			
30/12/72 2/1/73	14.3		Se, Na,	Cr, Br	s°,	Sb,	Zn,	Co,	Fe,	Mn,
15/ 17/1/75	16.5		Se, Br	Cr,	Sc,	Sb,	Zn,	Co,	Fe,	Mn,
19/ 22/1/73	24.8		Se, Na,	Cr, Br	Sc,	Sb,	Zn,	Üo,	Fe,	Mn,
22/ 24/1/73	7.1		Se,	Cr,	Fe					
26/ 29/1/73	9.9		Zn,	Co,	Fe					

I24

TABLE 3.23 (CONTD.)

Date	Amount of Rain (mililitres)		<u>E1</u>	emen	ts D	ecre	ased		
7/ 9/2/73	7.0	Se,	Na,	Br					
9/ 12/2/73	7.3	Se, Na,	Cr, Br	Sc,	Sb,	Zn,	Co,	Fe,	Mn,
14/ 16/2/73	6.8	Non	e						
23/ 26/2/73	7.1	Se,	Sb,	Co,	Fe,	Na			
2/ 5/3/73	6.9	Sb,	Zn,	Fe,	Mn,	Na,	Br		
30/3 2/4/73	12.8	Cr,	Zn,	Co,	Mn				
4/ 6/4/73	15.0	Zn,	Fe,	Mn					
6/ 9/4/73	4.9	Se,	Sc,	Sb,	Zn,	Mn,	Na,	Br	
30/4 2/5/73	19.6	Cr,	Mn						
4/ 7/5/73	19.8	Na,	Br						
11/ 14/5/73	8.6	Sc,	Sb,	Co					
21/ 22/5/73	5.9	Cr,	Sc,	Zn,	Co,	Fe,	Mn,	Na,	Br
18/ 20/7/73	30.4	Cr,	Sc,	Zn,	Fe,	Mn,	Na,	Br	
4/ 6/7/73	10.7	Sb,	Na						
13/ 16/7/73	19.4	Sc,	Sb,	Z n,	Mri,	Br			
18/ 20/7/73	11.3	Sb,	Mn,	Na,	Br				
20/ 23/7/73	13.6	Se,	Sc,	Zn,	Fe,	Mn,	Br		
3/ 7/8/73	17.1	Sb,	Zn,	Co,	Mn,	Br			
9/8/73	10.6	Se,	Sb,	Zn					
9/ 14/8/73	7.0	Non	е						
29/8/73	7.4	Se,	Sc,	Sb,	Co,	Fe,	Mn,	Na	

.

		4	ATHOSPHER	LIC	PAR	neu	LATE	2	
TABLE 3.24	<u>4 ELEMENTS AFFECTED</u>	D BY FOG	9LASGON	Vr	VIVER.	SITY	579-	<u>†3)</u>	
Date	<u>Visibility (yards)</u>	Duration	(hours)	<u>E</u> :	lemeı	nts l	Eleva	ated	
21/ 23/9/72	666	. 3		Zn,	Br				
26/ 28/9/72	800	4		Se, Na,	Cr, Br	Zn,	Co,	Fe,	Mn,
2/ 4/10/72	420	5		Se, Fe	Cr,	Sc,	Sb,	Zn,	Co,
5/ 6/10/72	235	2		Se, Na	Sb,	Zn,	Co,	Fe,	Mn,
17/ 18/10/72	332	6		Cr, Mn,	Sc, Na,	Sb, Br	Zn,	Co,	Fe
15/ 17/11/72	716	3		Se, Br	Sc,	Sb,	Zn,	Co,	Na,
8/ 11/12/72	337	. 9		Cr,	Sb,	Zn,	Co,	Mn	
1 7/ 19/1/73	220	19		Se, Mn,	Cr, Na,	Zn, Br	Sb,	Co,	Fe,
19/ 22/1/73	900	2		Se,	Mn,	Na,	Br		
22/ 24/1/73	443	10		Sb,	Zn,	Co,	Mn,	Br	
16/ 19/2/73	609	11		Cr,	Sc,	Zn,	Fe,	Mn,	Br
9/ 12/3/73	281	10		Cr, Br	Sb,	Zn,	Co,	Fe,	Mn,
14/ 16/3/73	413	4		Se,	Zn,	Fe,	Mn,	Br	
24/ 25/5/73	640	4		Non	e				

the results are given in Table 3.25. The effect, however, is obscured by calm, haze and foggy conditions prevailing during most of these samples and over the small number leaves little in the way of samples to be studied. However, the results are given and the frequency of elevation in Table 3.22.

On the whole this exercise was much more rewarding than the previous attempts, the expected trends being found. However, it was noted during the exercise that the trends were not linear and hence it was not thought prudent to subject these sub-samples to the computer programme previously used.

3.5.3.4.8 Multiple Regression Coefficient

The programme used was capable of handling all 30 variables simultaneously. The results obtained are (i) a multiple regression coefficient, i.e. the best straight line fit of a 31 dimensional graph; (ii) the number of values which lie on this line, i.e. the number of samples which can be explained by the variables; and (iii) an equation of the form:

 $y = m_1 x_1 + m_2 x_2 + \dots m_{30} x_{30} + C$ where y = elemental concentration

 $x_1 - - x_{30} =$ value of variables 1 - - 30 corresponding to y. $m_1 - - m_{30} =$ constant multiplication factor for that variable.

$$C$$
 = intercept on y axis

The correlation coefficients in general are much improved over the single variable approach and again Se which only appeared once in the whole of that section now has a regression

TABLE 3.25	ELEMENTS AFFECTED BY	CLOU	<u>D CO</u>	VER (ATH SHAS	<u>cs Pi</u>	1 ERIC VN	PF VERSIN	HRMC.	ulates 9-43)	
Date	Cloud Height (feet)			Ele	emen	ts E	leva	ted			•
3/ 4/8/72	4300	Cr,	Co								
17/ 18/10/72	3200	Cr,	Sc,	Sb,	Zn,	Co,	Fe,	Mn,	Na,	Br	
27/ 29/12/72	3140	Se, Br	Cr,	Sc,	Sb,	Zn,	Co,	Fe,	Mn,	Na,	
5/ 8/1/73	1375	Sc,	Sb,	Zn,	Co,	Fe,	Mn,	Na,	Br		
8/ 10/1/73	3840	Se,	Sc,	Sb,	Zn,	Co,	Fe,	Mn,	Na,	Br	
22/ 24/1/73	3015	Sb,	Zn,	Co,	Mn,	Br					
21/ 22/5/73	2525	Se	•								
22/ 23/5/73	1575	Cr,	Se,	Zn,	Co,	Fe,	Mn,	Na			
29/ 30/5/73	2600	Cr,	Sc,	Zn,	Fe,	Br					

I

TABLE 3.26 RESULTS OF MULTIPLE REGRESSION COEFFICIENT PROGRAMME

Manganese

multiple regression coefficient = 0.72						
number of samples explained	= 52%					
<u>Most Effect</u>	Least Effect					
Speed of East Winds	Duration of East Winds					
Temperature	Duration of NW Winds					
Speed of SW Winds	Duration of Fog					
Visibility	Duration of Rain					
Duration of SE Winds	Duration of NE Winds					
Number of Calm hours	Speed of NW Winds					

Bromine

multiple regression coefficient = 0.78							
number of samples explained	= 60%						
<u>Most Effect</u>	Least Effect						
Relative Humidity	Speed of W Winds						
Number of Calm hours	Duration of W Winds						
Atmospheric Pressure	Speed of SW Winds						
Duration of fog	Rate of rain						
Duration of SE Winds	Speed of East Winds						
Duration of NW Winds	Speed of NE Winds						
(Speed of NW Winds)							

.

ł

I29
<u>Zinc</u>

multiple regression co	efficient = 0.82
number of samples expl	ained = 68%
Most Effect	Least Effect
Relative Humidity	Temperature
Visibility	Duration of East Winds
Atmospheric Pressure	Duration of NE Winds
Speed of North Winds	Duration of SE Winds
Number of Calm hours	Vapour Pressure
Duration of Fog	Speed of South Winds

<u>Cobalt</u>

multiple regression coefficient = 0.42							
number of samples explained	= 18%						
Most_Effect	Least Effect						
Duration of Fog	Vapour Pressure						
Visibility	Relative Humidity						
Atmospheric Pressure	Number of Calm hours						
Duration of NE Winds	Speed of SE Winds						
Duration of S Winds	Speed of N Winds						
Duration of Rain	Duration of NH Winds						

. \

Iron

multiple regression coefficient = 0.39								
number of samples explained	d = 15%							
Most Effect	Least Effect							
Visibility	Duration of N Winds							
Duration of Fog	Speed of N Winds							
Duration of S Winds	Relative Humidity							
Duration of NE Winds	Speed of S Winds							
Atmospheric Pressure	Rate of Rain							
Duration of Rain	Duration of NW Winds							

Sodium

/

multiple regression coefficient = 0.61								
number of samples explained	= 37%							
Most Effect	Least Effect							
Duration of Fog	Duration of SE Winds							
Speed of SW Winds	Duration of W Winds							
Duration of NE Winds	Vapour Pressure							
Atmospheric Pressure	Speed of N Winds							
Visibility	Speed of East Winds							
Duration of Rain	Speed of NE Winds							

.

Selenium

multiple regression coeffici	ent = 0.49
number of samples explained	= 21%
<u>Most Effect</u> :	Least Effect
Duration of Rain	Speed of S Winds
Rate of Rain	Speed of SW Winds
Duration of NW Wunds	Temperature
Duration of S Winds	Relative Humidity
Duration of SE Winds	Duration of E Winds
Visibility	Duration of NE Winds

Chromium

multiple regression coefficient = 0.54								
number of samples explained	= 29%							
Most Effect	Least Effect							
Atmospheric Pressure	Amount of Rain							
Duration of SE Winds	Duration of W Winds							
Duration of Fog	Duration of SW Winds							
Speed of SW Winds	Speed of S Winds							
Duration of Calm	Speed of SE Winds							
Visibility	Duration of N Winds							

Scandium

multiple regression coefficient = 0.39								
number of samples explained	= 15%							
Most Effect	Least Effect							
Duration of Fog	Speed of N Winds							
'isibility Duration of Calm								
Atmospheric Pressure	Temperature							
Duration of S Winds	Vapour Pressure							
Duration of NE Winds	Duration of N Winds							
Speed of W Winds	Duration of NW Winds							
(Speed of NE Winds)								

Antimony

multiple regression coefficient = 0.60								
number of samples explained	= 36%							
Most Effect	Least Effect							
Speed of N Winds	Speed of NW Winds							
Duration of NW Winds	Rate of Rain							
Visibility	Speed of W Winds							
Duration of N Winds	Relative Humidity							
Duration of Calm	Duration of W Winds							
Duration of Fog	Speed of W Winds							

coefficient of 0.49. The coefficients range from 0.39 for Fe, to 0.82 for Zn. Mn and Br have coefficients of 0.72 and 0.78 respectively. This would be expected due to their frequency of appearance in the single variable approach.

The programme also computes errors in the estimation of the m values and in the majority of cases these were so high as to make these m values meaningless. This is due to the variables which are not affecting the levels and by elimination of these variables even higher coefficiences would be obtained. Unfortunately time did not allow this. However, from the student -T values calculated in the programme what can be stated is the variables which are having most and least effects. The 6 most and least effective are given in order in Table 3.26, together with the regression coefficients and the percentage of samples It will be noted that 9 out of the 10 elements have explained. visibility in the first 6, and 7 out of 10 have duration of fog. Surprisingly only 4 out of 10 have rain and 5 out of 10 have the number of calm hours, 4 of which had the best correlation previously (Sb. Zn. Mn. Br). It is also interesting to note that relative humidity is the main factor in the case of Br. (See Section 3.5.3.4.6).

3.5.4 Conclusions

The atmospheric system is so dynamic that no sensible conclusions car be drawn by simply examining the variables individually as if they were independent of each other.

There are two approaches which might be adopted -

 i) By making an empirical study of the data and selecting samples to suit chosen weather conditions. This allows human judgement to reject samples which have "interfering" conditions, but great care must be taken to remain as unbiased as possible.

The results of this study were extremely encouraging and supported what would have been expected, i.e. rain decreased levels while fog and periods of calm winds increased levels.

ii) By using a multiple regression programme to vary all the variables at once. This is a more scientific approach but does not allow for the intrinsic variability of atmospheric levels. The results, however, were extremely encouraging with correlation values much higher than in the single variable approach.

These correlation values could probably be further improved by a process of continually rejecting those parameters which are having least effect and eventually arriving at the precise conditions under which an element level will tend to rise or fall thus making predictions of possibly hazardous conditions much easier.

3.6 VARIATION IN ELEMENTAL CONCENTATION WITH LOCATION

An important part of air pollution monitoring is local fluctuations within a city and how these compare with sites outwith the city.

3.6.1 Sampling

The first problem in sampling is of course finding a site or in this case suitable sites throughout the city. Schools are an obvious choice and with the permission of the Education Authority several schools throughout the area were chosen. The city sites are illustrated in Figrue 3.9 and described in Table 3.27.

Ideally a number of sites should be set up and samples taken simultaneously to enable direct comparisons to be made. The Glasgow Smoke Officer has 15 permanent sampling sites for total particulate determination but unfortunately the samples were unsuitable for use in elemental determinations for the reasons -

i) The particulates were measured by a staining technique and therefore very dark stains were not desired, i.e. insufficient material.

ii) Glass fibre filter papers were used for sample collection.

Because of the two combinations, viz:- small sample size (about $0.34M^3$) and high filter blanks these samples had regrettably to be discarded.

The task of setting up 10-15 sites for simultaneous sampling was prohibitive and a three site comparison had to be used. One site was fixed (University) and the other two moved around at one month intervals.



- TABLE 3.27 DESCRIPTION OF SITES THROUGHOUT THE CITY USED FOR CONNECTING ATHOSYMERIC PARTICULATES
- G.U. This is the site at Glasgow University used as the reference site.
- MP., CC. These sites were chosen because they are in residential areas and lying on the south-west side of the City they should be least affected by City pollutants since prevailing winds blow from the south-west.
- B.A. This site is also residential but lies on the north side of the river and should therefore be more affected by City pollutants than MP or CC.
- WS. KS, BP These sites lie to the north of the Clyde and were chosen to study the effect of the various industries (shipbuilding, iron works, engine works, etc.) along this section of the river on atmospheric levels of elements. BP also lies across the river from the large coal burning Braehead Power Station.
- EH, KP These sites were chosen to study the effect of the large Clydebridge Steelworks/Clyde Iron Works complex on the atmospheric levels of elements. KP lies to the southwest and EH to the north-east of the complex.

- SS, SR, OL All these sites lie on main arterial roads into the City centre and were chosen for this reason. SR and OL are nearer the City centre than SS. SR is also close to six different metal works.
- QP This site lies on the same road as OL but further away from the City centre. It also lies to the north of the iron and steel complex mentioned above.
- S.A. This site lies on the north edge of the City and should reflect the accumulation of elements from the prevailing south-west winds as they blow over the City.

Since fluctuations in different areas were being studied, weekly samples were taken at roof top height. This was best achieved using a Charles Austin Capex Mark II pump which gave samples of approximately 45M³ over the 7 day period. Although this is a larger volume than used in Section 3.5 the slower sampling rate prevented the filters from clogging. This is illustrated in Figure 3.10 which indicates a uniform sampling rate over the sampling period.

Filter papers were changed every Monday morning.

3.6.2 Analysis

To extend the range of elements studied these samples were analysed by atomic absorption spectroscopy using a Perkin Elmer double beam 306 atomic absorption spectrophotometer with deuterium background corrector with attachments for both flame and HGA - 74 flameless systems. 11 elements were detected.

When dissolving samples for atomic absorption care must be taken to prevent the loss of volatile elements such as Pb and Cd. The two methods are wet and dry ashing.

Dry ashing at 500-600°C can result in severe loss of some metals (255). However, low-temperature ashers are now available which oxidise the carbonaceous matter at 100-120°C using an excited oxygen plasma, but this was not available for this work although it has been used elsewhere (53, 58, 73, 155, 159, 255).

Although wet ashing can also result in losses on a smaller scale it was used in this case. However, to reduce the risk of loss the filters were dissolved under pressure in teflon dissolution bombs with screw caps. A mixture of HNO_3 and H_2SO_4 was used as culphates



I4I

are generally less volatile than nitrates, thus reducing the risk of loss of volatile elements. HF was also added to dissolve the silicaceous material which has been shown to be present in atmospheric particulates (115). Also Kronborg <u>et al</u> (116) found that atomic absorption results for Fe in air particulates gave systematic low results compared to neutron activation analysis when filter papers were leached only with nitric acid since some of the Fe was in the silicate matrix.

An oven was found to be unsuccessful as the bottom of the bomb resting on the shelf and therefore receiving direct heat tended to expand out. This was overcome by using a sand bath and building the sand up round the sides of the bomb.

3.6.2.1 Procedure

The whole filter paper was placed in the previously acid washed teflon bomb, together with 5 mililitres concentrated Aristar H_2SO_4 and 3 mililitres concentrated Aristar HNO_3 and heated on a sand bath at 250°C for 30 minutes to destroy all organic matter. The bomb was thoroughly cooled, 1 mililitre of concentrated Analar HF added and again heated for 30 minutes. After being thoroughly cooled the sample was transferred to a 25 mililitre volumetric flask, made up to the mark with deionised water and stored in acid cleaned plastic bottles.

Standard solutions were prepared daily from stock solutions and in the same acid matrix as the samples.

3.6.2.2 Accuracy

The accuracy of this method was tested by analysing three standard air filters (241). Where the concentration range allowed, flame techniques were employed as it is the quicker of the two methods.

The values and range due to error are given in Table 3.28 and those obtained by analysis in Table 3.29. The results illustrate that Cu, Mn, Ni, Zn can be analysed with confidence by direct flame techniques, and Cd and Cr by direct HGA injection. However, Fe and Pb when prepared in this way gave low results by direct flame but when diluted 10 times for Pb and 100 times for Fe and analysed by HGA accurate results were obtained. This can only be explained by the dilution removing a suppression interference on This is in agreement with Quickert et al (256) who these elements. found severe interferences in the determination of V by flame due to Al, Si, Zn, K and Na but no interference at all when analysing by HGA.

The ninth element determined, Al was not spotted on the filters and presented a problem as to whether or not it could be determined by flame techniques. To be safe therefore samples were diluted 1,000 times and analysed by HGA.

It was found in the analysis of the samples that in fact only Zn could be determined by flame techniques, Ni, Mn and Cu being too low in concentration. These last 3 elements were therefore analysed by HGA which will in general add 5% on to the error. The accuracy of the determinations was therefore -

	"True Value (mg)	Range (19)
Cd	3.06	2.91-3.21
Cr	1.85	1.76-1.94
Cu	3.04	2. 89 - 3.19
Fe	194	184.3-203.7
Mn	4.2	4.15-4.41
Ni	5.30	5.04-5.56
Pb	153	145.4-160.6
Zn	74	70.3-77.7

TABLE 3.29		ELEMEN	TAL CONC	ENTRATIONS FO	UND IN ST	<u>CANDARD</u>	FILTERS (Mg
		BY	HTCHIC	ABSCRPTICN	SPECTROS	KUDA.	
	Flame				1	<u>H.G.A</u> .	
Cu	3.00						
	3.25	3.0)8 -0. 14				
	3.00						
Mn	4.25						
	4.50	4.3	33-0.14				
	4.25						
Ni	5.3						
	5.1	5.1	17-0.12				
	5.1						
Zn	75						
	75	74.5	7 - 0.58				
	74						
Cd						3.17	
						3.02	3.10 - 0.08
						3.12	
Cr						1.77	
						1.87	1.85-0.08
						1.92	
Fe	150		 		19	2	
	147.5	148.3	3 - 1,44		19	0	192.3-2.52
	147.5				19	5	
Pb	125		.'		15	3.7	
	122.5	124.2	2-1.44		15	3.7	152 .5-2. 14

125

150.0

.

.

I45

)

Cd,	Cr,	Zn		5%	
Ni			-	10%	
Cu,	Mn		-	14%	
Fe,	Pb,	Al		18%	

The Al was estimated to be of the same order as Fe.

3.6.2.3 Multiple Peaks

During the analysis of elements by HGA a problem of split peaks arose. Figure 3.11 illustrates a 20 μ l injection of 0.2 ppm Cr when in an acide mixture of 3 ml H₂SO₄/3ml HNO₃. This problem was encountered with most elements.

Because of the greater volatibility of nitrates it was considered that the two peaks illustrated the differing volatibilities of the sulphate and nitrate compounds which would be deposited on the carbon rod: as the sample was dried. This is in agreement with Matowsek <u>et al</u> (174) who when analysing for Pb in air particulates ashed the filter paper directly in the especially designed collection cup, and obtained multiple peaks due to the different forms of Pb in the atmosphere, e.g. bromide and oxide. Matowsek <u>et al</u> found that by adding 1000 ppm orthophosphoric acid to the samples single peaks were produced. They interpreted this as being due to the formation of the more stable phosphates.

Figure 3.11 illustrates the effect of adding 20, 40, 60 and 80 μ l of 1000 ppm H₂SO₄ to 20 μ l of 0.2 ppm Cr solution. It can be seen that the first peak (nitrate) is becoming smaller as excess sulphate is added. At 60 μ l the peak is single except for a small shoulder. The 80 μ l peak is smaller probably because the suppression of the signal by the sulphate is now severe.



This method, however, proved to be very expensive in terms of carbon rods and was therefore not suitable for routine analysis. In order to ensure an excess of sulphate in the sample the acid mixture was changed to 5 ml H_2SO_4 and 3 ml HNO_3 . Also two drying stages were used -

- i) just above the boiling point of HNO₃ which ensured that all nitrates had been removed.
- ii) just above the boiling point of H_2SO_4 the excess sulphate ensuring that only the sulphate of the element was present.

It must be said that this argument does not agree with the theory given previously where sulphate and nitrate are said to decompose to oxide and then reduction by carbon to metal and carbon monoxide. By this theory, even in a mixture of nitric and sulphuric acids, both should decompose to the oxide and then to the metal hence giving only one peak.

3.6.3 Results

The results are given in Table 3.30, together with results from other industrial sites in the world. The Glasgow results represent the average of 96 samples taken over 14 sites. The Sn and V figures are based on the analysis of 12 samples chosen at random.

It will be noted that the levels of Cr, Fe, Mn and to a lesser extent Zn in this section are higher than the previous levels. However, it is not strictly fair to compare the two sets of results directly due to the latter being the average of 14 sites. Moreover these latter results represent mainly winter months which will tend to elevate the levels.

3.6.3.1 Correlation Between Elements

The samples taken at Glasgow University were used to avoid variability of site and amounted to 37 samples. Using the same programme as previously (Section 3.5) the elements were plotted against each other and straight line correlation coefficients calculated. The results at the 95% and greater confidence level are given in Table 3.31. Corresponding results from Section 3.5 are also given for comparison. The Fe/Mn coefficient is much improved, Fe/Cr is about the same and the rest are all lower. The Zn/Mn coefficient in Section 3.5 was the highest correlation with other elements and here, although reduced is the only Zn coefficient above the 95% level, maintaining at least the position of Mn as being the best correlation with Zn.

There could be several reasons for these differences -

TABLE 3.30	<u>TA</u>	MOSF	HERI	C LE	VELS	OF	TRAC	<u>E EL</u>	EMEN	ts ($ng/M^3)$
<u>Industrial</u> Limits (mg/M ⁷) (244)	None	0.02	0.1	1.0	10	5	1.0	0.2	1.0	0.1	0.5
Glamorgan Wales (117)	364	. 22	13.5	48.6	609	24	65.6	253	367	I	24
<u>Paris</u> France (243)	1	19.5	15.1	ı	3500	82.5	I	1	I	ı	ı
San Francisco U.S.A. (121)	250-2000	I	2.2-16	27-100	610-3000	5-31	I	I	27-500	0.08	1.5-11.0
Texas U.S.A. (242) (50 Stations)	I	ł	50-1100	10-960	100-10,000	20-1700	IO	20-2560	5-1770	I	4
<u>Heidelberģ</u> Germany (129)	I	26.8	4.6	1	1041	23.6	I	I	I	71.6	I
<u>Miles</u> <u>U.S.A.</u> (124)	1200	ł	9.5	270	1900	62	I	I	140	I	5.0
<u>Chicago</u> 11. S. A. (124)	2175	ı	113	1150	13800	255	60	I	1400	I	18.1
<u>Row</u> Mean	o 6588	13.6	35.2	63.5	1209	109	128	260	388	< 0.7	<7.0
Glas (14 Si Range	1300-1800	0.2-407	2.7-163	0.7-665	233-6680	17.5-829	1.4- 581	13-1060	5-2400		
	ΤV	Cđ	Сr	Cu	ъе	Mn	Ϊ	$\mathbf{P}_{\mathbf{D}}$	Zn	Sn	Δ

- = Not Determined

TAB	LE 3.3	<u>1 CO</u>	RRELAT	ION	COEFF	ICIENTS	<u>S OF E</u>	LEMENTS	WITH EACH OT	HER	
		(AT	HOSPHE	PIR	PARTI	CULATE	s- C	LASSOW	UNIVERSITY	-SITE.)	
	í										
Al	Al										
Cd	-	Cd									
Cr	0.56	0.59	Cr								
Cu	-	0.50	-	Cu							
Fe	0.83	-	0.67	-	Fe						
Mn	0.67	-	0.56	-	0.84	Mn					
Ni	0.67	0.29	-	-	0.34	-	Ni				
Рb	0.28	0.43	0.46	-	-	0.45	0.50	Pb			
Zn	-	-	-	-	-	0.42	-				

Correlation Coefficients of Elements in Section 3.5 (9.94.T.3.10) (ATHOSPHERIC PARTICULATES - & LASCON UNITERS 17 '79-43)

- Cr Cr
- Fe 0.58 Fe
- Mn 0.27 0.30 Mn
- Zr 0.43 0.44 0.69

- i) As was shown in Section 3.5 there is a relationship between elemental levels and weather conditions although this is not linear. Over the longer sampling period used in Section 3.6 therefore the levels will represent the average effect of weather conditions, some elements being more affected than others.
- ii) A smaller number of samples was used in Section 3.6 (37)
 compared with Section 3.5 (158). Therefore small
 discrepancies will carry more weight in the correlation
 coefficient of Sec tion 3.6.

The most curious result in Section 3.6 is the total lack of relationship between Zn and Cd as they are closely related in their natural minerals. The correlation coefficient is 0.00047 with a confidence level well outside the 95% level. This could of course be explained by the reasons given above but the total lack of relationship is still striking.

Cr, Fe, Mn and Ni all have good relationships with Al and this is consistent with the fly-ash source theory of Section 3.5. Pb, Zn, Cd and to a lesser extent Cu represent those elements with the highest enrichment factors (see next section). Again the lack of relationship to Al is consistent with these elements having additional sources.

3.6.3.2 Enrichment Factors

Enrichment factors were calculated on each individual sample based on Fe to allow comparison with Section 3.5. The results, averaged for each site, are given in Table 3.32. The Glasgow average (14 sites), the range from individual samples, and those of Dams <u>et al</u> for Liege (Belgium) (251) are given in Table 3.33.

<u><u><u></u>TAÈLE</u></u>	3.32	ENRICHM	ENT FACT	ORS FO	OR GLASGO	W SITE	S	
		(ATHO	SPHERIC	PART	ICULATES	(BAS	ED CN	Seil)
<u>Site</u>	Cu	Ni	Zn	<u>A1</u>	Cd	Mn	Pb	Cr
GU	135	76	337	2	11050	4	880	12
BA	133	232	123	5	7930	5	1790	9
QP	78	1 7 1	439	3	3630	5	1080	9
KS	14	293	227	5	1920	3	1570	8
BP	363	146	233	6	13990	7	1830	20
CC	4	323	126	3	12360	8	2730	15
ΚŖ	2	. 87	335	3	3700	4	1.330	10
SS	5	80	427	2	3150	5	7 00	8
SR	26	43	205	2	2460	2	820	6
EH	39	61	243	2	5060	3	520	15
OL	23	52	147	4	1710	5	270	5
WS	83	222	134	4	3 4290	2	910	11
SA	405	146	232	5	29180	3	940	14
MP	42	181	219	4	10890	4	1440	17

TAB	LE 3.33 AV	ERAGE ENRICHMENT	FACTORS FOR GLA	SGOW AND LIEGE
	(A	THOSPHERIC PART	ICULATES.)	
	<u> </u>			
		Glasgow	Liege	(Basedon Al)
	Range	Average		
Al	0.3-7	4	1	
Cd	250- 82870	10100	41000	
Cr	5-46	. 12	-	
Cu	1-2005	96	200	
Fe		נ	. 3	
Mn	0.4-20	4	. 5	
Ni	2–528	151	• _	Ň
Pb	80-5500	1200) _	
Zn	5-954	245	2100	

From Table 3.33 the first point which must be made is that for Mn and Cr normalised to Fe enrichment factors are basically the same as in Section 3.5, although the actual levels were higher in Section 3.6. In although of the same order tended to be higher in Section 3.5, although the value does lie within the Section 3.6. range. These results are encouraging in that they should allay fears of there being an unknown factor involved which basically alters the aerosol collected and analysed from Section 3.5 to that from Section 3.6.

The results are again in good agreement with Liege which apart from Zn lie within the range of Glasgow.

The first striking feature about Table 3.32 is the uniformity of enrichment factors for Mn, Al, (Fe), and to a lesser extent Cr and Zn throughout the city. The first four elements are again those which are found in the inorganic fraction of coal and as already said Zn is also in this group but has additional industrial Sources which tend to raise the enrichment factor. Cu and Ni tend to be more variable. In at least four cases with Cu (CC, KP, SS, KS) the enrichment factors are not so very far from soil ratios and another six (QP, SR, EH, OL, WS, MP) have enrichment factors below 100. Ni tends to be higher with only six below 100 and none below It is probable therefore that Ni is predominantly from 50. pollution sources other than coal buining but that Cu is in some cases purely fly-ash in origin.

An attempt at explaining the variation in Cd, Pb, Cu and Ni with respect to the area was soon abandoned. The difficulty is that the samples were not taken simultaneously and the variation could reflect wind direction at time of sampling or variability of output, and is one of the disadvantages of this type of sampling.

3.6.3.3 Variation of Elements with Sites

As said previously samples were taken around the city and referred back to Glasgow University as a reference site.

It is safe to say that there is no statistical difference between areas but a dynamic system was being studied with no two samples from the same site giving the same reading. Statistics in this context are therefore meaningless and only general trends can be concluded. The average levels are given in Table 3.34 and the order of ratios to Glasgow University, together with the ratio at the sites in Table 3.35. It is here that a lack of statistics is a hindrance, e.g. with Zn the ratio starts at 0.2 and progresses through to 1.0 with only 0.6 absent, making it difficult to judge which if any are truly different. However, certain points can still be made.

- i) No site is consistently lower or higher than G.U. for all 9 elements.
- ii) WS which lies in the direct flight path of the airport has the greatest ratios for Cu, Ni, Cd and Cr, and is well up the list for Al, while Pb is not much different from G.U.
 Klein <u>et at</u> (6) when studying metals in soils found that Cd, Cr, Cu, Ni, Fe and Zn were higher around an airport but Pb was not. Five of the elements from WS agree with this, while Zn is the lowest of all sites at WS and Fe is about the same as G.U. General trends of agreement can be noted.
- iii) Areas which were thought to have no pollution sources (CC, MP) are not consistently low. Only for Cu at CC is the ratio the lowest of the sites. In fact MP-Fe, Ni, Al and Cr ratios are the second highest and Pb and Cd the third highest.

TABLE	<u>3.34</u>	AVERA	osshe	VELS OI	F ELEMENTS	<u>AT CI</u>	ITY SII	ES (Je	<u>s/M³)</u>	
<u>Site</u>	<u>Cu</u>	Ni	Zn	Fe	<u>A1(P3)</u>	Cd	<u>Mn</u>	<u>Pb</u>	Cr	
BA	41	158	95	600	5.6	6.6	63	298	14	
QP	43	178	507	1084	6.8	5.2	111	29 7	23	
KS	15	395	131	1391	14	3.2	90	481	29	
WS	23	199	127	876	5.6	26	59	211	31	
MP	27	236	411	1270	9.7	18	116	434	101	
SA	201	144	297	953	7.8	40	48	230	35	
BP	189	89	114	416	4.2	8.5	68	65	21	
CC	0.9	152	113	586	3.6	12	148	261	22	
KP	0.7	55	211	681	3.3	4.0	66	286	23	
SS	2.0	94	240	839	3.8	3.7	87	156	18	
SR	15	68	301	950	4.0	8.8	42	191	16	
EH	28	61	339	930	3.0	3.9	72	124	41	
OL	72	109	377	1980	12	5.0	142	680	3 8	

.

TABLE 3.35 ORDER OF SITE TO GLASGOW UNIVERSITY RATIOS (OF ELEMENTAL CONCENTRATIONS IN ATMOSPHERIC PARTICULAR
<u>Fe</u> EH = SR = OL = BP CC SA = BA = QP KP WS = SS <u>GU</u> MP KS 0.4 0.6 0.7 0.8 0.9 1.1 1.6
<u>Cu</u> CC QP KS SS $BA = KP$ <u>GU</u> = MP BP = OL SR EH SA WS 0.03 0.1 0.2 0.3 0.5 1 1.5 1.6 3.1 3.3 21
<u>Ni</u> EH = OL = KP SR = BP CC = SS <u>GU</u> BA QP KS SA MP WS 0.5 0.7 0.9 1.3 2.1 2.5 8.0 26 95
$\frac{Zn}{0.2} \text{ WS} = BA \text{ KS} = BP = 0L \text{ SA} = CC \text{ QP } MP = KP \text{ SS } SR \frac{GU}{0.4} = EH$ $0.2 0.3 0.4 0.5 0.7 0.8 0.9 1$
<u>Al</u> EH OL = SR CC BP BA <u>GU</u> SS KP QP SA WS MP KS 0.3 0.4 0.6 0.7 0.9 1.1 1.2 1.5 1.8 2.1 2.9 3.4
\underline{Cd} $BP = QP$ $OL = KS$ $BA = KP = EH = SS$ $\underline{GU} = CC$ SR MP SA WS 0.1 0.4 0.6 1 1.4 4.4 5.3 19
$\underline{Mn} OL = SR SA = EH BA BP MP \underline{GU} = QP = WS = KP SS CC KS$ 0.2 0.4 0.6 0.8 0.9 1 1.1 2.5 3.3
\underline{Pb} BFSS= EHSASRKP \underline{GU} BA= QPWSCCMPKSOL0.40.50.70.80.91.11.31.41.92.05.0
<u>Cr</u> SS BP = SR SA BA = OL QP CC EH <u>GU</u> KP KS MP WS 0.3 0.4 0.5 0.6 0.7 0.8 0.9 1.6 2.0 2.3 2.6

•

iv) OL is near the centre of the city and lies on a main arterial road. This is reflected by being highest on the list for Pb.Conversely BP is probably the most residential of the sites and has the lowest Pb ratio.

3.6.3.4 Effect of Wind

As in Section 3.5 wind direction was calculated from hourly observations at Glasgow Airport to give the total hours the wind was blowing from the eight compass points. Samples were then selected such that the wind was blowing from one site over another. Some sites could not be used at all since the wind direction was never in the right direction, especially the less common winds, e.g. east winds. The lack of wind in this direction disallowed the use of the G.U.-SR-EH system. The results are given in Table 3.36. The following points can be made -

- (a) There is a general rise in all elements from CC to G.U. except Pb and Mn. Pb was generally lower at G.U., but with Mn three out of four samples were higher at G.U. than at CC maintaining the same trend as other elements.
- (b) All G.U. results are higher than SS.
- (c) All SS results are higher except for Pb and Cu, and Al and Cd are about the same at both sites. The two Cu readings for KP were 0.8 and 23.6 and an average for such is not reliable.
- (d) All MP results are higher due to NE winds and this is a part explanation of why MP levels were higher than expected.
- (e) All WS results are higher except for Zn (Zn had the smallest ratio of all sites) and Mn which is about the same at both sites.

TAB	LE 3.36	EFF	ECT OF	WIND	DIRECT	ION ON	ELEMEN	TAL LE	<u>VELS (n</u>	$g/M^3)$
		(A-	THUSPI	HERIC	PAIX	IRVILATE	<u>(</u>			
		(a)	<u> CC –</u>	GU – A	verage	of fou	r SW w	ind Sa	mples	
	Cu	Ni	Zn	Fe	Al	Cd	Mn	Pb	Cr	
CC	0.9	152	113	586	3.6	11.8	148	261	22.5	
GU	34.4	165	258	918	6.4	12.0	60	185	26.5	
		•								
		(b)	<u> </u>	SS – A	verage	of two	E win	d Samp	les	
GU	19.0	110	3 09	1035	4.2	6.3	93	309	19.7	
SS	2.7	87	184	770	3.9	3.9	80	139	17.1	
		(c)	<u>KP - </u>	<u>SS – A</u>	verage	of two	SW wi	nd Sam	ples	
SS	1.0	88	802	931	3.9	4.5	94	194	19.5	
KP	12.2	64	242	806	4.0	4.3	75	226	17.3	
										·
		(d)	<u>GU - 1</u>	MP – A	verage	of two	NE wi	nd Samj	ples	
GU	0.7	1.5	267	620	2.0	2.0	34	98	6.0	
NP	13.0	142	316	962	6.3	16.9	68	363	46.7	
		(e)	<u>GU - 1</u>	WS - A	verage	of two	SE wir	nd Sam	ples	
GU	0.7	1.5	267	620	2.0	2.0	34	98	6.0	
WS	8.7	240	175	743	5.8	7.6	36	197	16.4	
	x									
,		(f)	<u>GU - 1</u>	BA – O1	ne NW	wind Sa	mple			
GU	89	76	301	715	5.3	11.7	66	200	17.6	
BĄ	61	92	63	483	3.2	13.7	36	177	14.6	
	• .									
		(g)	<u>Gu -</u>	QF - A	verage	of two	E/SE	wind S	amples	
GU	192	110	1038	1740	5.3	242	14.9	320	41.8	
QP	36	164	654	935	4.8	6.4	10.5	225	25.4	

; -

(f) All results except Ni and Cd are higher at G.U.

(g) All results except Ni are higher at G.U.

From these tables, as expected, in general the elements are increasing in concentration as the wind blows over the city and eventually out into the surrounding countryside.

3.6.3.4 Comparison of City with Rural Areas

Sites were set up at Ayr Academy (AA) which is one street removed from the West coast and at Loudon Academy, Galston, (LA) about 12 miles from the coast and half-way between Ayr and Glasgow.

The average results for 4 weekly samples are given in Table 3.37.

As expected Ayr levels are much lower than Glasgow except for Fe and Al which are about the same. LA tends to be more variable. However, after sampling it was discovered that due to the shape of the surrounding hills there was a funnelling effect carrying pollutants from Kilmarnock - a semi industrial town. This tends to obscure any trends, although, except for Cu and Al the levels are still lower than Glasgow. Al is about the same but Cu is much higher.

Enrichment factors were calculated for AA, LA, and the corresponding G.U. samples. The results are given in Table 3.33. Ayr enrichment factors are much closer to soil levels indicating the rural setting of this holiday resort. There is some light industry and this is probably responsible for the slight elevation of Zn and Cd. As would be expected Pb is also elevated but not so Euch as Glasgow thus reflecting the lower density of traffic in Ayr.

TAB	LE 3.37	ATMOS	SPHERIC	LEVEI	<u>S OF E</u>	LEMENT	S IN	RURAL	SITES	(ng/M ³)
	(وس) 🗚	Cd	Cr	Cu	Fe	Mn	Ni	Pb	Zn	
GU	2.3	9.6	63.4	16.0	1220	216	68.7	398	703	
LA	2.6	3.7	11.8	70.0	621	44	36.6	122	123	
AA (2.5	4.0	11.2	2.6	911	33	21.3	107	153	

TABLE 3.38	<u>ENRICHMENT</u>	FACTORS FOR RURA	L SITES
	(ELEMENTS	IN ATMOSPHERIC	PARTICULATES)
	Ayr	Galston	Glasgow
Al	1	2	1
Ca	2870	4336	5744
Cr	5	8	20
Cu	5	171	25
Fe	1	1	1
Mn	2	3	8
Ni	21	52	53
Pb	4 50	77 8	1250
Zn	130	147	427

Galston shows the same trends as Ayr except for Cu which is more enriched than Glasgow and Ni which is the same. Enrichment factors at Galston tend to be higher than Ayr probably reflecting the effect of industry at Kilmarnock.

3.6.4 Conclusions

Because of financial restrictions simultaneous samples could not be taken and a system of reference sampling based on the University was adopted. This of course complicates any deductions which may be made from the results but does not entirely invalidate them.

The levels of elements, as expected, show a wide range of values throughout the city but no area is consistently below or above the University level. The enrichment factors do, however, remain relatively constant for Mn, Al, Fe, Cr and Zn, i.e. the elements found in the inorganic fraction of coal and are close to the soil ratios with the exception of Zn. The reasons for this have already been discussed. Cd, Pb, Cu and Ni have generally higher enrichment factors and fluctuate throughout the city. These elements represent additional industrial sources.

On comparison with rural settings, as expected, Glasgow levels tend to be higher with the enrichment factors of the rural sites closer to the soil ratios.

Finally from a wind study it can be seen that the atmospheric levels generally increase as the air blows over the city and as was pointed out in Chapter I this could lead to problems in agricultural land by the settling out of particles as the wind speed drops.

3.7 PARTICLE SIZE DISTRIBUTION OF ELEMENTS IN THE ATMOSPHERE

The size of particulate matter in the atmosphere is important to both plants and animals. The smaller the particle the easier it will be for it to enter leaves directly through the stomata and to penetrate further into the human lung.

Inhalation and retention of particles in the lung in general induces a physiologic response reflected in a slowing of the ciliary beat. The clearing action of the lungs depends on the beating action of the cilia to sweep foreign particles upwards and out of the respiratory tract into the back of the throat where they are automatically swallowed (i.e. the pollutant will then act as if swallowed). Decrease in efficiency of activity of the cilia can cause pollutants to become fixed at a particular point in the tract thus increasing the likelihood of them creating a toxic effect. Also the slower the beating action the slower the upward sweep of matter. In time this would increase the residence time of foreign matter and therefore increase its ability to exert a toxic effect on the respiratory tract lining.

Not only is this a problem with the particles as chemical species but as physical entities they adsorb and absorb gases and solutes therefore increasing their effect by holding them more focally in deeper parts of the lung for longer periods.

Particle deposition within the respiratory tract is affected by

- i) Size of particle.
- ii) Air velocities and transit times of air within the system.
- iii) Obstructions and baffles encountered thus changing the . direction of air flow.

iv) The forces necessary to displace the particles sufficiently to make them impinge on to the surfaces of the respiratory tract lining.

The size believed to cause the most injury is the particle with an effective diameter of less than **)**. Smaller particles seem to be less readily removed from the lungs therefore adding additional dosage and residence time. Figure 3.12 illustrates the positions in the respiratory tract at which particles of various diameters are retained (2).

3.7.1 Sample Collection

In order to investigate the size distribution of atmospheric particles the sample must be fractionated. This can be achieved through the use of a cascedeimpactor. A four stage impactor with a back up filter was used in this work and is illustrated in Figure 3.13. However, impactors with up to nine stages are also commercially available. Millipore filter papers were used as the collection medium since the blank values were already known.

As the sample is now effectively collected over five filter papers a large sample volume is required. Since 20 M^3 was required for the analysis of the total sample, 100 M^3 was collected over a period of 15 days in July 1975 at a flow rate of 17.5 litres/minute.

3.7.2 Principles of Impaction

Impaction may be defined as the process whereby, under suitable conditions, an aerosol particle in an air flow collides with a surface and sticks.


FIGURE 3.13

GRAPHIC ILLUSTRATION OF CASCADE IMPACTOR



One way to cause a particle to impact is to make the air stream bend sharply. If the bend is sharp enough the inertia of the particle causes it to continue in a straight line so that it leaves the air stream and impacts on the filter paper. This process is shown in Figure 3.14. In order to keep the particle moving with a curved streamline a radial force of $F_r = \frac{mV^2}{r}$ must be exerted on the particle (m and V are the mass and velocity respectively and r is the radius of curvature of the streamline). This force arises from collisions with the surrounding air molecules, which are more densely packed nearer the impaction surface and thus exerts a net force away from it.

It is easy to see that large particles must impact more easily than the small particles. The radial force needed to keep a particle in the streamline varies with m or with R^3 (R = radius of particle) but the number of collisions of the particles with air molecules varies with the cross-sectional area of the particle or only with R^2 . Thus the larger the particle the more efficiently it will impact. This and all the other factors controlling impaction have been combined into an inertial impaction parameter K -

where **G** = particle density

d = particle diameter

V = particle velocity

 μ = air viscosity

D = jet diameter.

Large values of K indicate efficient impaction.



Another impaction parameter is the cut-off diameter, or the size for which 50% of the particles will be impacted on to a surface. It is usually considered as the minimum collection diameter and may be calculated from the formula -

$$d^{2}_{\min} = \frac{36\mu D}{\pi \sigma V}$$

where d_{\min} = the cut-off diameter.

Because the largest particles are most effectively impacted it is easy to see that by subjecting an air stream to a series of impactions progressively smaller particles may be removed. To do this it is only necessary to increase the velocity of the air stream and/or decrease the diameter of the jet with each successive stage. This means that these instruments must be operated at the flow rate stipulated to achieve the size of separation quoted.

3.7.3 <u>Analysis</u>

The samples were digested in the acid mixture and analysed by atomic absorption spectrometry for 9 elements as described in Section 3.6.

3.7.4 Results

On visual inspection stage 1 carried virtually no particulate matter and was not analysed. The general order of staining was $5 > 4 \approx 3 > 2 > 1$ although 5 may be misleading due to the collection area being larger.

Table 3.39 illustrates the cut-off diameter, range and amount of each element collected at each stage. In each case the total concentration collected lies within the reported range of elements collected routinely in Section 3.6 (see Table 3.30). The results are also shown graphically in Figure 3.15.

3.7.4.1 Inhalation

The amounts of each element collected at each stage are illustrated as a percentage of the total element collected over all stages in Table 3.40.

With reference to Figure 3.12 it can be seen that the more toxic elements have large proportions below 1 μ m diameter and therefore will pass easily into the alveoli. These are - Ni (88%), Cd (77%) and Pb(71%). The other element fractions which will enter the alveoli are Zn(83%), Cr(54%), Mn(53%), Fe (43%), Cu (35%) and Al (29%).

The remainder of each element will penetrate at least as far as the bronchi and some as far as the bronchiole.

The implications of these results are clear - any process which is installed to reduce the emission of elements into the atmosphere must be able to remove the smallest particles as a large portion of the elements are concentrated in this size range. Two recent TABLE 3.39 CONCENTRATIONS OF ELEMENTS AS A FUNCTION OF PARTICLE SIZE (ng/M³)

,

(ATHOSPHERIC PARTICULATES - GLASSON UNIVERSITY.)

1.

	Cut off Diameter (µm)	Range (Jum)	ΓY	Cđ	Gr	Сu	ЭŦ	MIn	ΪN	qd	Zn
Stage 1	12.0	6.3 - 17.0	1	ı	I	1	ł	ł	1	1	I
Stage 2	4.4	2.0 - 6.3	847.1	0.3	3.3	5.8	147.3	2.7	3.4	54.6	147.6
Stage 3	1.5	0.7 - 2.0	786.0	0.2	3.6	7.1	176.8	2.8	5.2	98.0	88 . 8
Stage 4	0.45	0.43- 0.7	252.3	1.1	1.5	3.3	87.4	2.9	42.9	172.8	1036.0
Stage 5	I	0.43	418.0	0.6	6.6	3.5	159.3	3.3	22.8	203.1	76.4
Total			2303.4	2.2	15.0	19.7	570.8	11.7	74.3	528.5	1348.8

TABLE 3.40	C	ONCENTRATIC	NS (OF ELEMEN	IS COL	LECTED A	T EACH	STAGE	
						TERCENT	AGE OF	TOTAD	
	Al	Cd	Cr	Cu	Fe	Mn	Ni	Pb	Zn
Stage 2	37	14	22	29	26	23	5	10	11
Stage 3	34	9	24	36	31	24	7	19	6
Stage 4	11	50	10	17	15	25	58	33	77
Stage 5	18	27	44	18	28	28	30	38	6
רא צי תווסאות	ħ	ACC MEDIAN			รา รพร		ᢄᢑ᠊ᠣᢕ᠌᠌ᠹ᠊ᠬ	FD ይ ልክሮፑ	q
IRDUE 2.41	M	AOO-NEDIAN	DIA	UBTERS UT	<u>e perie</u>	MIN VIN	IUSE VILL.		
			_/		-		· · /+ -	- 1	

Element	MMD (um)	$\frac{\text{Range}(\mu m)}{(2 + 15\%)} (+ 15\%)$
Al	2.1	4 - 7.0
Cu	2.0	0.6 - 7.0
Fe	1.7	1.5 - 6.0
Cr	1.4	1.5 - 7.0
Mn	1.4	1 - 7.0
Zn	0.8	0.7 - 7.0
Cd	0.6	1.5 - 10.0
Ni	0.6	1.0 - 1.5
Pb	0.5	0.2 - 0.7

* ATMOSPHERIC PARTICULATES - GLASGOW UNIVERSITY.

 (\cdot)



studies have investigated the trace element emission from coal fired boilers (257-258). In one (257) it was found that the highest concentrations of Pb and Cd appeared in the respirable size range and that the concentrations increased with decreasing particle size. This was found to be the case in this work for Pb. In the second study (258) it was found that the most toxic elements (As, Sb, Cd, Pb, Se and Tl) were concentrated in the smallest respirable particles. It was said that these are volatilised during combustion and preferentially adsorbed or condensed on to the small particles which can most easily pass through conventional control equipment.

3.7.4.2 Mass-Median Diameter (M.M.D.)

Another way of expressing cascade impactor results is by means of the mass-median diameter. This is the particle size above which 50% of the mass of that elementis found. The results are given in Table 3.41, together with the reported range found elsewhere (259). Only Al, Cd and Ni lie outwith the reported ranges and of these Ni is reasonably close. Both Al and Cd are well below the minimum reported ranges. Why this should be so is not clear. Ac this is the first study of its kind in Glasgow there are no criteria for comparisons and the low values may reflect different industrial activity.

5.7.4.3 Size Types

Using an Anderson Cascade Impactor with nine stages Rahn (251) divided the elements into five classes depending on the mass of the element associated with each size range. These are -

1)	L-type	-	mass	associated	with	largest	particles.
----	--------	---	------	------------	------	---------	------------

2) M-type - mass associated with medium sized particles.

3) S-type - mass associated with smallest sized particles.

F-type - mass divided equally among various size ranges.

5) Mixed patterns where two or more of the first three types are present to approximately the same extent.

The elements measured here were placed in one of these classes and the results are given in Table 3.42, together with those for Rahn for Liege in Belgium.

The partitioning of elements into different size ranges is of course arbitrary and borderline cases will exist. However, the results are in extremely good agreement with those of Rahn. Fe as can be seen from Table 3.39 has 57% on and above stage 3 and 28% on the back-up filter. Does this place Fe as an L-type or is 28% considered a large enough fraction to class Fe as an LS-type? Similarly Cd has 77% of its mass in the small size range but of this, 50% is on stage 4 and is extremely borderline. Cd can arguably be in either the S or M-type class.

The only contrary result is with Cu which in Glasgow is undoubtedly an L-type whereas in Liege it is an S-type. Since both mass-median diameter lie within the reported range then this again may simply reflect different industrial activity.

3.7.4.4 Enrichment Factors

The enrichment factors calculated to Fe for each stage are given in Table 3.43.

The following points can be made from these results.

i) All the enrichment factors found lie within the range reported in Section 3.6.

176

TABLE 3.42 ELEMENTS CLASSED AS A SIZE FUNCTION

Element	Type of Function	<u>Rahn (et al)</u>
Al	· L	L
Cu	Ľ	S
Cr	LS	–
Fe	LS	${f L}$
Zn	М	M
Cd	S	М
Ni	S	-
РЪ	MS	-
Mn	F	F

TABLE 3.4	<u>.3 E</u>	NRICHMENT	FACTORS	S OF	ELEMENTS	ON	EACH	STAGE	
	A	Cd	Cr	Cu	Fe	Mn	Ni	Pb	Zn
Stage 2		5 1358	9	74	1	l	22	1426	7 59
Stage 3	2	7 54	8	76	1	1	28	2 1 3 2	380
Stage 4	. 2	. 8390	7	71	1	l	372	7604	8980
Stage 5	l	2511	16	41	l	1	136	4904	363

* ATHOSPHERIC PARTICULATES - GLASCOW UNIVERSITY.

- ii) Al, Cr, Cu, Fe and Mn vary very little from stage to stage.The enrichment factors of 41 for Cu and 16 for Cr in stage 5 are probably not significant.
- iii) Cd, Ni and Pb are more enriched in the smaller particle range.
 - iv) Zn tends to be more enriched in the small-medium size range. These results are in partial agreement with those of Colovos <u>et al</u> (260) who found
 - i) enrichment factors for Pb and Cd were inversely proportional to particle size. This was found here for Pb but only for Cd in the very broadest sense.
 - ii) enrichment factors for Zn and Cu were independent of particle size. This was found for Cu but not for Zn which had a large fraction in stage 4.
- iii) Fe, Mn, Ca, and Mg in all size ranges had essentially identical ratios as average crustal material. This as can be seen from Table 3.43 was also found here.

The uniformity of Al, Mn, Fe and Cr, and the closeness to soil ratios provides further evidence of a common source for these elements.

Ni is close to the soil ratio down to 0.45 µm after which it increases dramatically. This is in agreement with this element having additional industrial sources. These high temperature processes favour the formation of small particles.

Cd, Pb and Zn are also more enriched in the small particle range suggesting industrial sources. The enhanced enrichment factor for Pb in the smaller particle range is also consistent with the emission size of the element from automobile exhausts.

3.7.5 Conclusions

The size distribution of elements in Glasgow's atmosphere in terms of the mass-median diameter and size functions are in good agreement with those reported from elsewhere with the exception of Al and Cd.

The most toxic elements (Cd, Ni, Pb) have extremely high proportions in the smallest respirable particle range which can easily enter into the alveoli. The other elements have 29 - 83% in the same size range, i.e. less than 1 µµm. The remaining fraction of these elements can easily pass the nasal passage and enter the upper sections of the lungs where adsorbed species can act more focally.

Enrichment factors for the elements suggest probable different sources and lie within the range of the whole sample analysis. (Section 3.6.3.2).

A note of caution must be made on conclusions drawn from the results as they are based on the analysis of one sample admittedly of 15 days duration and can only be considered to be of a tentative nature.

3.8 ASSESSMENT OF PROJECT AND POSSIBLE FUTURE WORK

This project has gone a long way to achieving the aims as set out in Chapter I.

The elemental levels of 21 elements have been measured over an extended period. From one site sampling the effect of weather has been studied and found to influence the levels and by studying various areas the city as a whole has been monitored. Possible sources of the elements have been suggested from their enrichment factors calculated to soil or rock ratios. The distribution of elements in the various size ranges has been investigated and the implications of the results discussed.

Future work would include as already mentioned further manipulation of the existing results to gain more precise information on the effect of weather conditions on elemental The range of elements could be extended to include such levels. elements as Be which has given cause for concern in recent years (1, 2, 261).Finally one study which was not carried out was the diurnal variation of the elements. This has been done in other cities finding peak values for certain elements during peak rush hours, e.g. (120). An attempt was made to do this but was abandoned because of insurmountable sampling problems. Because of the small pore size of the millipore filter paper large volumes of air could not be sampled in a short period. Conversely no pump was available which would sample enough air, drawn through for example a Whatman's filter paper to overcome the higher blank values of the latter filters.

WATER POLLUTION

4.1 INTRODUCTION

As stated previously the three main areas of human contamination by elements are through breathing air, eating food, and drinking water. The first of these has been extensively studied in Chapter III and the second was not considered here as it is monitored by the Department of the Regional Chemist which is now also involved in a survey of domestic drinking water for Pb, Fe and Cu. However, at the time this project was initiated this was not being done and to gain some information about this a survey of drinking water in the Glasgow area was initiated.

This survey is particularly important in the case of Glasgow because of the soft water supply and the use of lead piping. The problem is that soft water does not give pipes a natural coating of calcium carbonate and soft water itself tends to dissolve metals more readily than alkaline waters. Moreover many metals exhibit higher toxicities in soft water than in hard water. This is said to be partly caused by competition with the calcium present in hard water which can block the toxic action of metals by occupying enzymic sites.

There have been several studies carried out in the Glasgow area where attempts have been made to correlate mental retardation problems and the occurrence of cardiovascular disease with the presence of lead in drinking water (262-265). In a study of 71 houses in the Highlands a combination of soft water and lead lined

tanks was found to give cold tap water values of the order of 18 times the upper acceptable level of that element (266). Bones of people in soft water areas have been found to have higher lead content than those in hard water areas (267). There have been other studies of this type in Britain (268,269) and also in other countries (270-273).

However, several people feel that the relationship between soft water, lead content, mental retardation and cardiovascular disease have been over emphasised with the handling of the statistics a main weakness (274) and Elwood <u>et al</u> in particular has in at least three studies found:-

- i) No significant difference between the mean lead content of homes of mentally subnormal children and those of control normal children (275).
- ii) The returns for deaths for 1969-1973 in N.W. Wales failed to support an association between water lead values and mortality from cancer or cardiovascular disease (276).
- iii) That water lead was an unimportant source of blood lead levels in N.W. Wales an area of high lead content in the water (277).

Whatever the medical opinion, however, it is a wise precaution to know the metal levels in any soft water area for future reference.

4.2 METALS STUDIED

The obvious metals to study are those which are common in plumbing materials - Pb, Zn, Fe, Cu, Cr and Ni. Cd was also added to this list as it has been identified in drinking water in Sheffield and on that occasion was found to be above the W.H.O. maximum of 0.01 ppm (1).

4.3 SAMPLING

4.3.1 Contamination

Because of the ever present possibility of contaminating the sample or causing loss of analyte element, all stages of sample handling were executed with due care. The metals to be determined in these samples are present at trace levels and consequently even small amounts of contamination will drastically alter the concentration. Similarly, loss of even a few micrograms of element by absorption or precipitation will have a very serious effect. Consequently all plastic storage bottles were new and cleaned by standing overnight containing N/10 HCl and then thoroughly rinsed with de-ionised water. All glassware used was new and also cleaned with $^{\rm N}/10$ HCl and de-ionised water. Pipette tips for presenting the sample to the spectrophotometer which are an additional source of contamination especially Fe, were rinsed in $^{\rm N}$ /10 HCland de-ionised water immediately before use and only used once.

4.3.2 Collection

Bottles were distributed to members of the Chemistry department as this was found to give a reasonably good coverage of the whole city and surrounding area and amounted to some 120 samples. In 12 cases two samples were taken, one of the water standing in

the pipes overnight and the second after running the water for 5 minutes. In the remainder only the second type of sample was taken (running water).

Participants were asked to complete the questionnaire illustrated in Figure 4.1

4.3.3 Storage

Samples were acidified with glacial acetic acid (1 ml per 100 ml water) to stabilise the metals and prevent absorption on to container walls. Such samples are stable for up to one month and except for Pb can be stable for up to 4 months (278).

4.3.4 Preparation

Metals in water may be divided into four classes - dissolved, suspended, total and extractable, the terms being self-explanatory. Of these four only dissolved metals were studied here since this represents the "normal" background level, as suspended and total are subject to fluctuation with erosion of the piping. Extractable in terms of human ingestion is meaningless and is more important in river and loch systems.

Samples were therefore filtered through a 0.45 µm membrane filter and the filtrate then acidified with acid for storage. (The filter size although somewhat arbitrary is in common use).

4.4 ANALYSIS

Samples were analysed by atomic absorption spectroscopy using a Perkin Elmer 306 double beam instrument equipped with a heated graphite atomiser (H.G.A. - 74 model) and deuterium background corrector.

Directions:-

minutes before taking the sample to clear the pipes of water lying Run the cold water tap in the kitchen for approximately 5 in them, and fill the bottle provided. The following questionaire is to allow a comparison of all the factors to be made.

- (1) Name:-
- Address (street name and district only required) :-5
- (3) Bottle Number:-
- (4) Approximate age of house:-
- (5) *Material of plumbing (lead, copper etc.):-
- * Does the cold water come from a tank in the house or street mains? (9)
- (7) *If a tank material of tank:-
- Any other relevant information (water going brown etc. (8)

It has been found that with power plant effluents analysed by H.G.A. aqueous standards give good agreement with the standard additions method, hence samples could therefore be compared directly against the aqueous standards (278). Also as these water samples analysed here are characterised by low salinity and low dissolved solids content (even more so with soft water) the metals can therefore be determined directly against aqueous standards as calibration. Acetic acid was added to standards to give the same acid matrix.

A de-ionised water blank was processed in the same manner as the samples.

4.4.1 Accuracy

Since there is virtually no sample preparation the main sources of error in this analysis are:-

i) Measurement and pipetting of the sample into the furnace.

ii) Measurement of peak heights.

iii) Straight line fit of standards.

From duplicate analysis of samples using difference standard solutions the accuracy of these results has been estimated to be $\pm 10\%$.

4.4.2 <u>Reliability</u>

To determine the weight which can be placed on reporting the results of one set of analysis, samples were taken each morning for 7 days after running for five minutes from two chosen sites. Samples of overnight water were also taken on 3 days. The results are given in Table 4.1 and indicate that there is a fairly uniform level from day to day both in standing and running water.

4.5 RESULTS

The results of the survey are given in Table 4.2 together with the W.H.O. limits (279) and the comparison of standing and running water is given in Table 4.3. The Pb level is the maximum permitted by the World Health Organisation. Cu, Fe and Zn levels are the highest "desirable" levels which the W.H.O. would like. However, at present the maximum levels of these elements are 1500 μ g/1, 1000 μ g/1 and 15,000 μ g/1 respectively. The W.H.O. report recommends periodic checks for Cr levels but gives no limits and Ni is not mentioned at all.

Pb in Glasgow's domestic water is higher than the W.H.O. limit.

Cu is close to the maximum "desired" level but well below the present limit. Zn and Fe are well below the "desirable" levels as are presumably Ni and Cr since no limits are given.

The classification of Table 4.2 into plumbing materials was found to be the most convenient way of presenting the results as there were no relationships found between the age of house and the metal levels or area and levels. This was probably because in recent years due to government grants, both in private and corporation houses, the plumbing (together with electrical wiring) has been renewed in the older properties.

Statistically there is no difference between the metal levels found in the different classes, in fact the ranges tend to overlap.

TABLE 4.1	VA	RIATION	OF ME	TAL LE	VELS	IN D	RINKING	WATER DURING	-
								ONE WEEK	(µg /1)
<u>Site l</u>									
Copper Pi	pes	Pb	<u>Cu</u>	<u>Zn</u>	<u>Ni</u>	C	<u>r</u> <u>Fe</u>	Cd	
Monday Ru	nning	: 29	80	10	3	N	D 30	0.21	
Tuesday	11	32	78	10	3	N	D 22	0.19	
Wednesday	**	29	70	20	3	N	32	0.13	
Thursday	n	32	82	10	3	N	0 34	0.20	
Friday	n	28	82	10	2	N	28	0.32	
Saturday	**	29	80	20	2	N) 30	0.20	
Sunday	**	28	76	20	2	N	D 46	0.21	
Monday Sta	andin	.g 48	640	120	5	N.	D 20	0.18	
Wednesday	11	45	290	150	7	N) 22	0.20	
Saturday	**	48	600	150	7	N	20	0.19	
Site 2									
Lead Pipes	5	<u>Pb</u>	Cu	<u>Zn</u>	<u>Ni</u>	<u>C</u> :	<u>r</u> <u>Fe</u>	Cd	
Monday Run	ning	116	16	20	3	N	64	0.25	
Tuesday	n	96	14	20	3	N	60	0.21	
Wednesday	"	98	16	20	5	N) 50	0.16	
Thursday	11	92	14	10	8	N	94	0.22	
Friday	11	90	14	20	2	N	92	0.19	
Saturday	n	94	14	10	3	N) 92	0.22	
Sunday	н	109	16	10	3	N	36	0.22	
Monday Sta	andin	g 205	660	230	4	N) 188	0.23	
Wednesday	11	124	712	220	4	N) 164	0.20	
Saturday	"	143	144	20Ò	3	N) 136	0.18	

1.

ND = Not Detected

.

TABLE 4.2 AVEF	AGE LEVEL	S OF I	METALS	IN RU	<u>NNING</u>	TAP WATER	(rs/ 1)
Plumbing Materia	Pb 1	Cu	Ni	Cr	Zn	Fe	
Lead	264	51	10	ND	103	59	
Copper	137	68	44	ND	109	58	
Lead/Copper	402	50	25	ND	150	67	
W.H.O. Limit	100	50	-	-	5,000	100	

,

.

.

ND = Not Detected

.

TABLE 4.3	COMPARISON	OF	METAL	LEVELS	IN	STANDING	AND	RUNNING	. 1	
			a					WATER	(mg)1)
		•	IN S	LASCOW					• • •	

	<u>P</u>	<u>'b</u>	<u>C</u>	Cu		
Material	<u>Standing</u>	<u>Running</u>	<u>Standing</u>	Running		
	350	30	390	20		
	420	280	250	20		
Lead	1230	320	40	30		
	1300	600	50	50		
	500	50	20	20		
	30	30	490	130		
	20	30	110	120		
	240	110	190	40		
Copper	40	20	680	50		
	7 0	130	970	180		
	60	90	60	10		
	0	0	60	20		

A reason for this is quite probably that participants were unaware of the true nature of the plumbing. In many cases the plumbing consists of copper to the sink but lead under the floorboards leading from the street mains. Evidence for this can be seen where copper piping has a very high Pb content (Table 4.3).

However, there are some trends. <u>On average</u> it can be said that in the case of lead plumbing Pb levels tend to be higher than in copper plumbing. Conversely in copper plumbing Cu tends to be higher. Ni is higher in copper plumbing perhaps due to it being incorporated at manufacture to form an alloy rather than true copper. Zn and Fe are relatively constant and this is probably a general background level either by being intrinsically in the water itself or coming from street mains. Cd was only determined in 40 samples chosen to give a representative spread across the city and showed little variation throughout, having an average level of $0.24 \mu g/1$.

From Table 4.3 it can be seen that in some cases both Cu and Pb are high and as said above is evidence for the point made about the plumbing. In fact in the cases with copper piping the Pb level rose in the running water and could be due to water lying in the lead pipe section overnight. A further sample taken after running for a longer period would probably show reduced levels. In general, however, there is a marked decrease in the metal levels of running compared with standing water - as would be expected.

191

4.6 LEACHING FROM PIPES

The previous exercise has shown at least in terms of average values that the plumbing material will be leached into the drinking water. To study leaching in a more controlled environment the following experiment was carried out.

Five pipes, made of material commonly used in plumbing, were cut into 1 foot lengths. These were - lead, copper, stainless steel, plastic and lead/copper made from 6 inch lengths of each material and soldered at the joint with a tin based solder in common use.

For practical reasons pipes could not be obtained of exactly the same cross-section. In an attempt to compensate for this a volume of de-ionised water equal to half the internal volume of the pipe was added to each pipe. The pipes were placed on an end over end shaker overnight (16 hours). In the morning the water was filtered through a 0.45 μ m membrane filter. Glacial acetic acid was added to correspond to 1 ml per 100 ml water and the samples stored in clean plastic bottles.

This was repeated on three consecutive nights with the same pipes.

The samples were analysed by atomic absorption spectroscopy in the same manner as the domestic water.

The average results are given in Table 4.4 and the findings are more or less self evident. The lead pipe is high in Pb, the copper pipe in Cu and the lead/copper pipe in both. The latter two pipes are also higher than the lead pipe in Ni but not enough to confirm what was said previously about Ni perhaps being present in copper pipes as an alloy. All three are high in Zn and about the same in Fe.

TABLE 4.4 META	T TEAETS	IN LEAC	HING WATER	<u>a</u> (pa)	1)	
FRC	M NARUCI	US PLUI	HBING MA	TERIALS	<u>.</u>	
Pipe	Po	<u>Cu</u>	Zn	<u>Cr</u>	<u>Ni</u>	Fe
Lead	33,000	20	580	1	3	54
Lead/Copper	22,700	1640	220	2	7	54
Copper	100	2280	580	1	6	34
Stainless Steel	61	35	1300	140	120	116
Plastic (PNQ)	113	10	1500	4	10	104

1.5

Stainless steel has higher Zn and Fe values than the other three pipes examined and also is much higher in Cr and Ni. The latter two of course are the main constituents of stainless steel. Stainless steel pipes are usually given an internal zinc coating and this is reflected in the high level of that element in the water. Fe again is probably a contaminant from the manufacture of the pipes.

The interesting figures for the plastic pipes are the relatively high levels of Pb, Zn and Fe. Lead acetate is used as a filler in the manufacture of plastic pipes, zinc stearate as a stabilizer and Fe is again probably present as a contaminant at the time of manufacture.

4.7. CONCLUSIONS

The amount of element leached from domestic water pipes can be set into classes according to the plumbing material, with limited success.

Of the 7 elements studied in the survey only Pb gives cause for concern as judged by W.H.O. standards. In later years Cu could become more of a problem as the limit is reduced to the "desirable" level.

Water lying in pipes overnight has higher metal levels and should be run off before the water is used.

Plumbing material is leached out into the domestic water, but simply replacing lead piping with copper, stainless steel or plastic will only change the nature of the problem but will not necessarily rectify it.

4.8 ASSESSMENT OF PROJECT AND POSSIBLE FUTURE WORK

In terms of the aims as set out in Chapter I the project has been successful - as far as it goes. The levels of 7 trace elements in the domestic water supply have been measured and compared to W.H.O. standards, thereby emphasising the ones which do or might cause concern.

There would be little point in extending the range of elements studied, as the project has covered the main elements found in plumbing materials. More information would be obtained by extending the survey to include more houses covering a wider area. CHAPTER V

INDUSTRIAL WASTE

5.1 INTRODUCTION

The historical reasons for the comparatively large number of industrial wastetips in Central Scotland has been well documented elsewhere (280). The main type is of course colliery waste. In recent years these coal bings, as they are called, have attracted the attention of environmentalists presenting many interesting problems from the point of view of growing plants. This aspect has been studied extensively in recent years (280).

One of the biggest problems is associated with the oxidation of the mineral pyrite which produces large amounts of acid. This causes large amounts of Fe, Al and Mn to be brought into solution at levels toxic to plants when the pH falls below 4.5-5.0. Al is said to come from the breakdown of illite clay at low pH. Other trace metals present may also be brought into solution at toxic levels as the pH falls and in the case of extensive leaching could result in the removal of these metals from the root zone causing deficiencies and they could possibly ultimately pass out of the bing altogether.

When the situation is reached that the bing effluent is sufficiently acid, large amounts of soluble Fe will be leached away usually into neighbouring streams where the ensuing higher pH causes precipitation, thus coating the stream bed with iron oxides. This of course will not be limited to Fe, and other metals present will be leached in a similar manner. In the absence of obvious run-off there is also the possibility of seepage of metals into surrounding soil which could lead in severe cases to levels either high enough to stunt crop growth or cause problems to grazing animals.

It was therefore decided in this study to concentrate on two chemical aspects:-

i) Assessment of the metal content of the bings in terms of both total levels and also that which is immediately available to plants.

ii) Assessment of the effect of the bings on the immediate countryside in terms of seepage, blow-off and run-off.

5.2 TRACE METAL LEVELS IN SCOTTISH COAL MINE WASTE

5.2.1 Sample Collection and Preparation

Samples were collected from five sites in Central Scotland the map references of which are given in Table 5.1. They were air dried and sieved through a 2mm sieve. The 2 mm fraction was retained and analysed as this is standard soil procedure.

5.2.1.1 Acetic Acid Extractable

0.5M acetic acid was used as an extracting medium as this again is accepted soil practice when assessing the availability of nutrients to plants and therefore will allow comparison to be made with soil data.

lOg of sample were shaken on a reciprocating shaker for 18 hours with 100 ml, 0.5M acetic acid and the suspension filtered and stored in pre-acid cleaned plastic bottles.

5.2.1.2 Total

A method similar to that of Hartstein <u>et al</u> (281) was used to dissolve the sample. A representative 100 mg of sample was placed in a pre-acid washed teflon dissolution bomb with a screw top. 5 ml "Aristar" H_2SO_4 and 4 ml "Aristar" HNO₃ were added and heated at $200^{\circ}C$ for 2.5 hours on a sand bath. The bomb was thoroughly cooled and 5 ml "Aristar" HF added to the clear solution and heating continued for a further 1 hour. The bomb was again thoroughly cooled and the contents transferred to a 100 ml standard flask containing 3g boric acid and made up to the mark with deionised water.

The H_2SO_4/HNO_3 mixture was used for the reasons given in Section 3.6, i.e. split peaks.

199

۰.

i

TABLE	5.	1	
	· ·	-	

1 MAP REFERENCES OF SAMPLE SITES (CCHLERY WASTE)

(Company 1110)			
Site	<u>Re</u> :	ferer	lce
Lumloch	ns	640	702
North Lodge	NS	753	553
Stane	NS	884	587
Stoneyburn	NS	978	634
Dykehead	NS	86 7	608
Meadowhall	NS	770	518
Larkhall	ns	7 62	525
Meikle Earnock	ns	712	531

.

 $\widehat{}$

.

The boric acid is present to suppress Si interference during analysis as it will be present rather than removed as SiF_6 as would be the case if, e.g. a sodium fusion dissolution technique was adopted.

The best way of getting the boric acid into solution was found to be to add the boric acid to about 50 ml de-ionised water in the standard ilask and with frequent shaking most of the boric acid was dissolved before the sample was added.

A blank solution containing all reagents was made and analysed together with the samples.

5.2.2 Analyses

The samples were analysed by atomic absorption spectroscopy, using a Perkin Elmer 306 double beam instrument with deuterium arc background corrector and H.G.A. 74 flameless atomiser.

5.2.2.1 Total

Fe, Ni, Mn, Cu and Zn were determined using an air/acetylene flame, Al using a nitrous-oxide/acetylene flame and Cr and Pb using a heated graphite atomiser.

100 mg of sample per 100 ml solution is a relatively weak solution 1A terms of metal content and the main interferences will arise from the matrix effects of the acids and salinity from salts of Na, K, Ca and Mg. These 4 elements were determined using the appropriate flame and ionisation suppressor, and were found to have average values of 100 ppm sample solution for Na and Mg, 20 ppm for K; and 30 ppm for Ca. These concentrations were added to the standards together with the acid mixture with the background corrector compensating for the discrepancies in salt concentration from sample to sample.

200

To test the reliability of this method of dissolution and matrix matching a crushed rock standard (282) was dissolved and analysed by this procedure. The results of duplicate analysis together with those of the average values of inter-laboratory studies are given in Table 5.2. The Al and Fe are expressed as a percentage and the remainder as ppm. The Pb concentration in the standard rock was only 7.8 ppm which after dissolution will not be detected. No Pb was detected in the samples at least indicating that levels greater than the true value in the samples will not be obtained.

To test the accuracy of the method one sample was analysed 5 times using five separate 100 mg batches. From these results the standard error for the analysis was calculated to be:-

Ni,	Cu,	Al,	Fe		:	2%
Cr					:	3%
Mn,	Zn				:	6%
Pb				,	:	12%

This error is not purely analytical but also includes a sampling error and as such is a very good indicator that a uniform well mixed sample is being analysed.

5.2.2.2 Acetic Acid Extractable

The "available" levels were analysed in the Department by I.D. Pufford who from a similar exercise has calculated the standard errors to be:-

Cr,	Al		:	2%
Mn,	Zn,	Pb	:	5%
Ni			:	7%
Cu,	Fe		:	10%
<u>TABI</u>	E 5.2 RESULTS FOR STAND	ARD ROCK (U.S.G.	(S, w, I)	
-------------	-------------------------	------------------	-----------------	
	Interlaboratory Value	Sample 1	<u>Sample 2</u>	
Al	7.9%	7.0%	7.2%	
Fe	7.8%	8.3%	8.1%	
Cu	110 ppm	107 ppm	109 ppm	
Zn	86 ppm	93 ppm	90 ppm	
Ni	76 ppm	7 8 ppm	80 ppm	
Mn	1278 ppm	1241 ppm	1276 ppm	
Cr	ll4 ppm	108 ppm	112 ppm	
Pb	7.8 ppm	ND	ND	

.

ND = Not Detected

.

.

5.2.3 <u>Results</u>

As said earlier the biggest problem in coal bings is the oxidation of pyrite. Duncan <u>et al</u> (283) have found that this takes the form of a cycle in terms of pH and sulphate content in unburnt material. Based on this cycle, bing material may conveniently be sub-divided into pre-acid, acid and post-acid. Burnt material forms a fourth class.

The samples were divided into these four classes and the range and average values for the 8 metals are given in Table 5.3. The results are not statistically different as can be seen from the range of values. This is in retrospect probably due to variation of material with different sites and a detailed study of one site would have produced better statistical differences. However general trends can be projected from the average values although at this stage these must be treated with caution.

5.3.2.1 Total

The total levels found in bing material, with the exception of Cu which is higher, lie within the normal soil ranges (284). There is a general increase in the metal concentrations as the oxidation of pyrite progresses except for Pb which decreases. This increase in post-acid over pre-acid is probably due to reprecipitation in the fine fraction, of the elements which were released from the coarse fraction as the acid was leached from the bing leaving this fine fraction more enriched in trace metals.

Combustion within the bing also causes an increase in the total metal content of the material, again with the exception of

TABLE 5.3	METAL LEVELS	OF	SCOTTISH	COLLIERY	WASTE	(ppm)

1				[
		<u>Fe</u>		Al		Mn		Cu	
		<u>Total %</u>	<u>Ext</u>	Total %	<u>Ext</u>	<u>Total</u>	Ext	<u>Total</u>	Ext
Pre-Acid	D	1.30 -	ND-	6.1 -	20–	294 -	14 - 89	39-	ND-
Unburnt material	Kange	5.31	108	12.2	700	1822		205	10
· `	Mean	3.35	56	8.4	188	716	40	107	3.1
Acid	Range	4.14-	140-	7.4-	15-	10-	9-	29–	2.1-
Unburnt material		8.38	825	17.2	100	3330	284	319	9.0
	Mean	6.20	307	11.2	46	1 37 6	69	101	4.0
Post-		1.77-	ND	6.2-	ND	1122-	23-	40-	ND
Acid	каnge	7.29	40	19.8	155	8109	190	491	5.8
Unburnt material	Mean	4.38	15	12.1	67	3047	84	199	2.3
		2.89-		7.9-	140-	99-	1.2-	20-	2.0-
Burnt material	Range	12.59	ND-233	13.8	1070	6263	165	535	19.0
	Mean	7.08	51	11.2	543	1937	39.5	337	9.1

.

ND = Not Detected

TABLE 5.3 (CONTD.)

		<u>Z</u> :	<u>n</u>	<u>Ni</u>		<u>Pb</u>		<u>Cr</u>	
		Total	Ext	Total	Ext	<u>Total</u>	<u>Ext</u>	Total	<u>Ext</u>
Pre-Acid	Banco	58 -	6.2 -	98 -	ND-	1.1-	0.2-	151 -	0.13-
Unburnt	nange	478	17	201	3.6	108.5	1.0	3 <u>0</u> 3	0.28
material	Mean	247	10.4	137	2.2	37.2	0.5	224	0.21
Acid	Range	38-	3.8-	89–	0.8	1.3	0.1-	150-	0.05
Unburnt		1226	13	369	-8.0	-128	0.5	531	-0.54
material	Mean	276	7.0	182	2.6	31.5	0.2	2 7 7	0.19
· Post-	Domes	121–	6.9-	121-	1.7-	2.5	0.5-	169 -	0.01-
Acid	Kange	2658	177	47 0	10.5	-47	4.7	440	0.32
Unburnt material	Mean	916	45.9	223	5.7	12.4	1.6	268	0.12
Denud	 D	197-	60–	138-	ND-	1.2 -	0.1-	2 <u>3</u> 3-	0.11-
burnt	кange	1794	160	515	12.9	14.0	0.8	559	0.95
material	Mean	680	36	291	3.1	6.4	0.4	342	0.41

1.2

ND = Not Detected

•

.

Pb. This concentrating effect is due to the loss of organic material which can constitute up to 80% of the unburnt material. The temperature within the bing must be greater than 450°C to burn off the organic material and convert the black to red material. At this temperature Pb could be lost due to volatilization and this could explain the decrease in Pb concentration in the red material.

5.2.3.2 Acetic Acid Extractable

These levels again lie within the usual soil range with the exception of Cu which is higher.

The trends for acetic extractable levels are not as definitive as those for the totals. Fe increases dramatically from pre-acid to acid showing the influence of the oxidation of pyrite.

Al and Cr show a decrease in the post-acid material compared with pre-acid. Mn, Zn, Ni and Pb are all present in higher concentrations in the post-acid stage and Cu is surprisingly constant throughout.

As with the total levels, burning influences considerably the extractable metal levels. The level of extractable Al is substantially increased, due to the lattice breakdown of illite mentioned earlier. Fe and Mn levels are of the same order as those found in pre-acid material, probably due to the formation of higher oxides. Pb levels are comparable with pre-acid. Cu, Zn, Ni and Cr levels are all increased. It would appear therefore that certain trace elements have been made more extractable by combustion and this has obvious implications on the suitability of burnt material as a plant growth medium.

5.3 EFFECT ON THE SURROUNDING COUNTRYSIDE

The bing situated at Stoneyburn (NS 978634) was chosen to study the effect of coal waste tips on the environment for the following reasons -

- i) It is situated in the countryside surrounded by agricultural land and well removed from the nearest main road.
- ii) The surrounding land is generally level, therefore there will be no obvious bias towards seepage in a particular direction.
- iii) The surrounding land was all under grass giving similar conditions in all directions at the time of sampling.
 - iv) Two streams run past the bing one on either side allowing a study to be made of the seepage from the bing.
 - v) One stream ran along the north side. Therefore the prevailing south-west winds would be expected to carry material over the stream thus eliminating any possible confusion with seepage material.
 - vi) The bing had been shown to contain both pre-acid and acid material as well as red material.

A map of the bing and surrounding area is given in Figure 5.1. This represents the completed project which was not the case at the time of sampling. The ditch along the west side and the new ditch along the south side were not present.

5.3.1 Effect on Streams

There was evidence of actual bing material being carried into the stream on the north side caused simply by downward movement presumably by water. This stream was therefore not studied. The stream on the south side was more uniform with no obvious signs of plant life and there was evidence of iron oxide precipitation on

FIGURE 5.1 CHART OF STONEYBURN BING AND SURROUNDING LAND



FIGURE 5.1 (CONTD.)



1.

the bed but no obvious signs of actual bing material could be recognised.

Samples of sediment and water were taken at 100 yard intervals along the stream as indicated in Figure 5.1. The stream is flowing from sampling point 1 to sampling point 5. The water was analysed by atomic absorption spectroscopy using the method of standard additions after being filtered through a 0.45 por membrane filter. The sediment was analysed by the method used previously for the determinations of total metals in the bing material.

•

: : :

The results are given in Table 5.4. Of the stream sediments Al, Zn, Pb and Cr are lower before and after contact with the bing. Fe and Cu are lower before the bing but remain high after contact indicating that these elements are carried further before precipitation occurs. The trends for Mn and Ni are confused by high values before the bing, but higher values were noted for both, adjacent to the bing.

The water samples all show an increase in metal levels as the water passes over the sediment. The most dramatic increases were noted for Al, Mn, Zn and Ni. Fe does not exhibit this big increase presumably due to the insoluble nature of the iron oxide precipitate.

5.3.2 Effect on Land

Soil samples were taken in January to minimise possible conflictions due to plant uptake, covering the right compass points at distances of 0, 5, 10, 15, 20, 50, 75 and 100 yard intervals out from the bing. The samples were taken to a depth of 12 inches and well mixed.

	Fe		<u>A1</u>		Mn		<u>Cu</u>	
Sample No.	Sediment	<u>Water</u> ppm	Sediment %	<u>Water</u> ppm	Sediment ppm	<u>Water</u> ppm	Sediment ppm	<u>Water</u> ppm
l	2.2	0.018	3.7	0.085	985	1.7	4.9	0.02
2	4.1	0.026	8.0	0.70	731	6.5	20.9	0.05
3	3 3•3	0.026	4.2	1.89,	1870	10.9	17.8	0.05
4	24.8	0.041	3.8	3.60	614	16.6	40.9	0.07
5	32.2	0.025	1.9	1.32	685	15.3	19.6	0.07

	Z	<u>n</u> .	<u>. N</u>	Ni		<u>b</u>	Cr	
Sample No.	Sediment ppm	<u>Water</u> ppm	Sediment ppm	<u>Water</u> ppm	Sediment ppm	<u>Water</u> ppm	<u>Sediment</u> ppm	<u>Water</u> ppm
l	125	0.06	118	0.05	118	0.018	24.7	ND
2	595	0.28	219	0.12	4590	0.026	65.1	0.002
3	597	0.42	45	0.25	219	0.026	23.4	0.006
4	194	0.99	82	0.30	184	0.041	35.9	0.001
5	127	0.84	29	0.30	160	0.025	19.8	0.001

)

ND = Not Detected

.

.

۰ . The samples were sieved through a 2 mm sieve and air dried within 2 days of sampling. After statistical sampling the total and acetic extractable levels were determined in the same manner as for the bing material.

ļ

A problem of split peaks was encountered with the acetic extractable Pb determinations on the heated graphite atomiser. This was possibly due to the presence of mixed anions (see Section 3.6.2.3). Adding excess phosphate resulted in the formation of a single peak. However when the phosphate solution was added directly to the carbon rod reproducible results were not obtained. The best results were obtained by adding 1 ml of sample solution to a 1 ml solution containing 1000 ppm P as phosphoric acid in a test tube. After shaking and leaving for 5 minutes, aliquots were then injected on to the rod. Cr was also analysed by the heated graphite atomiser but the problem of split peaks did not arise.

5.3.2.1 Results

A brief description of the land in each direction is included below to help in the interpretation of the results. The results are illustrated graphically in Figure 5.2. With only a few exceptions the levels lie within the normal soil range. The exception is acetic extractable Cu which like the bing material itself is higher than normal.

5.3.2.1.1 North (N):- The land in this direction runs downhill fairly steeply from 0-75 yards to the stream with the 100 yard sample taken just across the stream.



























FIGURE 5.2 (CONTD.)



N



FIGURE 5.2 (CONTD.)





FIGURE 5.2 (CONTD.)



,















FIGURE 5.2












236

FIGURE 5.2

FIGURE 5.2 (CONTD.)



237

个

N









FIGURE 5.2 (CONTD.)



N





Total Levels

Al and Pb have V-shapes from 0-15 or 20 yards after which the Al falls to 75 yards while Pb fluctuates randomly. Cu and Ni have U-shapes over the ranges 0-15 and 0-50 yards respectively. Ni then shows a big drop to 100 yards while the fall of Cu is not so dramatic. Cr and Zn generally fall to 75 or 100 yards while Fe rises to 75 yards. Mn tends to fluctuate randomly.

Acetic and Extractable

The general trend is a decrease with increasing distance from the bing. Al, Cr, Zn fall to 75 yards with the decrease of Zn being steep initially. Ni falls only to 50 yards. Cu falls to 10 yards and levels out. Fe only falls to 5 yards before levelling out. Pb has a U-shape from 0-20 yards before falling to 75 yards and Mn has a broad \mathbf{n} -shape covering the whole 0-100 yards range.

There is no evidence from the results for the stream preventing the movement of metals across to land on the other side although in general there is a drop on crossing the stream. Only in the case of Mn (total and acetic extractable) Pb (total) and Cu (total) is there a dramatic drop on crossing the stream.

2.5.1.2 North East (NE) The land in this direction is separated from the bing by a stream. Zero point was taken across the stream. The land slopes upwards to the 75 yards sampling point with the 100 yards point across a tarmac road. This is the direction which will be subject to the prevailing winds.

Total Levels

Al, Fe, Mn, Ni, Zn fluctuate randomly along the sampling line while Cr remains fairly constant. Pb remains approximately constant to 50 yards before rising to 100 yards. Cu has a U-shape from 0-20 yards, thereafter falling to 100 yards.

Acetic Extractable Levels

Zn, Pb, Cu and Al all remain very much the same throughout. Fe is more variable but by and large remains constant. Cr falls to 10 yards and rises to 100 yards. Ni falls to 5 yards, rises to 20 yards and again falls to 100 yards. Mn fluctuates randomly.

2.5.2.1.3 East (E):- A rough track cut out of the bing separates the bing and land in this direction; the land remaining fairly level.

Total Levels

Al, Cr, Mn, and Zn show a rise to 15 yards (20 yards for Cr) and a fall to 100 yards: Mn and Zn being more dramatic. Ni, Fe, Pb and Cu show no trends but fluctuate randomly, Fe less so than others.

Acetic Extractable Levels

All elements except Ni and Fe are in general the same throughout. Ni increases to 2C yards, falls to 50 yards before increasing again to 100 yards: Al also starts to rise after 50 yards. Fe fluctuates randomly.

2.5.2.1.4 South East (SE):- As can be seen from the map this direction has been modified to carry drainage water which will tend to obscure any trends.

Total Levels

Cr is generally the same throughout with slight fluctuations. Pb is the same to 15 yards before increasing to 100 yards. Cu also increases over the range 15-100 yards but is random up to 15 yards. Zn and Fe have U-shaped trends from 0-20 yards after which Fe falls sharply to 100 yards, while Zn peaks at 75 yards. Ni and Al levels fluctuate randomly. Changes in Mn levels are most dramatic with a fall to 15 yards, a rise to 50 yards and a further fall to 100 yards. Acetic Extractable Levels

Zn is generally the same throughout as are Mn and Pb up to 75 yards and 50 yards respectively before rising to 100 yards. Ni is generally the same except for a peak at 15-20 yards and a trough at 75 yards. Cr peaks at 15 yards with a gradual increase to 100 yards. Cu falls gradually to 20 yards, rising slightly to 75 yards. Al levels are random, peaking at 15 and 75 yards. Fe shows the most dramatic changes with a V-shape covering 0-20 yards following by a dramatic fall to 100 yards.

2.5.2.1.5 South (S):- The land in this direction lies across the stream studied earlier for metal burdens with the zero point taken across the stream from the bing.

Total Levels

Al, Fe, Mi, show a general fall to 20 yards (50 yards for Fe) rising to 75 yards and falling again to 100 yards. Zn, Pb and Cu show a U-shape over 0-20 yards after which Zn fluctuates, Pb remains the same and Cu has a drop to 50 yards before generally levelling out. Cr demonstrates the reverse over the 0-75 yards range, ie. \bigcap -shaped and thereafter remains the same. Mn fluctuates randomly.

Acetic Extractable Levels

Cu and Pb are generally the same throughout. Cr and Zn are generally the same up to 50 yards, but rise to 75 or 100 yards. Mn and Fe show a general rise to 50 or 75 yards and fall again to 75 or 100 yards. Ni shows a general fall to 50 yards, rising to 100 yards with a peak at 75 yards. Al fluctuates randomly.

2.5.2.1.6 South West (SW):- The land in this direction is level well beyond the 100 yards point. Total Levels

Fe and Cr are generally the same throughout. Al, Pb and Cu generally fall to 100 yards. The reduction for Cu is dramatic to 15 yards and more gradual to 100 yards. Ni has a \bigwedge -shape covering the range 0-15 yards remaining the same to 100 yards. Zn is generally

the same to 15 yards, while Mn falls to 20 yards after which they both fluctuate, rising to 50 yards and them falling to 100 yards.

Acetic Extractable Levels

Al shows a general rise to 100 yards, while Fe shows a dramatic fall to 100 yards. Pb falls to 10 yards with a more gradual fall to 100 yards. Zn, Cu, Cr and Ni have U-shapes over a 0-20 yards range before falling to 100 yards. Mn has a general rise to 20 yards falling to 50 yards and rising again to 100 yards.

2.5.2.1.7 West (W);) Like SW the land is this direction is level well beyond the 100 yards point.

Total Levels

Fe shows a slight fall from 15-100 yards but generally remains the same as does Zn. Pb shows a dramatic fall to 20

yards and remains at that level to 100 yards. Al shows a V-shape from 0-20 yards followed by a steady fall to 100 yards. Cu is generally the same to 15 yards followed by a dramatic rise to 20 yards where it remains constant to 100 yards. Cr shows a gradual fall to 15 yards and after a slight rise to 20 yards remains constant to 100 yards. Ni and Mn fluctuate to 15-20 yards after which the Ni level falls to 75 yards while the Mn level remains the same.

Acetic Extractable Levels

Al shows a gradual fall to 100 yards. Cu, Zn, and Fe show an initial fall to 5 yards, and generally remain at that level to 100 yards. Ni shows a gradual fall to 20 yards, while Mn fluctuates, after which both generally remain the same to 100 yards. Pb fluctuates falling to 15 yards then gradually rising to 75 yards with a further fall to 100 yards. Cr shows a V-shape from 0-20 yards with a much broader U-shape from 20-100 yards.

2.5.2.1.8 North West (NW):- The ground in this direction is level to 50 yards with the 75 to 100 yards points slightly higher in the neighbouring field.

Total Levels

Ni shows a general rise to 100 yards with a deep trough at 75 yards. Fe is generally the same throughout but has a gradual fall to 100 yards. Mn and Zn have a general \bigcap -shape. That for Mn covers the whole range while Zn exhibits two humps. One of the humps is over 0-20 yards followed by a broader one over 20-100 yards with an overall drop with distance from the bing to 100 yards. Cu fluctuates to 20 yards followed by a steady steep fall to 100 yards. Pb is generally the same to 15 yards falling to 20

yards after which it remains at this level to 50 yards before rising slightly to 100 yards. Cr after a slight fall is generally the same from 5-50 yards rising to 100 yards. Al fluctuates to 20 yards rising to 75 yards and falling again to 100 yards.

Acetic Extractable Levels

The general trend is for the levels to decrease with increasing distance from the bing. In and Pb both fall over the range O-100 yards except for a slight peak of In at 75 yards. Cu, Ni, Fe and Al all decrease from O-15 or 20 yards after which Al remains constant while Fe has a slight rise to 100 yards. Ni has a broad \hat{n} -shape over the range 20-100 yards and Cu fluctuates, rising to 75 yards and falling again to 100 yards. Cr has a gradual fall to 50 yards and remains constant, rising slightly towards the 100 yard point. Mn fluctuates to 20 yards before falling to 100 yards.

From the foregoing descriptions of fluctuations in metal levels it is clear that no definite conclusions can be drawn and the absence of other similar work does not help. However certain overall trends do suggest themselves in terms of direction.

i) Directions Separated from the Bing

In two directions, NE which is separated from the bing by a stream and E which is separated from the bing by a dirt track road seepage will be restricted. In these directions the acetic extractable levels tend to remain more or less the same from 0-100 yards, while total levels are more variable. This might suggest blow-off of material particles rather than a seepage of solution.

-9*

ii) Directions in Contact with the Bing

These directions include N, NW, W and SN. SE is not included since as stated previously this direction shows obvious signs of being influenced by bing seepage. In the former four directions the overall trend, with some exceptions, is for acetic extractable levels to decrease with increasing distance from the bing. In N and SW directions this is also the case for the total levels but the fall is not so dramatic also in the W and NW there are no overall trends for total levels. The findings in these directions in contact with the bing might suggest that seepage is occurring since the variations are now being found in the acetic extractable rather than in the total levels.

The fluctuations in acetic extractable levels are not due to local fluctuations of pH value as can be seen from Table 5.5 which shows the pH values as calculated from a soil : water ratio of 1/2.5. As expected the SE direction has the lowest pH values due to bing drainage but in any one direction there is little variation in pH.

TABLE 5.5	pН	VALUES	OF	SOIL	AROUND	STONEYBURN	BING

<u>Distance (yards)</u>	<u>0</u>	5	<u>10</u>	<u>15</u>	<u>20</u>	<u>50</u>	<u>75</u>	100
North	4.8	5.0	5.4	5.2	5.5	5.8	5.5	6.2
North East	5.6	5.6	6.0	5.9	5.1	5.5	5.3	5.7
East	4.4	4.2	4.2	4.2	4.2	4.1	4.4	4.4
South East	3.1	3.1	3.3	3.9	3.0	3.3	4.5	4.5
South	4.4	4.8	4.7	5.9	5.5	5.2	4.9	4.8
South West	4.8	5.1	5.0	5.2	5.2	5.0	5.1	4.9
West	5.2	6.1	6.2	6.2	6.1	5.9	6.2	5.6
North West	5.4	6.0	6.0	6.1	6.3	6.3	5.8	6.3

5.4 CONCLUSIONS

As has been said repeatedly in this Chapter the statistics are not so good as they could be. This applies particularly to the splitting of the bing material into the four classes given on the basis of metal content. However even accepting that difficulties were experiencedin implementing the classification of bing material into pre-acid, acid, post-acid and burnt material, the metal levels themselves were accurate and represent the total and acetic acid extractable metal levels in bing material in Central Scotland. From the average values however trends suggest themselves and should be studied further. There is an increase in total levels in the 2 mm fraction as the pyrite oxidation cycle proceeds. Acetic acid extractable levels are more variable. Burnt material also shows higher values of total and acetic acid extractable levels due to loss of organic matter and mineral breakdown. Pb is the exception and decreases probably due to volatilisation.

The leaching of metals from the bing also presents a problem. This can be seen in terms of increased levels of metals in stream sediment and in the water passing over that sediment carrying the metals further down the aquatic system. Trends of metal levels in surrounding soil suggests that the bing is affecting the land in terms of seepage of acetic acid soluble forms and blow-off of total forms.

25I

5.5 ASSESSMENT OF PROJECT AND POSSIBLE FUTURE WORK

This project has been partially successful. The levels of eight metals in coal-bing waste have been recorded and the analysis of soil and stream sediments and water have provided evidence for seepage and blow-off of metals from the bing.

Future work would include the analysis of more definitive samples carefully selected from sites to include pre-acid, acid and post-acid material to determine whether or not the theories put forward concerning the placing of the metals into these classes are statistically correct or not. A study of further sites is also required to determine the full effect of bings on the neighbouring land in any attempt to confirm or refute the tentative findings arrived at for this one site.

REFERENCES

- 1. Tucker, A., 1972. <u>The Toxic Metals</u> Pan/Ballantine, London.
- 2. McCormac, B.M., 1971 <u>Introduction to the Scientific Study of</u> <u>Atmospheric Pollution</u>. Reidel, Dordrecht, Holland.
- Walker, C., 1971. <u>Environmental Pollution by Chemicals</u>. Hutchinson Educational, London.
- British N.E.R.C., 1971. The Sea Bird Wreck in the Irish Sea Autumn 1969. Publication Series C, NO. 4.
- 5. Kneip, T.J.; Lauer, G.J., 1973. <u>Trace Metal Concentration</u> <u>Factors in Aquatic Ecosystems</u> in <u>Progress in Analytical Chemistry</u> <u>Volume 5 Chemical Analysis of the Environment</u> Ahuja, S.; Cohen, E.H.; Kneip, T.J.; Lambert, J.L.; Zweig, G. (Editors) Plenum Press, New York.
- Klein, D.H., 1972. Mercury and other metals in urban soils Environmental Science and Technology, <u>7(4)</u>, p 560.
- Klein, D.H., 1973. Heavy metal fallout around a power plant.
 Environmental Science and Technology, <u>7(4)</u>, p 357.
- Buchauer, M.J., 1973. Contamination of soil and vegetation near a zinc smelter by zinc, cadmium, copper and lead. Environmental Science and Technology, <u>7(2)</u>, p 131.

- Purves D.; Mackenzie, E.J., 1969. Trace element contamination of parklands in urban areas. Journal of Soil Science, <u>20</u>, p 288.
- 10. Little, P.; Martin, M.H., 1972. A survey of zinc, lead and cadmium in soil and natural vegetation around a smelting complex. Environmental Pollution, <u>3(3)</u>, p 241.
- 11. Asami, T., 1972. Pollution of paddy soils by cadmium, zinc, lead and copper in the dust and fumes and waste water from the Nisso Aizu smelter. Nippon Dojo - Hiryogaku, <u>43(9)</u>, p 339 (cited CA, 1973, <u>78</u>, 150786).
- Suzuki, Yukio; Suzuki, Yasuo; Fujii Nobuo; Mouri, Takoji;
 1974. Environmental contamination by arsenic around a smelter. Shikoku Igaku Zasshi, <u>30(3)</u>, p 213 (cited CA, 1974, <u>18</u>, 175294).
- 13. Itano, Yoshitoro; Kawata, Masaharu; Yoda, Kinya, 1974. Condensation and emission of heavy metals during heating and dust collection at the Fujivora factory of the Onoda Cement Co. Kagaku (Tokyo), <u>44(8)</u>, p 487 (cited CA 1974, <u>81</u>, 158060).
- 14. Vetter, H., 1974. Strain and damage by heavy metals in the vicinity of lead and zinc smelting plants in lower Spascony.
 Staub-Reinshaft Luft., <u>34(1)</u> p 10 (cited CA 1974, <u>80</u>, 136908).

- 15. Roesler, H.J.; Beuge, P.; Mueller, E., 1969. Effect of foundry smoke at Freiberg and Halsbucke on the trace element content in soil. Bergakademie, <u>21(7)</u>, p 386 (cited CA 1969, <u>71</u>, 115291).
- 16. Bohne, H., 1973. Effect of dust emissions on plants. Helmt Staub - Reinhalt Luft, <u>33(12)</u>, p 489 (cited CA 1974, <u>81</u>, 5852).
- Wilfried, E., 1972. Zinc and cadmium pollution of soils and plants in the vicinity of a zinc smelting plant. Ber. Deut. Bot. Ges., 85(7-9), p 295 (cited CA 1974, 80, 99700).
- Greszta, J.; Godzik, S., 1969. Effect of zinc metallurgy on soils. Rocz. Glebozn, <u>20(1)</u>, p 195 (cited CA 1970, <u>72</u>, 78009).
- 19. Breshkov, T.; Nedyalkov, N.; Khinov, G.; Antonov, St., 1967. Air pollution effects on crops in the region of G. Damyonov copper plant at Pirdop - indirect effect of smoke through soil pollution. Rastenievud Nauki, <u>4(1)</u>, p 63 (cited CA 1967, <u>67</u>, 107712).
- 20. Orhein, R.M.; Lippman, L.; Johnson, C.J.; Bovee, H.H., 1974. Lead and arsenic levels of dairy cattle in proximity to a copper smelter. Environmental Letters, <u>7(9)</u>, p 229.
- 21. Chisnall, K.T.; Morkland, J., 1971. Contamination of pasture land by lead. Journal of the Association of Public Analysts. 9(4), p 116.

- 22. De Koning, H.W., 1974. Lead and cadmium in the area immediately surrounding a lead smelter. Water, Air and Soil Pollution, <u>3(1)</u>, p 63.
- 23. Leibetseder, J.; Skalicky, M.; Said, A.H.; Kment, A.; Glawischnig, E.; Schlerka, G., 1972. Toxic effects of hay damaged by industrial smoke on cattle. Enzmetall, <u>25(10)</u>, p 498 (cited CA 1973, <u>78</u>, 80471).
- 24. Torazzo, P.; Anselmetti, M.P., 1972. Case of lead pollution in Vencelli Province. Boll. Lab. Chim. Prov., <u>23(2)</u>, p 129 (cited CA 1973, 78, 33475).
- 25. Winchester, J.W.; Nifong, G.D., 1969. Water pollution in Lake Michigan by trace elements from pollution aerosol fallout. Paper submitted for publication in Chemistry of the Great Lakes (Advances in Chemistry Series). (Beeton, A.M.; Allen, H.E., Editors).
- 26. Carr-Brion, K.G.; Payne, K.W., 1970. X-ray Fluorescence Analysis. Analyst 95(1137), p 977.
- 27. Grennfeli, P.; Akerstom, A.; Brosset, C., 1971. Determination of filter collected airborne matter by X-ray fluorescence. Atmospheric Environment <u>5(1)</u>, p 1

- 28. Gilfrich, J.V.; Burkhalter, P.G.; Birks, L.S., 1973. X-ray spectrometry for particulate air pollution. Analytical Chemistry, <u>45(12)</u>, p 2002.
- 29. Dittrich, T.R.; Cothern, C.R., 1971. Analysis of trace metal particulates in atmospheric samples using X-ray fluorescence. Journal of the Air Pollution Control Association, <u>21(11)</u>, p 716.
- 30. Luke, C.L.; Kometani, T.Y.; Kessler, J.E.; Loomis, T.C.; Bove, J.L.; Nathanson, B., 1972. X-ray spectrometric analysis of air pollution dust. Environmental Science and Technology, <u>6(13)</u>, p 1105.
- 31. Chessin, H.; McLaren, E.H., 1973. X-ray spectrometric determination of atmospheric aerosols. Advances in X-ray Analysis, <u>16</u>, p 165
- 32. Birks, L.S., 1973. Application of X-ray fluorescence techniques to air pollution particles. Prepr. Pap. Natl. Meet., Div. Water, Air, Waste Chem., Am. Chem. Soc. <u>13(1)</u>, p 54 (cited CA 1974, <u>18</u>, 1755416).
- 33. Tamemura, T.; Suita, H., 1972. Air pollution analysis by fluorescent X-ray spectrometry using radioisotopes. Radioisotopes <u>21(11)</u>, p 641.

- 34. Moriyama, N., 1972. Application of fluorescent X-ray analysis to pollution measurement. Nenryo Oyolu Nensho, <u>39(11)</u>, p 1056 (cited CA 1973, <u>78</u>, 150832).
- 35. Wood, W.G., 1973. X-ray energy analysis of particulate matter on filter paper. Advances in X-ray Analysis 16, p 134.
- 36. Rhodes, J.R.; Pradzynski, A.H.; Sieberg, R.D., 1972. Energy dispersion X-ray emission spectrometry for multielement analysis of air particulates. Analytical Instruments <u>10</u>, p 143.
- 37. Giauque, R., 1974. Characterisation of aerosols in California by X-ray induced X-ray fluorescence analysis. Environmental Science and Technology <u>8(5)</u>, p 436.
- 38. Holynska, B., 1974. Use of chelating ion exchange in conjunction with radioisotope X-ray spectrometry for determination of trace amounts of metals in water. Radioanalytical Letters <u>17(5-6)</u>, p 313.
- 39. Tanaka, Hideki; Moriguchi, Yizo; Yamamoto, Tadashi; Hashizume, Genzo., 1972. Determination of traces of metals by X-ray fluorescence spectrometry II. Determination of metals in river and sea sediments by X-ray fluorescence spectrometry. Bunseki Kagaku, <u>21(11)</u>, p 1456 (cited CA 1973, <u>78</u>, 88378).

- 40. Watanbe, H.; Berman, S.; Russell, D.S., 1972. Determination of trace metals in water using X-ray fluorescence spectrometry. Talanta, <u>19(11)</u>, p 1363.
- 41. Toussaint, C.J.; Aina, G.; Broothaerts, J., 1972. X-ray spectroscopic determination of metals in water. EURATOM Rep. No. 5060, p 376 (cited CA 1974, 80, 40839).
- 42. Kajita, Tsutomu; Mizuno, Koichi, 1972. Determination of heavy metal ions in water by X-ray fluorescence spectrometry. Nagoya-Shi Kogye Kenkyusho Kenkyu Hokoku <u>48</u>, p 16 (cited CA 1973, <u>78</u>, 115067).
- 43. Bodart, F.; Deconninck, G.; Hontoy, J.; Wilk, S., 1973.
 Filter paper analysis by fluorescence measurements. Radioanalytical Letters, <u>13(3)</u>, p 161.
- 44. Gilfrick, J.V.; Burkhatler, P.G.; Birks, L.S., 1972. X-ray fluorescence analysis of air pollution particulates. Prep. Pap. Nat. Meet, Div. Water, Air, Waste Chemistry., Amer Chem., Soc. 12(2), p 222 (cited CA 1974, 81, 5833).
- 45. Wedberg, G.H.; Chan, Kwok-Chi; Cohan, B.L.; Frohliger, J.O.,
 1974. X-ray fluorescence study of atmospheric particulates in
 Pittsburgh. Environmental Science and Technology, <u>3(13)</u>, p 1090.

- 46. Lochmueller, C.H.; Galbraith, J.W.; Walter, R.L., 1974. Trace metal analysis in water by proton induced X-ray emission analysis of ion - exchange membranes. Analytical Chemistry, <u>46(3)</u>, p 440.
- Barendrecht, E., 1967. Stripping Voltametry. Eletroanalytical Chemistry, <u>2</u>, p 53.
- 48. Ebner, F.; Gams, H., 1973. Heavy metals in the Austrian Danube River. Oesterr Abwasser Rundsch, <u>18(3)</u>, p 47 (cited CA 1973, <u>79</u>, 128997).
- 49. Portretnyi, V.P.; Malyuta, V.F.; Chuiko, V.T., 1973. Determination of tin in water using its concentration by coprecipitation and stripping analysis. Zh. Anal. Khim, <u>28(7)</u>, p 1337 (cited CA 1974, <u>80</u>, 19272).
- 50. Siegerman, H.; O'Dom, G., 1972. Differential pulse anodic stripping of trace metals. American Laboratory, <u>4(6)</u>, p 59.
- 51. Rojahn, T., 1972. Determination of copper, lead, cadmium and zinc in estuarine water by anodic stripping voltametry alternating current voltametry on the hanging drop mercury electrode. Analytica Chimica Acta, <u>62(2)</u>, p 438.

- 52. Niguel, A.H.; Jankowski, C.H., 1974. Determination of trace metals in aquatic environments by anodic stripping voltametry with a vitreous carbon rotating electrode. Analytical Chemistry, <u>46(12)</u>, p 1832.
- 53. Macleod, K.E.; Lee, R.E., 1973. Selected trace metal determination of spot tape samples by anodic stripping voltametry. Analytical Chemistry, 45(14), p 2380.
- 54. Harrison, P.R.; Matson, W.R.; Winchester, J.W., 1971. Time variations of lead, copper and cadmium concentrations in aerosols in Ann Arbor Michigan. Atmospheric Environment, <u>5</u>, p 613.
- 55. Lininger, R.L.; Duce, R.A.; Winchester, J.W.; Matson, W.R., 1966. Chlorine, bromine, iodine and lead from Cambridge, Massachusetts. Journal of Geophysical Research, <u>71(10)</u>, p 2457.
- 56. Gillette, D.A.; Winchester, J.W., 1972. Aging of lead aerosols I. Observations. Atmospheric Environment, <u>6(7)</u> p 443.
- 57. Cohan, E., 1972. Determination of heavy metal traces in air by anodic stripping polarography. Bull. Soc. Chim. Fr. <u>1</u>, p 416 (cited CA 1972, <u>76</u>, 131120).

- 58. Kusuo, T., 1971. Suspended particles in Air I. Determination of copper, lead, cadmium and zinc in air by low-temp radio frequency oxidation and polarography. Eisei Skikenyo Hokoku, 89, p 21 (cited CA 1972, 76, 117142).
- 59. Gladowski, S., 1974. Determination of copper, lead, cadmium and zinc in atmospheric air by anodic stripping voltametry using the hanging mercury drop electrode. Chem. Anal. (Warsaw) 19(5), p 1041 (cited CA 1975, 82, 63722).
- 60. Colovas, G.; Wilson, G.S.; Moyers, J., 1973. Determination of trace amounts of zinc, cadmium, lead and copper in airborne particulate matter by anodic stripping voltametry. Analytica Chimica Acta. <u>64(3)</u>, p 457.
- Florence, T.M., 1974. Determination of bismuth in marine samples by anodic stripping voltametry. J. Electroanal. Chem. Interfacial Electrochem, <u>49(2)</u>, p 255 (cited CA 1974, <u>80</u>, 112392).
- 62. Searle, B.; Chan, Wing; Davidow, B., 1973. Determination of lead in blood and urine by anodic stripping voltametry. Clinical Chemistry <u>19(1)</u>, p 76.
- 63. Recosels D.; Heulens, L., 1970. Polarographic determination of lead in whole blood using the anodic stripping polarography technique. Biochemical Medicine, 4(2), p 89.

- 64. Sinko, I., Kosta, L., 1972. Determination of lead, cadmium, copper and zinc in biological materials by anodic stripping voltametry. International Journal of Environmental Analytical Chemistry, 2(2), p 167.
- 65. Reed, R.I., 1965. <u>Mass Spectrometry</u> Academic Prep. New York and London.
- 66. Phillips, M.A., 1973. Levels of both airborne beryllium and beryllium in coal at the Hayden power plant near Hayden, Colorado. Environmental Letters, 5(3), p 183.
- 67. Brown, R.; Vassen, P.G.T., 1970. Spark Source Mass Spectro-scopic survey analysis of air pollution particulates.
 Analytical Chemistry, <u>42(14)</u>, p 1880.
- Perry, R., 1971. Mass Spectroscopy in the determination and identification of air pollutants. Int. Symp. Identification Meas. Environ. Pollut. (Proc.) p 130 (cited CA 1974, <u>80</u>, 86978).
- 69. Brown, R., 1971. The quantitative analysis of inorganic pollutants in air and water using spark source mass spectroscopy. AEI Scientific Apparatus. publication TP 36.
- 70. Hamilton, E.I.; Minski, M.J., 1972. Comments on the trace element chemistry of water. Sampling a key factor in water quality surveillance. Environmental Letters, 3(1), p 53.

- 71. Wahlgren, M.A.; Edgington, D.N.; Rawlings, F.F., 1971. Determination of selected trace elements in water samples using spark source mass spectromtry and neutron activation analysis. Nucl. Methods Environ. Res., Proc, Amer, Nucl. Soc. Top. Meet. p 97 (cited CA 1973, 79, 96748)
- 72. Wahlgren, M.A., 1972. Trace element determination in Lake Michigan tributary water supplies using spark source mass spectrometry. Proc. Conf. Gt. Lake Res., <u>15</u>, p 298 (cited CA 1973, <u>79</u>, 9645).
- 73. Crocker, I.H.; Merritt, W.F., 1972. Analysis of environmental samples by spark source mass spectrometry. I. Trace elements in Water. Water Research, <u>6(3)</u>, p 285
- 74. Ball, D.F.; Barber, M.; Vassen, P.G.T., 1973. Use of spark source mass spectrometry for the determination of mercury in fish in environmental surveillance. Science of the Total Environment, 2(1), p 101.
- 75. Fitchett, A.W.; Buch, R.P.; Mushak, P., 1974. Direct determination of heavy metals in biological media by spark source mass spectrometry. Analytical Chemistry, <u>46(6)</u>, p 710.
- 76. Guidoboni, R.J., 1973. Determination of trace elements in coal and coal ash by spark source mass spectrometry.
 Analytical Chemistry, <u>45(7)</u>, p 1275.

- 77. Gallorini, M.; Genova, N.; Meloni, S.; Mascia, V., 1973. Determination of lead, manganese, aluminium and vanadium in atmospheric dust using flame emission atomic absorption. Inquinamento, <u>15(9)</u>, p 13 (cited CA 1974, <u>80</u>, 136958).
- 78. Imai, Sakingo; Ito, Kiyoshi, 1971. Emission spectrographic determination of trace elements in air pollutants. Application of a coprecipitation method with organic chelating agents. Taiki Osen Kenkyw, <u>6(3)</u>, p 322 (cited CA 1972, 77, 38753).
- 79. Yamane, Yaswhiro; Miyazaki, Motoich; Nakazawa, Hiroyuki, 1973. Spectrographic determination of trace elements in airborne particulates. Bunseki Kayaku, <u>22(9)</u>, p 1135. (cited CA 1974 <u>80</u>, 73857).
- 80. Laamanen, A.; Lofgren, A; Portamen, T.J., 1971. Trace metals in particulates in the air of Helsinki and Turku 1964-1968. Work Environ. Health. <u>8(2)</u>, p 63, (cited CA 1972, <u>77</u>, 130187).
- 81. Jugimae, Akiyoshi,1974. Emission spectrographic determination of trace elements in airborne particulate matter. Analytical Chemistry, <u>46(8)</u>, p 1123.
- 82. Lander, D.W.; Steiner, R.L.; Anderson, D.H.; Dehm, R.L.,
 1971. Spectrographic determination of elements in airborne dirt. Applied Spectroscopy, <u>25(2)</u>, p 270.
- 83. Sugimae, Akiyoshi; Hasegawa, Toshio, 1973. Emission spectrographic determination of trace metals in airborne particulates collected on membrane filter. Bunseki Kayaku, <u>22(1)</u>, p 3 (cited CA 1973, 79, 23145).
- 84. Sugimae, Akiyoshi, 1975. Emission spectrographic determination of trace elements in airborne particulate matter collected on glass fibre filters. Analytica Chimica Acta, <u>78(1)</u>, p 107.
- 85. Frigieri, P.; Trucco, R.; Anzani, R.; Caretta, F, 1972. Spectrographic analysis of elements in airborne material. Chim. Ind. (Milan), <u>54(1)</u>, p 12 (cited CA 1972, <u>76</u>, 103405).
- 86. Seekey, J.L.; Skogerloe, R.K., 1974. Combined sampling analytical method for the determination of trace elements in atmospheric particulates. Analytical Chemistry, <u>46(3)</u>, p 415.
- 87. Pepin, D.; Gardes, A.; Petit, J.; Berger, J.A.; Gailland, G., 1973. Spectrographic study of the trace elements in mineral waters. I^T. Determination of some elements after extracting Cu, Ni, Sn, Ag, Pb, V and Mo. Analysis, 2(8), p 549 (cited CA 1974, 80, 112385).
- 88. Rodriguez Vazquez, J.A., 1974. Quantative inorganic analysis by gas-chromatography. Analytica Chimica Acta, <u>73(1)</u>, p 1.

- 89. Sievers, R.E.; Ponder, B.W.; Morris, M.L.; Mashier, R.W., 1963. Gas phase chromatography of metal chelates of acetylacetone, trifluoroacetylacetone and hexafluoracetylacetone. Inorganic Chemistry, 2(4), p 693.
- 90. Fujinaga, T.; Kuwamoto, T.; Murai, S., 1974. Gas chromatography of metal chelates with carrier gas containing ligand vapour Thorium (IV) trifluoroacetylacetonate. Analytica Chimica Acta, 71(1), p 141.
- 91. Ross, W.D.; Sievers, R.E., 1972. Environmental air analysis for ultratrace concentrations of beryllium by gas chromatography. Environmental Science and Technology. <u>6(2)</u>, p 155.
- 92. Drugov, Yu. S.; Murov eva, G.V.; Grimberg, K.M.; Nesterenko, G.N.; Sokolov, D.N., 1972. Gas chromatographic determination of beryllium in air. Zavod. Lab. 38(11), p 1305 (cited CA 1973, <u>78</u>, 88284).
- 93. Yovorovskaya, S.F.; Brimberg, K.M., 1974. Gas chromatographic determination of metals (Be, Cr, Al) in the air of aerosols. Gig. Sanit, (11), p 5/ (cited CA1975, 82, 76598).
- 94. Lee, M.L.; Meng-Lein, Lee; Durrell, D.C., 1973. Soluble aluminium in marine and fresh water by gas liquid chromatography. Analytica Chimica Acta, <u>66(2)</u>, p 245.

- 95. Yasuaki, Shimoishi, 1973. Determination of selenium in seawater by gas chromatography with electron capture detection. Analytica Chimica Acta, $\underline{64(2)}$, p 465.
- 96. Gosink, T.A., 1975. Rapid simultaneous determination of picogram quantities of aluminium and chromium from water by gas phase chromatography. Analytical Chemistry, <u>47(1)</u>, p 165.
- 97. Minear, R.A.; Palesh, C.M., 1974. Application of gas chromatography of volatile metal chelates in water analysis. Chem. Water Supply. Treat. Distrib. Symp. 1973, p 38 (cited CA 1974, <u>18</u>, 175981).
- 98. Ross, W.D.; Hansen, L.C.; Scribner, W.G., 1971. Trace analysis for metals in aerospacematerials by gas chromatography. Nat. Tech. Inform. Serv. AD Rep. 1971. No. 739865 p 239 (cited CA 1972, <u>77</u>, 96583).
- 99. Talmi, Yair; Norvell, V.E., 1975. Determination of arsenic and antimony in environmental samples using gas chromatography with a microwave emission spectrometric system. Analytical Chemistry, 47(9), p 1510.
- 100. Yair, Talmi; Andren, A.W., 1974. Determination of selenium in environmental samples using gas chromatography with a microwave emission spectrometric detection system. Analytical Chemsitry, 46(14), p 2122.

- 101. Barratt, R.S.; Belcher, R.; Stephen, W.I.; Uden, P.C., 1972. The determination of traces of nickel by gas liquid chromatography. Analytica Chimica Acta. 59(1), p 59.
- 102. Nakamuro, Katsukiko; Sayato, Yasuyoshi; Tonomura Masaharu Ose Youki, 1972. Selenium-related compound I. Determination of eelenium by gas chromatogrsphy. Eisei Kagaku, <u>18(4)</u>, p 237 (cited CA 1973, <u>78</u>, 67735).
- 103. Hartung, R., 1972. Determination of mono and di-methyl mercury compounds by gas chromatography. Environ. Mercury Contam. Int. Conf. 1970, p 157 (cited CA 1973, <u>78</u>, 53510).
- 104. Foreman, J.K.; Gough, T.A.; Walker, E.A., 1970. Determination of traces of beryllium in human and rat urine samples by gas chromatography. Analyst (London), <u>95(1134)</u>, p 797.
- 105. Savory, J.; Mushak, P.; Sunderman, F.W., 1969. Gas chromatographic determination of chromium in serum. Journal of Chromatographic Science, <u>7(11)</u>, p 674.
- 106. Lenihan, J.M.A.; Thomson, S.J., 1965. <u>Activation Analysis</u>, Academic Press, New York, London and San Francisco.
- 107. Aras, N.K.; Zoller, W.H.; Gordon, G.E.; Lutz, G.U., 1973. Instrumental photon activation analysis of atmospheric particulate matter. Analytical Chemistry, <u>45(8)</u>, p 1481.

- 108. Almez, I.; Aras, N.K.; Gordon, G.E.; Zoller, W.H., 1974. Nondestructive analysis for silicon, rubidium and yttrium in atmospheric particulate material. Analytical Chemistry, 46(7), p 935.
- 109. Lutz, G.J., 1971. Determination of lead in environmental samples by photon activation analysis. Nucl. Methods. Environ. Res. Proc. Amer. Nucl. Soc. Top. Meet, p 144 (cited CA 1973, <u>79(6)</u>, 96398).
- 110. Lutz, G.J.; Aras, N.K.; Zoller, W.H.; Gordon, G.E., 1973. Multielement analysis of atmospheric particulate matter by instrumental photon activation. Journal of Radioanalytical Chemistry, <u>16(2)</u>, p 609.
- 111. Riddle, D.C.; Schweikert, E.A., 1974. Determination of lead using charged particle activation analysis. Analytical Chemistry, 46(3), p 395.
- 112. Desaedeleer, G.; Ronneau, C.; Apens, D., 1976. Lead determination in airborne particulate matter by proton activation analysis. Analytical Chemistry, <u>48(3)</u>, p 572.
- 113. Friarman, K., 1972. Proton activation analysis for trace elements. Environmental Science and Technology, <u>6(1)</u>, p 79.
- 114. Bankart, S.F.; Bloom, S.D.; Sauter, G.D., 1973. Trace element analysis in water by proton activation. Analytical Chemistry, 45(4), p 692.

- 115. Van Grieken, R.; Dams, R., 1973. Determination of silicon in natural and pollution aerosals by 14 - MeV neutron activation analysis. Analytica Chimica Acta, 63(2), p 369.
- 116. Kronborg, O.J.; Steinnes, E., 1975. A routine procedure for multielement analysis of atmospheric particulates by instrumental neutron activation analysis. Radiochemical and Radionanalytical Letters, <u>21(6)</u>, p 379.
- 117. Peirson, D.H.; Cawse, P.A.; Salmon, L.; Cambray, R.S., 1973. Trace elements in the atmospheric environment. Nature, <u>241</u>, p 252.
- 118. Harrison, P.R.; Rahn, K.A.; Dams, R.; Robbins, J.A.; Winchester, J.W.; Brar, S.S.; Nelson, D.M., 1971. Areawide trace metal concentrations measured by multielement neutron activation analysis. Journal of the Air Pollution Control Association, <u>21(9)</u>, p 563.
- 119. Hewitt, P.J., 1972. Instrumental neutron activation analysis of airborne contaminants using Ge/Li detectors. Annals of Occupational Hygiene. <u>15</u>, p 341.
- 120. Rahn, K.; Wesolowski, J.J.; John,W.; Ralston, H.R., 1971. Diurnal variation of aerosol trace element concentrations in Livermore, California. Journal of the Air Pollution Control Association, <u>21(7)</u>, p 406.

- 121. John W.; Kaifer, R.; Rahn, K.; Wesolowski, J.J., 1973. Trace element concentrations in aerosols from the San Francisco Bay area. Journal of the Atmospheric Environment, 7(1), p 107.
- 122. Mamuro, Tetsuo; Matsuda, Yatsuka; Mizohata, Akira; Matsunami, Tadac, 1972. Activation analysis of airborne dust II. Radioisotopes, 21(3), p 164.
- 123. Pillay, K.K.S.; Thomas, C.C., 1971. Determination of the trace element levels in atmospheric pollutants by neutron activation analysis. Journal of Radioanalytical Chemistry, <u>7(1)</u>, p 107.
- 124. Dams, R.; Robbins, J.A.; Rahn, K.A.; Winchester, J.W., 1970. Nondestructive neutron activation analysis of air pollution particulates. Analytical Chemistry, <u>42(8)</u>, p 861.
- 125. Zoller, W.H.; Gordon, G.E., 1970. Instrumental neutron activation analysis of atmospheric pollutants utilising Ge (Li) X-ray detectors. Analytical Chemistry, <u>42(2)</u>, p 257.
- 126. Mamuro, Tetsuo; Matsuda, Yatsuka, Mizohata, Akira; Takeuchi, Takaguki; Fujita, Akira, 1971. Activation analysis of airborne dust. Radioisotopes, <u>20(3-4)</u>, p 117.
- 127. Tanner, T.M.; Young, J.A.; Cooper, J.A., 1974. Multielement analysis of St. Louis aerosols by nondestructive techniques. Chemosphere, <u>3(5)</u>, p 211.

- 128. Belot, Y.; Marini, T., 1974. Neutron activation analysis of soild atmospheric pollutants with an internal standard technique. Journal of Radioanalytical Chemistry, <u>19(2)</u>, 0 319
- 129. Bogen, J., 1973. Trace elements in atmospheric aerosols in the Heidelberg area, measured by instrumental neutron activation analysis. Journal of the Atmospheric Environment. T(11), p 1117.
- 130. Mamuro, Tetsuo; Matsuda, Yatsuka; Mizohata, Akira, 1972. Survey of an air pollution source by instrumental neutron activation analysis. Radioisotopes, <u>21(3)</u>, p 183.
- 131. Dale, I.M.; Duncan, H.J.; McDonald, C., 1973. Neutron activation analysis of atmospheric particulates. Radiochemical and Radioanalytical Letters, <u>15(2)</u>, p 77.
- 132. Brune, D., 1973. Interference-free trace element analysis of water, biological specimens and food sources by means of nuclear techniques. The Science of the Total Environment, 2(1), p 111.
- 133. Clemente, G.F.; Mastinu, G.G., 1974. Instrumental method for the determination of trace elements in water samples by neutron activation analysis. Journal of Radioanalytical Chemistry, <u>20(2)</u>, p 707

- 134. Thatcher, L.L.; Johnson, J.O., 1973. Determination of trace elements in water and quatic biota by neutron activation analysis. Bioassay Tech. Environ. Chem. p 277 (cited CA 1973, <u>79</u>, 57393).
- 135. Kay, M.A., McKown, D.M.; Gray, D.H.; Eicher, M.E.; Vogt, J.R.,
 1973. Neutron Activation Analysis in environmental chemistry.
 American Laboratory, <u>5(7)</u>, p 39.
- 136. Kolaczkowski, A.; Jester, W.A., 1973. Activation analysis of heavy metals in surface waters using ion exchange filter papers and cyanide complexing. Journal of Radioanalytical Chemistry, . <u>16(1)</u>, p 21.
- 137. Merritt, W.F., 1971. Identification and measurement of trace elements in fresh water by neutron activation. Int. Symp. Ident. Meas. Environ. Pollut. (Proc.) p 358 (cited CA 1974, <u>80</u>, 87351).
- 138. Fuijnoga, T.; Kusaka, Y.; Koyama, M.; Tsuji, H.; Mitsuji, T.; Imai, S.; Okuda, J.; Takamatsu, T.; Ozaki, T., 1973. Radioactivation analysis of Al, V, Cu, Mo, Zn and U in natural water semples using organic coprecipitants. Journal of Radioanalytical Chemistry, <u>13(2)</u>, p 301
- 139. Das, H.A.; Evendijk, J.E.; De Jong, J.P.M., 1973. Determination of nickel in air dust by neutron activation analysis. Journal of Radioanalytical Chemistry, <u>13(2)</u>, p 413.

- 140. Pillary, K.K.S., 1971. Activation analysis of airborne selenium as a possible indicator of atmospheric sulphur pollutants. Environmental Science and Technology, 5(1), p 74.
- 141. Livingston, H.D.; Smith, H.; Stojanovic, N., 1967. Simultaneous determination of copper, zinc, cadmium and mercury in biological material by neutron activation analysis. Talanta, <u>14</u>, p 505.
- 142. Steinnes, E., 1967. Determination of traces of selenium in biological tissue by neutron activation. International Journal of Applied Radiation and Isotopes, <u>18</u>, p 731.
- 143. Sjostrand, B., 1964. Simultaneous determination of mercury and arsenic in biological and organic materials by activation analysis. Analytical Chemistry, 36(4), p 814.
- 144. Tanner, T.M.; Rancitelli, L.A.; Haller, W.A., 1972. Multielement analysis of natural waters by neutron activation group chemical separations, and gamma-ray spectrometric techniques. Water, Air and Soil Pollution, <u>1</u>, p 132.
- 145. De Goejj, J.J.M.; Guinn, V.P., Young, D.R.; Mearns, A.J., 1973. Activation Analysis. Trace element studies of Dover sole liver and marine sediments. I.A.E.A./FAO/WHO Symposium. Nuclear Techniques in Comparative Studies of Food and Environmental Contamination, Otamiemi (Finland), 27-31 August.

- 146. Willshire, F.W.; Lambert, J.P., Butler, F.E., 1975. Rapid radiochemical separation of selected toxic elements in environmental samples prior to gamma ray spectrometry. Analytical Chemistry, <u>47(14)</u>, p 2399.
- 147. Crouthamel, C.E., 1970. <u>Applied X-ray Spectroscopy</u>. Revised by Adams, F., Dams, R., Pergamon Press, Oxford, New York, Toronto, Sydney, Braunschweig.
- 148. Kirkbright, G.F., Sargent, M., 1974. <u>Atomic Absorption and</u> <u>Fluorescence Spectroscopy</u>. Academic Press, New York, London and San Francisco.
 - ۲
- 149. Elwell, W.T.; Gidley, J.A.F., 1966. <u>Atomic Absorption</u> <u>Spectrophotometry</u>. Second (revised) Edition. Pergamon Press, Oxford, London, Edinburgh, New York, Toronto, Sydney, Pairs, Braunschweig.
- 150. Culver, B.R.; Suries, T., 1975. Interference of molecular spectra due to alkali halides in non-flame atomic absorption spectrometry. Analytical Chemistry, <u>47(6)</u>, p 920.
- 151. Ottaway, J.M., 1975. The atomisation of metal oxides using carbon furnace atomic-absorption spectrometry. Proceedings of the Analytical Division of the Chemical Society, <u>12(6)</u>, p 176.
- 152. Campbell, W.C.; Ottaway, J.M., 1974. Atom-formation processes in carbon-furnace atomizers used in atomic-absorption spectrometry. Talanta, <u>21</u>, p 837.

- 153. Fuller, C.W., 1974. A kinetic theory of atomisation for nonflame atomic-absorption spectrometry with a graphite furnace. The kinetics and mechanism of atomisation for copper. Analyst, <u>99</u>, p 739.
- 154. Maebashi,K; Katuragawa, H., 1970. Atomic absorption of atmospheric aerosols. Tenki, <u>17(11)</u>, p 549 (cited CA 1972, <u>77</u>, 24319).
- 155. Thompson, R.J.; Morgan, G.B.; Purdue, L.J., 1969. Analysis of selected elements in atmospheric particulate matter by atomic absorption. Analytical Instruments, 7, p 9.
- 156. Huang, J.Y.; Feldman, F.J., 1970. Determination of atmospheric trace elements by atomic absorption spectroscopy. Applied Spectroscopy, <u>24(3)</u>, p 371.
- 157. Kometani, T.Y.; Bove, J.L.; Nathanson, B.; Siebenberg, S.; Magyar, M., 1972. Dry ashing of airborne particulate matter on paper and glass fibre filters for trace metal analysis by atomic absorption spectrometry. Environmental Science and Technology, <u>6(7)</u>, p 617.
- 158. Jackson, G.B.; Myrick, H.N., 1970. Analytical methods utilising AAS (atomic absorption spectroscopy) for iron, lead and zinc metallic ions in airborne particulates. American Laboratory, <u>August</u> p 19.

- 159. Tusuji, K., 1972. Suspended particles in air. II. Determination of zinc, cadmium, lead and copper in air by low temperature radio frequency oxidation apparatus and atomic absorption spectroscopy. Eissi Shikenjo Hokoku, <u>90</u>, p 33 (cited CA 1973, <u>78</u>, 150859).
- 160. Oddo, N., 1972. Application of atomic absorption spectroscopy to the analysis of polluting elements. Chim. Ind., <u>54(11)</u>, p 1009. (cited CA 1973, <u>78</u>, 139770).
- 161. Ranweiler, L.E.; Moyers, J.L., 1974. Atomic absorption procedure for analysis of metals in atmospheric particulate matter. Environmental Science and Technology, 8(2), p 152.
- 162. Boudene, C.; Chovin, P.; Favart, L.; Truffert, L.; Godin, J.,
 1973. Determination of lead in suspension in atmospheric dusts.
 Igiena, 28(1), p 37 (cited CA 1974, 80, 18883).
- 163. Beyer, M., 1969. Preparation of metallic air pollutants for atomic absorption analysis. Atomic Absorption Newsletter, 8(1), p 23.
- 164. Sachder, S.L.; Robinson, J.W., West, P.W., 1967. Determination of manganese, iron, cobalt and nickel in air and water by atomic absorption spectroscopy. Analytica Chimica Acta, <u>38(2)</u>, p 499.
- 165. Hwang, J.Y., 1972. Trace metals in atmospheric particulates. Analytical Chemistry, <u>44(14)</u>, p 20A.

- 166. Haschler, M.E.; Kanabrocki, E.L.; Moore, C.E.; Hattari, D.M., 1973. Determination of calcium, magnesium, copper and iron in airborne particulates over Chicago metropolitan area. Applied Spectroscopy, <u>27(3)</u>, p 185.
- 167. Hwang, J.Y., 1971. Lead analysis in air particulate samples by atomic absorption spectrometry. Canadian Spectroscopy, <u>16(2)</u>, p 43.
- 168. Janssens, M.; Dams, R., 1973. Determination of lead in atmospheric particulates by flameless atomic absorption spectrometry with a graphite tube. Analytica Chimica Acta, $\underline{65(1)}$, p 41.
- 169. Janssens, M.; Dams, R., 1974. Determination of cadmium in air particulates by flameless atomic absorption spectrometry with a graphite tube. Analytica Chimica Acta, <u>70(1)</u>, p 25.
- 170. Cristian, C.M.; Robinson, J.W., 1971. The direct determination of cadmium and mercury in the atmosphere. Analytica Chimica Acta, <u>56(2)</u>, p 466.
- 171. Omang, S.H., 1971. The determination of lead in air by flameless atomic absorption spectrophotometry. Analytica Chimica Acta, 55(2), p 439.
- 172. Siemer, D.D.; Woodriff, R., 1974. Direct atomic absorption determination of metallic pollutants in air with a carbon fod atomizer. Spectrochimica Acta Part B, <u>29B(9-10)</u>, p 269.

- 173. Woodriff, R.; Lech, J.F., 1972. Determination of trace lead in the atmosphere by furnace atomic absorption. Analytical Chemistry, <u>44(7)</u>, p 1323.
- 174. Matowsek, J.P.; Brodie, K.G., 1973. Direct determination of lead airborne particulates by non-flame atomic absorption. Analytical Chemistry, <u>45(9)</u>, p 1606.
- 175. Brodie, K.G.; Matowsek, J.P., 1974. Determination of cadmium in air by non-flame atomic absorption. Analytica Chimica Acta, <u>69(1)</u>, p 200.
- 176. Begnoche, B.C.; Risby, T.H., 1975. Determination of metals in atmospheric particulates using low volume sampling and flameless atomic absorption spectrometry. Analytical Chemistry, <u>47(7)</u>, p 1041.
- 177. Zdrojewski, A.; Quickert, M.; Dubois, L., 1973. Accurate measurement of cadmium in airborne particulates. International Journal of Environmental Analytical Chemistry, <u>2(4)</u>, p 331.
- 178. Childs, E.A.; Gaffke, J.N., 1974. Organic solvent extraction of lead and cadmium from aqueous solutions for atomic absorption spectrophotometric measurement. J. Ass. Off. Anal Chem. <u>57(2)</u>, p 360 (cited CA 1974, <u>81</u>, 24252).
- 179. Goulden, P.D.; Brooksbank, P., 1974. Automated atomic
 absorption determination of arsenic, antimony and selenium in
 natural waters. Analytical Chemistry, <u>46(11)</u>, p 1431.

- 180. Tinsley, D.A.; Idden, A., 1974. Use of a liquid iOnexchanger in the solvent extraction and atomic absorption determination of trace copper in water. Talanta, <u>21(6)</u>, p 633.
- 181. Kulo, Yasuo; Nakazawa, Noboru; Sato, Masao, 1973. Determination of trace amounts of lead, chromium and copper in sea water by solvent extraction atomic absorption spectrophotometry. Sekiyu Gakkai Shi, <u>16(7)</u>, p 588 (cited CA 1973, <u>79</u>, 108001).
- 182. Uno, Toyoko; Kobayashi, Shigeo, 1972. Determination of lead in waste water from mines by atomic absorption spectroscopy. Hokkaido - Ritsu Kogyo Shikenyo Hokoku, <u>204</u>, p 17 (cited CA 1974, <u>80</u>, 63606).
- 183. Morrow, R.W.; McElhaney, R.J., 1974. Determination of chromium in industrial waste water by flameless atomic absorption. Nucl. Sci. Astr., <u>29(2)</u>, 4787 (cited CA 1974, <u>80</u>, 112200).
- 184. Fernandez, F.J.; Manning, D.C., 1971. Atomic absorption analysis of metal pollutants in water using a heated graphite atomizer. Atomic Absorption Newsletter, <u>10(3)</u>, p 65.
- 185. Edmunds, W.M., Giddings, D.R.; Morgan-James, M., 1973. Application of flameless atomic absorption in hydrogeochemical analysis. Atomic Absorption Newsletter, <u>12(2)</u>, p 45.

- 186. Yasuda, Seiji; Kakiyama, Hitoo, 1974. Determination of trace amounts of cadmium by flameless atomic absorption spectroscopy using a carbon rod atomizer. Bunseki, Kagaku, <u>23(4)</u>, p 406 (cited CA 1974, 81, 41240).
- 187. Yasuda, Seiji, Kakiyama, Hitoo, 1974. Determination of trace amounts of copper and lead in water by flameless atomic absorption spectroscopy. Bunseki, Kagaku, <u>23(6)</u>, p 670 (cited CA 1974, 81, 158535).
- 188. Lerche, H.; Breckle, S.W., 1974. Lead in the ecologic system of the highway periphery. Naturwissenschaften, <u>61(5)</u>, p 218 (cited CA 1974, <u>81</u>, 67913).
- 189. Lauterback, K.; Gottschling, E., Koenig, R., 1974. Contamination of roadway apples with lead. Nahrung, <u>18(4)</u>, p 461 (cited CA 1974, <u>81</u>, 150445).
- 190. Sapetti, C.; Arduino, E., 1973. Lead pollution in the area of Toriro - Milano highways. Agrochimica, <u>17(6)</u>, p 540 (cited CA 1974, <u>81</u>, 28988).
- 191. Williamson, P.; Evans, P.R., 1972. Lead levels in roadside invertebrates and small mammals. Bull. Environ. Contam. Toxicol., <u>8(5)</u>, p 280 (cited CA 1973, <u>78</u>, 67808).

- 192. Gish, C.D.; Christensen, R.E., 1973. Cadmium nickel, lead and zinc in earthworms from roadside soil. Environmental Science and Technology, <u>7(11)</u>, p 1060.
- 193. Price, P.W.; Ratcke, B.J., Gentry, D.A., 1974. Lead in terrestrial arthropods. Evidence for biological concentration. Environmental Entomology, 3(3), p 370.
- 194. Stern, A.C., 1968. <u>Air Pollution I</u> 2nd Edition Academic Press, New York and London.
- 195. Ledbetter, J.O., 1972. <u>Air Pollution Part A</u> <u>Analysis</u>. Marcel Dekker Inc., New York.
- 196. Schaefer, V., 1969. The inadvertent modification of the atmosphere by air pollution. Bulletin of the American Meteorological Society, <u>50</u>, p 199.
- 197. Ashworth, J.R., 1929. The influence of smoke and hot gases from factory chimneys on rainfall. Quarterly Journal of the Royal Meteorogical Society, <u>55</u>, p 341.
- 198. Changnon, S.A., 1968. The Laporte precipitation anomaly Bulletin of the American Meteorogical Society, <u>49</u>, p 4.
- 199. Ogden, T.L., 1969. The effect on rainfall of a large steelworks. Journal of Applied Meteorology, <u>8</u>, p 585.

- 200. Ricks, G.R.; Williams, R.J.M., 1974. Effects of atomospheric pollution on deciduous woodland. 2. Effects of particulate matter upon stomatal diffusion resistance in leaves in quercus petraea. Environmental Pollution, 6(2), p 87.
- 201. Singhal, R.L.; Kacew, S.; Sutherland, D.J.B., 1974. Persistence of cadmium - induced metabolic changes in liver and kidney. Science, <u>183</u>, p 1094.
- 202. Gieske, T.H., Foulkes, E.C., 1974. Acute effects of cadmium on proximal tubular function in rats. Toxicology and applied Pharmacology, <u>27(2)</u>, p 292.
- 203. Toshioka, Teruo; Ishida, Masaaki; Iwasaki, Kinji; Koyanagi, Noahisa; Isobe, Mari; Kubo, Tokuko; Aizawa, Ken, 1973. Effects of cadmium on the function of several systems in the blood with special reference to bactericidal system I. Preliminary Experiments. Nihon. Univ. J. Med. <u>15(4)</u>, p 281 (cited CA 1974, <u>80</u>, 141623).
- 204. Whanger, P.D., 1973. Effects of dietary nickel on enzyme activities and mineral contents in rats. Toxicology and Applied Pharmocology, <u>25(3)</u>, p 323.
- 205. Sokolovskii, V.V.; Davletov, E.G., 1974. Mechanism of the hemolytic action of heavy meatls. Tsitologiya, <u>16(5)</u>, p 684 (cited CA 1974, <u>81</u>, 100304).

- 206. Chandra, S.V.; Iman, Z., 1973. Manganese induced histochemical and histological alterations in gastrointestinal mucosa of guinea pigs. Acta Pharmacologica Toxicologica, <u>32(5-6)</u>, p 449.
- 207. Antov, G.; Zlateva, M.; Vulcheva, V., 1973. Changes in the activity of some phosphatases in renal and hepatic tissues under the effect of arsenic and lead. Khig, Zdraveopazvare, <u>16(6)</u>, p 543 (cited CA 1974, <u>81</u>, 100278).
- 208. Paulet, G.; Chevrier, R., 1969. Effect of cobalt on carbon monoxide detoxification. C.R. Soc. Biol., <u>163(5)</u>, p 1199 (cited CA 1970, <u>73</u>, 23634).
- 209. Jones, R.B.; Nelson, D.P.; Shapiro, S.; Kiesow, L.A., 1974. Symergism in the toxicities of lead and oxygen. Experientia, <u>30(4)</u>, p 327.
- 210. Sansi, K.A.O.; Pond, W.G., 1974. Pathology and dietary cadmium toxicity in growing rats and the protective effect of injected iron. Nutr. Rep. Int., <u>9(6)</u>, p 407 (cited CA 1974, <u>81</u>, 100322).
- 211. Lee, B.J., 1971. Effects on cadmium on rats' teeth and on the compensating effects of chromium. Yonse Uidae Nonmunjip, <u>4(2)</u>, p 55 (cited CA 1972, 77, 97428).

- 212. Webb, M., 1972. Protection by zinc against cadmium toxicity. Biochemical Parmacology, <u>21(20)</u>, p 2767.
- 213. Willoughby, R.A.; MacDonald, E.; McSherry, B.J., 1972. Interaction of toxic amounts of lead and zinc fed to young growing horses. Veterinary Record, 91(16), p 382.
- 214. Gantler, H.E.; Goudie, C.; Sunde, M.L.; Kopecky, M.J.; Wagner, P.; Oh Sang-Hwan; Hoekstra, W.G., 1972. Selenium relation to decreased toxicity of methyl-mercury added to diets containing tuna. Science, <u>175(4026)</u>, p 1122.
- 215. Stern, A.C., 1968. <u>Air Pollution II.</u> 2nd Edition, Academic Press, New York and London.
- 216. Martinei, J.D., Natharny, M.; Dharmarajan, V., 1971. Spanish moss - a sensor for lead. Nature (London), 223(5321), p 564.
- 217. Goodman, G.T.; Roberts, T.M., 1971. Plants and soils as indicators of metals in the air. Nature (London), <u>231(5301)</u>, p 287.
- 218. Shacklette, H.T.; Connor, J.J., 1973. Airborne chemical elements in spanish moss. U.S. Geol. Surv. Prof. Pap. No. 574-E, p 46 (cited CA 1974, <u>80</u>, 141585).
- 219. Gill, R.; Martin, M.H.; Nickless, G.; Shaw, T.L., 1975. Regional monitoring of heavy metal pollution. Chemosphere, <u>4(2)</u>, p 113.

- 220. Roberts, T.M., 1972. Plants as monitors of airborne metal pollution. Journal of Environmental Planning and Pollution Control, <u>1(1)</u>, p 43.
- 221. Rukling, A.; Tyler, G., 1973. Heavy metal deposition in Scandinavia. Water, Air and Soil Pollution, 2(4), p 445.
- 222. Elgmork, K.; Hagen, A.; Langeland, A., 1973. Polluted snow in Southern Norway during winters. Environmental Pollution, <u>4(1)</u>, p 41.
- 223. Giessing, Y.T.; Giessing, E.T., 1973. Chemical composition of an Antarctic snow profile. Vatten, <u>29(3)</u>, p 233 (cited CA 1974, <u>80</u>, 136905).
- 224. Weiss, H., 1972. Review of lead, sulphur, selenium and mercury in permanent snowfields. Mar. Electrochem. Pap. Symp. (Published 1973) p 385 (cited CA 1974, <u>81</u>, 95987).
- 225. Jernelov, A.; Walbin, T., 1973. Airborne mercury fallout on snow around five Swedish chloroalkali plants. Atmospheric Environment, 7(2), p 209.
- 226. Kerin, Z., 1973. Lead in new fallen snow near a leadsmelter. Archives of Environmental Health, <u>26(5)</u>, p 256.
- 227. La Barrie, N.; Milne, J.B.; Oliver, B.G., 1973. Lead contamination of snow. Water Research, <u>7(8)</u>, p 1215.

- 228. Hamilton, W.L.; Miller, J.E., 1972. High lead concentrations in Columbus snow. Ohio Journal of Science, <u>71(5)</u>, p 313.
- 229. Kainz, G.; Boehm, G.; Schoeller, F., 1973. Heavy metal content of snow samples, ground and surface water. Gas: Wasser Waerme <u>27(4)</u>, p 84 (cited CA 1973, 79, 23125).
- 230. Hagen, A.; Langeland, A., 1973. Polluted snow in Sern, Norway, and the effect of the melted water on fresh water and acquatic organisms. Environmental Pollution, <u>5(1)</u>, p 45.
- 231. Rolfe, G.L., 1974. Lead distribution in tree rings. Forensic Science, <u>20(3)</u>, p 283.
- 232. Ishizaki, Arinobu.; Fukushima, Masaaki; Kurochi, Teru; Sakamato, Michiko; Kayashi, Etsuko., 1970. Relation between cadmium and zinc contents and annual rings of Cryptomeria japonica in the basin of the Jinzu River. Nippon Eiseigaku Zasshi, <u>25(4)</u>, p 376 (cited CA 1973, <u>78</u>, 38883).
- 233. Lepp, N.W., 1975. The potential of tree-ring analysis for monitoring heavy metal pollution patterns. Environmental Pollution, 9(1), p 49.
- 234. Tansy, M.F.; Roth, R.P., 1970. Pigeons: A new role in air pollution. Journal of the Air Pollution Control Association, 20(5), p 307.

- 235. Braja, M.; Natale, P.; Marchese, G., 1972. Atmospheric pollution from lead in Turin. Ig. Mod., <u>65(9-10)</u>, p 439 (cited CA 1973, <u>79</u>, 69855).
- 236. <u>National Survey of Smoke and Sulphur Dioxide</u>, Instruction Manual. 1966. Warren Spring Laboratory.
- 237. Bowen, H.J.M., 1969. Standard materials and intercomparisons. Advances in Activation Analysis, <u>1</u>, p 101.
- 238. <u>Handbook of Chemistry and Physics</u>, 1972. 53rd Edition. Chemical Rubber Company, Ohio.
- 239. Covell, D.F., 1959. Determination of gamma-ray abundance directly from the total absorption peak. Analytical Chemistry, <u>31(11)</u>, p 1785.
- 240. Verheijke, M.L., 1972. A computer programme in Algol-60 for the location and evaluation of peaks in Ge (Li) gamma-ray spectra. Journal of Radioanalytical Chemistry, <u>10</u>, p 299.
- 241. Gorski, L.; Heinonen, J.; Suschny, O., 1973. Final Report on simultated air filter analysis intercomparison Air - 1, Air - 2, Air - 3. International Atomic Energy Agency, I.A.E.A./R.L./21, Vienna.

- 242. Rhodes, J.R.; Pradzyniski, A.H.; Hunter, C.B.; Payne, J.S.; Lindgren, J.L., 1972. Energy dispersive X-ray fluorescence analysis of air particulates in Texas. Environmental Science and Technology, <u>6(10)</u>, p 922.
- 243. Belot, Y.; Dioh, B.; Marini,T., 1971. Symp. Int. de la Radioiecolgie appliquee ala protection de l'homme et de son Environnement, Rome, cited Valkovic, V., 1975. <u>Trace</u> <u>Element Analysis</u>. Taylor and Francis, Ltd., London.
- 244. Department of Employment, H.M. Factory, Inspectorate, 1973. <u>Technical Data Note 2/73 Threshold Limit Values for 1973</u>.
- 245. Rancitelli, L.A.; Cooper, J.A.; Perkins, R.W., 1973. Multielement characterisation of atmospheric aerosols by neutron activation analysis and direct X-ray analysis and X-ray fluorescence analysis. Comp. Stud. Food, Environ. Contam., Proc. Symp. (Published 1974) (cited CA 1975, <u>82</u>, 115499).
- 246. Wilkmiss, P.E.; Larson, R.E., 1970. Use of activation analysis to determine the chemical composition and origin of particulates collected in the marine atmosphere. Proceedings of Symposium on Nuclear Techniques in Environmental Pollution, I.A.E.A. Vienna. (Published 1971) p 159.

- 247. Dams, R.; Robbins, J.A.; Rahn, K.A.; Winchester, J.W., 1970. Quantative relation among trace elements over industrialised northwest Indiana. Proceedings of Symposium on Nuclear Techniques in Environmental Pollution, I.A.E.A. Vienne (Published 1971) p 139.
- 248. Wilson, G.S.; Moyers, J.L., 1974. Particle size determination of trace elements in urban air. International Journal of Environmental Analytical Chemistry, <u>3(3)</u>, p 239.
- 249. Roberts, N.J., 1974. Aerosol trace elementanalysis using neutron activation analysis and X-ray fluorescence. Report
 U.C.R.L. 51499, p 127. (University of California, Livermore, California) (cited CA 1975, <u>82</u>, 128858).
- 250. Heindrychx, R.; Dams, R., 1974. Continental, marine and anthropogenic contribution to the inorganic composition of the aerosol of an industrial zone. Journal of Radioanalytical Chemistry, <u>19(2)</u>, p 339.
- 251. Rahn, K.A., 1972. <u>Study of national air pollution by combustion</u> Progress Report 1972.
- 252. Bowen, H.J.M., 1966. <u>Trace Elements in Biochemistry</u>, Academic Press, New York and London.
- 253. Mason, B., 1966. Principles of Geochemistry. 3rd Edition. Wiley and Sons, Inc., New York, London and Sydney.

- 254. Kneip, T.J.; Eisenbud, M.; Strehlow, C.D.; Freudenthal, P.C., 1970. Airborne particulates in New York City. Journal of the Air Pollution Control Association, <u>20(3)</u>, p 144.
- 255. Burnham, C.D.; Moore, C.E.; Kowalski, T.; Krasniewski, J., 1970. Detailed study of lead determinations of airborne particulates over Morton Grove, Illinois, by atomic absorption spectroscopy. Applied Spectroscopy, <u>24(4)</u>, p 411.
- 256. Quickert, N.; Zdrojewski, A.; Dobois, L., 1974. Accurate measurement of vanadium in airborne particulates. International Journal of Environmental Analytical Chemistry, <u>3(3)</u>, p 229.
- 257. Toca, F.M.; Cheever, C.L.; Berry, C.M., 1973. Lead and cadmium distribution in the particulate effluent from a coal fired boiler. Journal of the American Industrial Hygiene Association, <u>34(9)</u>, p 396.
- 258. Natusch, D.F.S.; Wallace, J.R.; Evans, C.A., 1974. Toxic trace elements. Preferential concentration in respirable particles. Science, <u>183</u> (4121), p 202.
- 259. Rahn, K.A., 1974. Physiochemical regularities of the global atmospheric aerosol. Talk given at Purdue University, 11th December.

- 260. Coloves, G.; Wilson, G.S.; Moyers, J.L., 1974. Particle size determination of trace elements in urban air. International Journal of Environmental Analytical Chemistry, <u>3(3)</u>, p 239.
- 261. Reeves, A.L., 1974. Environental danger of peryllium. Zentralbl Arbeitsmed Arbeitsschutz, <u>24(2)</u>, p 46 (cited CA 1974, <u>81</u>, 81814).
- 262. Beattie, A.D.; Moore, M.R.; Goldberg, A., 1975. Role of chronic low-level lead exposure in the aethiology of mental retardation. Lancet, 1, p 589.
- 263. Beevers, D.G.; Erskine, E.; Robertson, M.; Beattie, A.D.; Campbell, B.C.; Goldberg, A.; Moore, M.R.; Hewthorne, V.M., 1976. Blood-lead and hypertension. Lancet, <u>2</u>, p 1.
- 264. Addis, G.; Moore, M.R., 1974. Lead levels in water of suburban Glasgow. Nature (London), <u>252(5479)</u>, p 120.
- 265. Beattie, A.D.; Moore, M.R.; Devenay, W.T.; Miller, A.R.; Goldberg, A., 1972. Environmental lead pollution in an urban soft-water area. British Medical Journal, <u>2</u>, p 491.
- 266. Coldberg, A., 1974. Drinking water as a source of lead pollution. Environmental Health Perspective, 7, p 103.

- 267. Crawford, M.D.; Crawford, T., 1969. Lead content of bones in a soft and a hard water area. Lancet, <u>1</u>, p 699.
- 268. Bacon, A.P.C.; Froome, K.; Gent, A.E.; Cooke, T.K.; Sowerby, P., 1967. Lead poisoning from drinking soft water. Lancet, 1, p 264.
- 269. Crawford, M.D.; Morris, J.N., 1967. Lead in drinking water. Lancet, <u>2</u>, p 1087.
- 270. Chaineuse, J., 1971. Lead. Toxic elements in drinking water in the eastern part of Liege province. Arch. Belg, Med. Soc., Hyg., Med. Trov. Med. Leg. <u>29(9-10)</u>, p 573 (cited CA 1972, <u>77</u>, 122716).
- 271. Heusghem, C.; De Graeve, J., 1973. Contamination with lead of water in Eastern Belgium. Trib. Cent. Belge. Etude. Doc. Eaux Air, 26(354), p 204 (cited CA 1973, <u>79</u>, 88016).
- 272. Tifenboch, A.; Sobezak, W., 1975. Lead and copper concentrations in tap water in the province of Poznan. Rocz. Fanstw Zakl. Hig., <u>26(1)</u>, p 35 (cited CA 1975, <u>83</u>, 32812).
- 273. Hoffman, B., Komenda, W.; Sadowska, A.; Razanski, A., 1972.
 Vanadium in drinking water in Bialystok area. Rocz. Panstw
 Zakl. Hig., <u>23(2)</u>, p 155 (cited CA 1972, <u>77</u>, 66047).

- 274. Heasman, M.A., 1975. Chronic low-level lead and mental retardation. Lancet, <u>1</u>, p 982.
- 275. Elwood, P.C.; Morton, M.; St. Leger, A.S., 1976. Lead in water and mental retardation. Lancet, 1, p 590.
- 276. Elwood, P.C.; St. Leger, A.S.; Moore, F.; Morton, M., 1976. Lead in water and mortality. Lancet, <u>1</u>, p 749.
- 277. Elwood, P.C.; St. Leger, A.S.; Morton, M., 1976. Dependence of blood lead on domestic water lead. Lancet, <u>1</u>, p 1295.
- 278. Guillanmin, J.C., 1974. Determination of trace metals in power plant effluents. Atomic Absorption Newsletter, <u>13(6)</u>, p 135.
- 279. World Health Organisation, 1971. <u>International Standards for</u> Drinking Water. 3rd Edition., Geneva.
- 280. Pulford, I.D., 1976. <u>Reclamation of Industrial Waste and</u> <u>Derelict Land</u>. Ph.D. thesis, University of Glasgow.
- 281. Hartstein, A.M.; Freedman, K.W.; Platter, D.W., 1973. Noval wet digestion procedure for trace metal analysis by atomic absorption. Analytical Chemistry, <u>45(3)</u>, p 611.
- 282. Flanagan, F.J., 1973. 1972 values for international geochemical reference samples. Geochimica et Casonochimica

- 283. Pulford, I.D.; Duncan, H.J., 1975. Predicting the potential acidity in reclaimed coal mine waste. Surveyor, <u>31st January</u>, p 36.
- 284. Mitchell, R.L., 1964. <u>Trace Elements in Soils</u> in <u>Chemistry of</u> <u>the Soil</u> 2nd Edition (Bear, F.E., Editor), Reinhold Publishing Corporation, New York.

